

Pixton's formula and Abel-Jacobi theory on the Picard stack

Bae, Y.; Holmes, D.S.T.; Pandharipande, R.; Schmitt, J.; Schwarz, R.M.

Citation

Bae, Y., Holmes, D. S. T., Pandharipande, R., Schmitt, J., & Schwarz, R. M. (2023). Pixton's formula and Abel-Jacobi theory on the Picard stack. *Acta Mathematica*, *230*(2), 205-319. doi:10.4310/ACTA.2023.v230.n2.a1

Version:Publisher's VersionLicense:Leiden University Non-exclusive licenseDownloaded from:https://hdl.handle.net/1887/3714186

Note: To cite this publication please use the final published version (if applicable).

Acta Math., 230 (2023), 205–319 DOI: 10.4310/ACTA.2023.v230.n2.a1 © 2023 by Institut Mittag-Leffler. All rights reserved

Pixton's formula and Abel–Jacobi theory on the Picard stack

by

Younghan Bae

ETH Zürich Zürich, Switzerland Mathematisch Instituut Leiden Leiden, Netherlands

DAVID HOLMES

RAHUL PANDHARIPANDE

ETH Zürich Zürich, Switzerland Johannes Schmitt

University of Zürich Zürich, Switzerland

Rosa Schwarz

Mathematisch Instituut Leiden Leiden, Netherlands

Contents

0.	Intro	oduction	207
	0.1.	Double ramification cycles $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots$	207
	0.2.	Twisted double ramification cycles $\ldots \ldots \ldots \ldots \ldots$	207
	0.3.	Pixton's formula	210
	0.4.	Vanishing	217
	0.5.	Twisted holomorphic and meromorphic differentials $\ \ldots \ \ldots$	217
	0.6.	Invariance properties	219
	0.7.	Universal formula in degree zero	223
	0.8.	Acknowledgements	224
1.	Nota	ation, conventions, and the plan	225
	1.1.	Ground field	225
	1.2.	Basics	225
	1.3.	Plan of the paper	226
2.	Pica	rd stacks and operational Chow	227
	2.1.	The Picard stack and relative Picard space	227
	2.2.	Operational Chow groups of algebraic stacks	228
	2.3.	Relationship to usual Chow groups	231

3. The universal double ramification cycle2433.1. Overview2433.2. DR $_{\mathcal{P},A}^{op}$ by closure2443.3. Logarithmic definition of DR op 2463.4. Logarithmic rubber definition of DR op 2493.5. The image of the Abel-Jacobi map2503.6. Proof of the equivalence of the definitions2543.7. Proof of Theorem 0.12543.8. The double ramification cycle in b-Chow2554. Pixton's formula2564.1. Reformulation2564.2. Comparison to Pixton's k-twisted formula2604.3. Comparison to Pixton's formula with targets2625. Proof of Theorem 0.72635.1. Overview2635.2. On an open subset of $\overline{\mathcal{M}}_{g,n}(\mathbb{P}^l,\beta)$ 2635.3. For sufficiently positive line bundles2665.4. With sufficiently many sections2675.5. Proof in the general case2686. Comparing rubber and log spaces2726.1. Overview2726.2. Refined definition of the logarithmic rubber space2726.3. The stack of prestable rubber maps2736.4. Comparing the virtual classes2797. Invariance properties2857.2. Proof of Invariance II (Unweighted markings)2867.3. Proof of Invariance II (Weight translation)2907.4. Proof of Invariance II (Weight translation)2907.5. Proof of Invariance II (Weight translation)2907.6. Proof of Invariance II (Vertical twisting)2947.7. Proof of Invariance II (Parti		2.4. Constructing an operational Chow class	233
3.1. Overview2433.2. $DR_{g,A}^{ep}$ by closure2443.3. Logarithmic definition of DR^{ep} 2463.4. Logarithmic rubber definition of DR^{ep} 2493.5. The image of the Abel–Jacobi map2503.6. Proof of the equivalence of the definitions2543.7. Proof of Theorem 0.12543.8. The double ramification cycle in b-Chow2554. Pixton's formula2564.1. Reformulation2564.2. Comparison to Pixton's k-twisted formula2604.3. Comparison to Pixton's formula with targets2625. Proof of Theorem 0.72635.1. Overview2635.2. On an open subset of $\overline{\mathcal{M}}_{g,n}(\mathbb{P}^l, \beta)$ 2635.3. For sufficiently positive line bundles2655.4. With sufficiently many sections2675.5. Proof in the general case2686. Comparing rubber and log spaces2726.1. Overview2726.2. Refined definition of the logarithmic rubber space2726.3. The stack of prestable rubber maps2736.4. Comparing the virtual classes2797. Invariance properties2857.2. Proof of Invariance I (Dualizing)2867.3. Proof of Invariance II (Weight translation)2907.5. Proof of Invariance II (Weight dmarkings)2877.4. Proof of Invariance IV (Vertical twisting)2947.5. Proof of Invariance IV (Vertical twisting)2947.6. Proof of Invariance IV (Vertical twisting)2927.6. Proof of Invariance IV (Vertical twistin	3.	The universal double ramification cycle	243
3.2. $DR^{ep}_{g,A}$ by closure2443.3.Logarithmic definition of DR^{op} 2463.4.Logarithmic rubber definition of DR^{op} 2493.5.The image of the Abel–Jacobi map2503.6.Proof of the equivalence of the definitions2543.7.Proof of the equivalence of the definitions2543.8.The double ramification cycle in b-Chow2554.Pixton's formula2564.1.Reformulation2564.2.Comparison to Pixton's k-twisted formula2604.3.Comparison to Pixton's formula with targets2625.Proof of Theorem 0.72635.1.Overview2635.2.On an open subset of $\overline{\mathcal{M}}_{g,n}(\mathbb{P}^{\ell},\beta)$ 2635.3.For sufficiently positive line bundles2655.4.With sufficiently many sections2675.5.Proof in the general case2686.Comparing rubber and log spaces2726.1.Overview2726.2.Refined definition of the logarithmic rubber space2726.3.The stack of prestable rubber maps2736.4.Comparison to the logarithmic space2766.5.Comparing the virtual classes2797.Invariance I (Dualizing)2867.3.Proof of Invariance II (Weight markings)2877.4.Proof of Invariance II (Weight markings)2867.5.Proof of Invariance II (Weight markings)287 <td></td> <td>3.1. Overview</td> <td>243</td>		3.1. Overview	243
3.3. Logarithmic definition of DR^{op} 2463.4. Logarithmic rubber definition of DR^{op} 2493.5. The image of the Abel-Jacobi map2503.6. Proof of the equivalence of the definitions2543.7. Proof of Theorem 0.12543.8. The double ramification cycle in b-Chow2554. Pixton's formula2564.1. Reformulation2564.2. Comparison to Pixton's k-twisted formula2604.3. Comparison to Pixton's k-twisted formula2604.3. Comparison to Pixton's formula with targets2625. Proof of Theorem 0.72635.1. Overview2635.2. On an open subset of $\overline{\mathcal{M}}_{g,n}(\mathbb{P}^l, \beta)$ 2635.3. For sufficiently positive line bundles2665.4. With sufficiently many sections2675.5. Proof in the general case2726.1. Overview2726.2. Refined definition of the logarithmic rubber space2726.3. The stack of prestable rubber maps2736.4. Comparison to the logarithmic space2766.5. Comparing the virtual classes2797. Invariance properties2857.1. Overview2857.2. Proof of Invariance II (Unweighted markings)2867.3. Proof of Invariance IV (Twisting by pullback)2927.6. Proof of Invariance V (Vertical twisting)2947.7. Proof of Invariance V (Vertical twisting)2947.7. Proof of Invariance V (Vertical twisting)2947.7. Proof of Invariance V (Vertical twisting)2947.7		3.2. $DR_{g,A}^{op}$ by closure	244
3.4. Logarithmic rubber definition of DR^{op} 2493.5. The image of the Abel–Jacobi map2503.6. Proof of the equivalence of the definitions2543.7. Proof of Theorem 0.12543.8. The double ramification cycle in b-Chow2554. Pixton's formula2564.1. Reformulation2564.2. Comparison to Pixton's k-twisted formula2604.3. Comparison to Pixton's formula with targets2625. Proof of Theorem 0.72635.1. Overview2635.2. On an open subset of $\overline{\mathcal{M}}_{g,n}(\mathbb{P}^l, \beta)$ 2635.3. For sufficiently positive line bundles2655.4. With sufficiently many sections2675.5. Proof in the general case2686. Comparing rubber and log spaces2726.1. Overview2726.2. Refined definition of the logarithmic rubber space2726.3. The stack of prestable rubber maps2736.4. Comparing the virtual classes2797. Invariance properties2857.2. Proof of Invariance I (Dualizing)2867.3. Proof of Invariance II (Weight translation)2907.5. Proof of Invariance IV (Twisting by pullback)2927.6. Proof of Invariance IV (Partial stabilization)2988. Applications3098.1. Proof of Theorem 0.9 and Conjecture A3098.2. Closures3118.3. k-twisted DR cycles with targets3128.4. Proof of Theorem 0.8314References316		3.3. Logarithmic definition of DR^{op}	246
3.5. The image of the Abel–Jacobi map2503.6. Proof of the equivalence of the definitions2543.7. Proof of Theorem 0.12543.8. The double ramification cycle in b-Chow2554. Pixton's formula2564.1. Reformulation2564.2. Comparison to Pixton's k-twisted formula2604.3. Comparison to Pixton's formula with targets2625. Proof of Theorem 0.72635.1. Overview2635.2. On an open subset of $\overline{\mathcal{M}}_{g,n}(\mathbb{P}^l, \beta)$ 2635.3. For sufficiently positive line bundles2655.4. With sufficiently many sections2675.5. Proof in the general case2686. Comparing rubber and log spaces2726.1. Overview2726.2. Refined definition of the logarithmic rubber space2726.3. The stack of prestable rubber maps2736.4. Comparing the virtual classes2797. Invariance properties2857.1. Overview2857.2. Proof of Invariance I (Dualizing)2867.3. Proof of Invariance IV (Twisting by pullback)2927.6. Proof of Invariance IV (Vertical twisting)2947.7. Proof of Invariance V (Vertical twisting)2947.7. Proof of Invariance VI (Partial stabilization)2988. Applications3098.1. Proofs of Theorem 0.9 and Conjecture A3098.2. Closures3118.3. k-twisted DR cycles with targets3128.4. Proof of Theorem 0.8314References316<		3.4. Logarithmic rubber definition of DR^{op}	249
3.6. Proof of the equivalence of the definitions2543.7. Proof of Theorem 0.12543.8. The double ramification cycle in b-Chow2554. Pixton's formula2564.1. Reformulation2564.2. Comparison to Pixton's k-twisted formula2604.3. Comparison to Pixton's formula with targets2625. Proof of Theorem 0.72635.1. Overview2635.2. On an open subset of $\overline{\mathcal{M}}_{g,n}(\mathbb{P}^l, \beta)$ 2635.3. For sufficiently positive line bundles2665.4. With sufficiently many sections2675.5. Proof in the general case2686. Comparing rubber and log spaces2726.1. Overview2726.1. Overview2726.3. The stack of prestable rubber maps2736.4. Comparison to the logarithmic rubber space2766.5. Comparing the virtual classes2797. Invariance properties2857.2. Proof of Invariance I (Dualizing)2867.3. Proof of Invariance II (Weight translation)2907.5. Proof of Invariance IV (Twisting by pulback)2927.6. Proof of Invariance V (Vertical twisting)2947.7. Proof of Invariance VI (Partial stabilization)2988. Applications3098.1. Proofs of Theorem 0.83138.5. Connections to past and future results314References316		3.5. The image of the Abel–Jacobi map	250
3.7. Proof of Theorem 0.12543.8. The double ramification cycle in b-Chow2554. Pixton's formula2564.1. Reformulation2564.2. Comparison to Pixton's k-twisted formula2604.3. Comparison to Pixton's formula with targets2625. Proof of Theorem 0.72635.1. Overview2635.2. On an open subset of $\overline{\mathcal{M}}_{g,n}(\mathbb{P}^l, \beta)$ 2635.3. For sufficiently positive line bundles2665.4. With sufficiently many sections2675.5. Proof in the general case2686. Comparing rubber and log spaces2726.1. Overview2726.3. The stack of prestable rubber maps2736.4. Comparison to the logarithmic rubber space2766.5. Comparing the virtual classes2797. Invariance properties2857.2. Proof of Invariance I (Dualizing)2867.3. Proof of Invariance II (Weight translation)2907.5. Proof of Invariance IV (Twisting by pullback)2927.6. Proof of Invariance V (Vertical twisting)2947.7. Proof of Invariance VI (Partial stabilization)2988. Applications3098.1. Proofs of Theorem 0.9 and Conjecture A3098.2. Closures3118.3. k-twisted DR cycles with targets3128.4. Proof of Theorem 0.8314References316		3.6. Proof of the equivalence of the definitions	254
3.8. The double ramification cycle in b-Chow2554. Pixton's formula2564.1. Reformulation2564.2. Comparison to Pixton's k-twisted formula2604.3. Comparison to Pixton's formula with targets2625. Proof of Theorem 0.72635.1. Overview2635.2. On an open subset of $\overline{\mathcal{M}}_{g,n}(\mathbb{P}^t, \beta)$ 2635.3. For sufficiently positive line bundles2655.4. With sufficiently many sections2675.5. Proof in the general case2686. Comparing rubber and log spaces2726.1. Overview2726.2. Refined definition of the logarithmic rubber space2726.3. The stack of prestable rubber maps2736.4. Comparing the virtual classes2797. Invariance properties2857.1. Overview2857.2. Proof of Invariance I (Dualizing)2867.3. Proof of Invariance IV (Twisting by pullback)2927.4. Proof of Invariance IV (Vertical twisting)2947.7. Proof of Invariance V (Vertical twisting)2947.8. Applications3098.1. Proofs of Theorem 0.9 and Conjecture A3098.2. Closures3118.3. k-twisted DR cycles with targets3128.4. Proof of Theorem 0.83138.5. Connections to past and future results314References		3.7. Proof of Theorem 0.1	254
4. Pixton's formula2564.1. Reformulation2564.2. Comparison to Pixton's k-twisted formula2604.3. Comparison to Pixton's formula with targets2625. Proof of Theorem 0.72635.1. Overview2635.2. On an open subset of $\overline{\mathcal{M}}_{g,n}(\mathbb{P}^l, \beta)$ 2635.3. For sufficiently positive line bundles2655.4. With sufficiently many sections2675.5. Proof in the general case2686. Comparing rubber and log spaces2726.1. Overview2726.2. Refined definition of the logarithmic rubber space2726.3. The stack of prestable rubber maps2736.4. Comparing the virtual classes2797. Invariance properties2857.1. Overview2857.2. Proof of Invariance I (Dualizing)2867.3. Proof of Invariance IV (Twisting by pullback)2927.4. Proof of Invariance IV (Vertical twisting)2947.7. Proof of Invariance V (Vertical twisting)2947.7. Proof of Invariance V (Vertical twisting)2947.7. Proof of Invariance V (Partial stabilization)2988. Applications3098.1. Proofs of Theorem 0.9 and Conjecture A3098.2. Closures3118.3. k-twisted DR cycles with targets3128.4. Proof of Theorem 0.83138.5. Connections to past and future results314References316		3.8. The double ramification cycle in b-Chow	255
4.1. Reformulation2564.2. Comparison to Pixton's k-twisted formula2604.3. Comparison to Pixton's formula with targets2625. Proof of Theorem 0.72635.1. Overview2635.2. On an open subset of $\overline{\mathcal{M}}_{g,n}(\mathbb{P}^l,\beta)$ 2635.3. For sufficiently positive line bundles2655.4. With sufficiently many sections2675.5. Proof in the general case2686. Comparing rubber and log spaces2726.1. Overview2726.2. Refined definition of the logarithmic rubber space2726.3. The stack of prestable rubber maps2736.4. Comparing the virtual classes2797. Invariance properties2857.1. Overview2857.2. Proof of Invariance I (Dualizing)2867.3. Proof of Invariance II (Weight translation)2907.5. Proof of Invariance V (Vertical twisting)2947.7. Proof of Invariance V (Vertical twisting)2947.7. Proof of Invariance VI (Partial stabilization)2988. Applications3098.1. Proofs of Theorem 0.9 and Conjecture A3098.2. Closures3118.3. k-twisted DR cycles with targets3128.4. Proof of Theorem 0.83138.5. Connections to past and future results314References316	4.	Pixton's formula	256
4.2. Comparison to Pixton's k-twisted formula2604.3. Comparison to Pixton's formula with targets2625. Proof of Theorem 0.72635.1. Overview2635.2. On an open subset of $\overline{\mathcal{M}}_{g,n}(\mathbb{P}^l, \beta)$ 2635.3. For sufficiently positive line bundles2655.4. With sufficiently many sections2675.5. Proof in the general case2686. Comparing rubber and log spaces2726.1. Overview2726.2. Refined definition of the logarithmic rubber space2726.3. The stack of prestable rubber maps2736.4. Comparing the virtual classes2797. Invariance properties2857.1. Overview2857.2. Proof of Invariance I (Dualizing)2867.3. Proof of Invariance II (Unweighted markings)2907.5. Proof of Invariance IV (Twisting by pullback)2927.6. Proof of Invariance V (Vertical twisting)2947.7. Proof of Invariance VI (Partial stabilization)2988. Applications3098.1. Proofs of Theorem 0.9 and Conjecture A3098.2. Closures3118.3. k-twisted DR cycles with targets3128.4. Proof of Theorem 0.83138.5. Connections to past and future results314References316		4.1. Reformulation	256
4.3. Comparison to Pixton's formula with targets2625. Proof of Theorem 0.72635.1. Overview2635.2. On an open subset of $\overline{\mathcal{M}}_{g,n}(\mathbb{P}^l,\beta)$ 2635.3. For sufficiently positive line bundles2655.4. With sufficiently many sections2675.5. Proof in the general case2686. Comparing rubber and log spaces2726.1. Overview2726.2. Refined definition of the logarithmic rubber space2726.3. The stack of prestable rubber maps2736.4. Comparing the virtual classes2797. Invariance properties2857.1. Overview2857.2. Proof of Invariance I (Dualizing)2867.3. Proof of Invariance II (Weight translation)2907.5. Proof of Invariance V (Vertical twisting)2947.7. Proof of Invariance V (Vertical twisting)2947.7. Proof of Invariance V (Partial stabilization)2988. Applications3098.1. Proofs of Theorem 0.9 and Conjecture A3098.2. Closures3118.3. k-twisted DR cycles with targets3128.4. Proof of Theorem 0.83138.5. Connections to past and future results314References316		4.2. Comparison to Pixton's k-twisted formula	260
5. Proof of Theorem 0.72635.1. Overview2635.2. On an open subset of $\overline{\mathcal{M}}_{g,n}(\mathbb{P}^l,\beta)$ 2635.3. For sufficiently positive line bundles2655.4. With sufficiently many sections2675.5. Proof in the general case2686. Comparing rubber and log spaces2726.1. Overview2726.2. Refined definition of the logarithmic rubber space2726.3. The stack of prestable rubber maps2736.4. Comparing the virtual classes2797. Invariance properties2857.1. Overview2857.2. Proof of Invariance I (Dualizing)2867.3. Proof of Invariance III (Weight translation)2907.5. Proof of Invariance IV (Twisting by pullback)2927.6. Proof of Invariance V (Vertical twisting)2947.7. Proof of Invariance VI (Partial stabilization)2988. Applications3098.1. Proofs of Theorem 0.9 and Conjecture A3098.2. Closures3118.3. k-twisted DR cycles with targets3128.4. Proof of Theorem 0.83138.5. Connections to past and future results314References316		4.3. Comparison to Pixton's formula with targets	262
5.1. Overview2635.2. On an open subset of $\overline{\mathcal{M}}_{g,n}(\mathbb{P}^l,\beta)$ 2635.3. For sufficiently positive line bundles2655.4. With sufficiently many sections2675.5. Proof in the general case2686. Comparing rubber and log spaces2726.1. Overview2726.2. Refined definition of the logarithmic rubber space2726.3. The stack of prestable rubber maps2736.4. Comparing the virtual classes2797. Invariance properties2857.1. Overview2857.2. Proof of Invariance I (Dualizing)2867.3. Proof of Invariance II (Unweighted markings)2877.4. Proof of Invariance IV (Twisting by pulback)2927.6. Proof of Invariance V (Vertical twisting)2947.7. Proof of Invariance VI (Partial stabilization)2988. Applications3098.1. Proofs of Theorem 0.9 and Conjecture A3098.2. Closures3118.3. k-twisted DR cycles with targets3138.5. Connections to past and future results314References316	5.	Proof of Theorem 0.7	263
5.2. On an open subset of $\overline{\mathcal{M}}_{g,n}(\mathbb{P}^l,\beta)$ 2635.3. For sufficiently positive line bundles2655.4. With sufficiently many sections2675.5. Proof in the general case2686. Comparing rubber and log spaces2726.1. Overview2726.2. Refined definition of the logarithmic rubber space2726.3. The stack of prestable rubber maps2736.4. Comparison to the logarithmic space2766.5. Comparing the virtual classes2797. Invariance properties2857.1. Overview2857.2. Proof of Invariance I (Dualizing)2867.3. Proof of Invariance III (Weight translation)2907.5. Proof of Invariance IV (Twisting by pulback)2927.6. Proof of Invariance IV (Vertical twisting)2947.7. Proof of Invariance VI (Partial stabilization)2988. Applications3098.1. Proofs of Theorem 0.9 and Conjecture A3098.2. Closures3118.3. k-twisted DR cycles with targets3128.4. Proof of Theorem 0.83138.5. Connections to past and future results314References316		5.1. Overview	263
5.3. For sufficiently positive line bundles 265 5.4. With sufficiently many sections 267 5.5. Proof in the general case 268 6. Comparing rubber and log spaces 272 6.1. Overview 272 6.2. Refined definition of the logarithmic rubber space 272 6.3. The stack of prestable rubber maps 273 6.4. Comparison to the logarithmic space 276 6.5. Comparing the virtual classes 279 7. Invariance properties 285 7.1. Overview 285 7.2. Proof of Invariance I (Dualizing) 286 7.3. Proof of Invariance II (Unweighted markings) 287 7.4. Proof of Invariance IV (Twisting by pullback) 292 7.5. Proof of Invariance IV (Twisting by pullback) 292 7.6. Proof of Invariance VI (Partial stabilization) 298 8. Applications 309 8.1. Proofs of Theorem 0.9 and Conjecture A 309 8.2. Closures 311 8.3. k-twisted DR cycles with targets 313 8.5. Connections to past and future results 314		5.2. On an open subset of $\overline{\mathcal{M}}_{q,n}(\mathbb{P}^l,\beta)$	263
5.4. With sufficiently many sections2675.5. Proof in the general case2686. Comparing rubber and log spaces2726.1. Overview2726.2. Refined definition of the logarithmic rubber space2726.3. The stack of prestable rubber maps2736.4. Comparison to the logarithmic space2766.5. Comparing the virtual classes2797. Invariance properties2857.1. Overview2857.2. Proof of Invariance I (Dualizing)2867.3. Proof of Invariance III (Weight translation)2907.5. Proof of Invariance IV (Twisting by pullback)2927.6. Proof of Invariance V (Vertical twisting)2947.7. Proof of Invariance V (Partial stabilization)2988. Applications3098.1. Proofs of Theorem 0.9 and Conjecture A3098.2. Closures3118.3. k-twisted DR cycles with targets3138.5. Connections to past and future results314References316		5.3. For sufficiently positive line bundles	265
5.5. Proof in the general case2686. Comparing rubber and log spaces2726.1. Overview2726.2. Refined definition of the logarithmic rubber space2726.3. The stack of prestable rubber maps2736.4. Comparison to the logarithmic space2766.5. Comparing the virtual classes2797. Invariance properties2857.1. Overview2857.2. Proof of Invariance I (Dualizing)2867.3. Proof of Invariance III (Weight translation)2907.5. Proof of Invariance IV (Twisting by pullback)2927.6. Proof of Invariance V (Vertical twisting)2947.7. Proof of Invariance VI (Partial stabilization)2988. Applications3098.1. Proofs of Theorem 0.9 and Conjecture A3098.2. Closures3118.3. k-twisted DR cycles with targets3128.4. Proof of Theorem 0.83138.5. Connections to past and future results314References316		5.4. With sufficiently many sections	267
6. Comparing rubber and log spaces 272 6.1. Overview 272 6.2. Refined definition of the logarithmic rubber space 272 6.3. The stack of prestable rubber maps 273 6.4. Comparison to the logarithmic space 276 6.5. Comparing the virtual classes 279 7. Invariance properties 285 7.1. Overview 285 7.2. Proof of Invariance I (Dualizing) 286 7.3. Proof of Invariance II (Unweighted markings) 287 7.4. Proof of Invariance III (Weight translation) 290 7.5. Proof of Invariance IV (Twisting by pullback) 292 7.6. Proof of Invariance V (Vertical twisting) 294 7.7. Proof of Invariance VI (Partial stabilization) 298 8. Applications 309 8.1. Proofs of Theorem 0.9 and Conjecture A 309 8.2. Closures 311 8.3. k-twisted DR cycles with targets 312 8.4. Proof of Theorem 0.8 313 8.5. Connections to past and future results 314 References 316		5.5. Proof in the general case	268
6.1. Overview2726.2. Refined definition of the logarithmic rubber space2726.3. The stack of prestable rubber maps2736.4. Comparison to the logarithmic space2766.5. Comparing the virtual classes2797. Invariance properties2857.1. Overview2857.2. Proof of Invariance I (Dualizing)2867.3. Proof of Invariance II (Unweighted markings)2877.4. Proof of Invariance III (Weight translation)2907.5. Proof of Invariance IV (Twisting by pullback)2927.6. Proof of Invariance V (Vertical twisting)2947.7. Proof of Invariance VI (Partial stabilization)2988. Applications3098.1. Proofs of Theorem 0.9 and Conjecture A3098.2. Closures3118.3. k-twisted DR cycles with targets3128.4. Proof of Theorem 0.83138.5. Connections to past and future results314References316	6.	Comparing rubber and log spaces	272
6.2. Refined definition of the logarithmic rubber space 272 6.3. The stack of prestable rubber maps 273 6.4. Comparison to the logarithmic space 276 6.5. Comparing the virtual classes 279 7. Invariance properties 285 7.1. Overview 285 7.2. Proof of Invariance I (Dualizing) 286 7.3. Proof of Invariance II (Unweighted markings) 287 7.4. Proof of Invariance III (Weight translation) 290 7.5. Proof of Invariance IV (Twisting by pullback) 292 7.6. Proof of Invariance V (Vertical twisting) 294 7.7. Proof of Invariance VI (Partial stabilization) 298 8. Applications 309 8.1. Proofs of Theorem 0.9 and Conjecture A 309 8.2. Closures 311 8.3. k-twisted DR cycles with targets 312 8.4. Proof of Theorem 0.8 313 8.5. Connections to past and future results 314 References 316		6.1. Overview	272
6.3. The stack of prestable rubber maps 273 6.4. Comparison to the logarithmic space 276 6.5. Comparing the virtual classes 279 7. Invariance properties 285 7.1. Overview 285 7.2. Proof of Invariance I (Dualizing) 286 7.3. Proof of Invariance II (Unweighted markings) 287 7.4. Proof of Invariance III (Weight translation) 290 7.5. Proof of Invariance IV (Twisting by pullback) 292 7.6. Proof of Invariance V (Vertical twisting) 294 7.7. Proof of Invariance VI (Partial stabilization) 298 8. Applications 309 8.1. Proofs of Theorem 0.9 and Conjecture A 309 8.2. Closures 311 8.3. k-twisted DR cycles with targets 312 8.4. Proof of Theorem 0.8 313 8.5. Connections to past and future results 314 References 316		6.2. Refined definition of the logarithmic rubber space	272
6.4. Comparison to the logarithmic space 276 6.5. Comparing the virtual classes 279 7. Invariance properties 285 7.1. Overview 285 7.2. Proof of Invariance I (Dualizing) 286 7.3. Proof of Invariance II (Unweighted markings) 287 7.4. Proof of Invariance III (Weight translation) 290 7.5. Proof of Invariance IV (Twisting by pullback) 292 7.6. Proof of Invariance V (Vertical twisting) 294 7.7. Proof of Invariance VI (Partial stabilization) 298 8. Applications 309 8.1. Proofs of Theorem 0.9 and Conjecture A 309 8.2. Closures 311 8.3. k-twisted DR cycles with targets 312 8.4. Proof of Theorem 0.8 313 8.5. Connections to past and future results 314 References 316		6.3. The stack of prestable rubber maps	273
6.5. Comparing the virtual classes 279 7. Invariance properties 285 7.1. Overview 285 7.2. Proof of Invariance I (Dualizing) 286 7.3. Proof of Invariance II (Unweighted markings) 287 7.4. Proof of Invariance III (Weight translation) 290 7.5. Proof of Invariance IV (Twisting by pullback) 292 7.6. Proof of Invariance V (Vertical twisting) 294 7.7. Proof of Invariance VI (Partial stabilization) 298 8. Applications 309 8.1. Proofs of Theorem 0.9 and Conjecture A 309 8.2. Closures 311 8.3. k-twisted DR cycles with targets 312 8.4. Proof of Theorem 0.8 313 8.5. Connections to past and future results 314 References 316		6.4. Comparison to the logarithmic space	276
7. Invariance properties 285 7.1. Overview 285 7.2. Proof of Invariance I (Dualizing) 286 7.3. Proof of Invariance II (Unweighted markings) 287 7.4. Proof of Invariance III (Weight translation) 290 7.5. Proof of Invariance IV (Twisting by pullback) 292 7.6. Proof of Invariance V (Vertical twisting) 294 7.7. Proof of Invariance VI (Partial stabilization) 298 8. Applications 309 8.1. Proofs of Theorem 0.9 and Conjecture A 309 8.2. Closures 311 8.3. k-twisted DR cycles with targets 312 8.4. Proof of Theorem 0.8 313 8.5. Connections to past and future results 314 References 316		6.5. Comparing the virtual classes	279
7.1. Overview 285 7.2. Proof of Invariance I (Dualizing) 286 7.3. Proof of Invariance II (Unweighted markings) 287 7.4. Proof of Invariance III (Weight translation) 290 7.5. Proof of Invariance IV (Twisting by pullback) 292 7.6. Proof of Invariance V (Vertical twisting) 294 7.7. Proof of Invariance VI (Partial stabilization) 298 8. Applications 309 8.1. Proofs of Theorem 0.9 and Conjecture A 309 8.2. Closures 311 8.3. k-twisted DR cycles with targets 312 8.4. Proof of Theorem 0.8 313 8.5. Connections to past and future results 314 References 316	7.	Invariance properties	285
7.2. Proof of Invariance I (Dualizing) 286 7.3. Proof of Invariance II (Unweighted markings) 287 7.4. Proof of Invariance III (Weight translation) 290 7.5. Proof of Invariance IV (Twisting by pullback) 292 7.6. Proof of Invariance V (Vertical twisting) 294 7.7. Proof of Invariance VI (Partial stabilization) 298 8. Applications 309 8.1. Proofs of Theorem 0.9 and Conjecture A 309 8.2. Closures 311 8.3. k-twisted DR cycles with targets 312 8.4. Proof of Theorem 0.8 313 8.5. Connections to past and future results 314 References 316		7.1. Overview	285
7.3. Proof of Invariance II (Unweighted markings)2877.4. Proof of Invariance III (Weight translation)2907.5. Proof of Invariance IV (Twisting by pullback)2927.6. Proof of Invariance V (Vertical twisting)2947.7. Proof of Invariance VI (Partial stabilization)2988. Applications3098.1. Proofs of Theorem 0.9 and Conjecture A3098.2. Closures3118.3. k-twisted DR cycles with targets3128.4. Proof of Theorem 0.83138.5. Connections to past and future results314References316		7.2. Proof of Invariance I (Dualizing)	286
7.4. Proof of Invariance III (Weight translation)2907.5. Proof of Invariance IV (Twisting by pullback)2927.6. Proof of Invariance V (Vertical twisting)2947.7. Proof of Invariance VI (Partial stabilization)2988. Applications3098.1. Proofs of Theorem 0.9 and Conjecture A3098.2. Closures3118.3. k-twisted DR cycles with targets3128.4. Proof of Theorem 0.83138.5. Connections to past and future results314References316		7.3. Proof of Invariance II (Unweighted markings)	287
7.5. Proof of Invariance IV (Twisting by pullback) 292 7.6. Proof of Invariance V (Vertical twisting) 294 7.7. Proof of Invariance VI (Partial stabilization) 298 8. Applications 309 8.1. Proofs of Theorem 0.9 and Conjecture A 309 8.2. Closures 311 8.3. k-twisted DR cycles with targets 312 8.4. Proof of Theorem 0.8 313 8.5. Connections to past and future results 314 References 316		7.4. Proof of Invariance III (Weight translation)	290
7.6. Proof of Invariance V (Vertical twisting) 294 7.7. Proof of Invariance VI (Partial stabilization) 298 8. Applications 309 8.1. Proofs of Theorem 0.9 and Conjecture A 309 8.2. Closures 311 8.3. k-twisted DR cycles with targets 312 8.4. Proof of Theorem 0.8 313 8.5. Connections to past and future results 314 References 316		7.5. Proof of Invariance IV (Twisting by pullback)	292
7.7. Proof of Invariance VI (Partial stabilization) 298 8. Applications 309 8.1. Proofs of Theorem 0.9 and Conjecture A 309 8.2. Closures 311 8.3. k-twisted DR cycles with targets 312 8.4. Proof of Theorem 0.8 313 8.5. Connections to past and future results 314 References 316		7.6. Proof of Invariance V (Vertical twisting)	294
8. Applications 309 8.1. Proofs of Theorem 0.9 and Conjecture A 309 8.2. Closures 311 8.3. k-twisted DR cycles with targets 312 8.4. Proof of Theorem 0.8 313 8.5. Connections to past and future results 314 References 316		7.7. Proof of Invariance VI (Partial stabilization)	298
8.1. Proofs of Theorem 0.9 and Conjecture A 309 8.2. Closures 311 8.3. k-twisted DR cycles with targets 312 8.4. Proof of Theorem 0.8 313 8.5. Connections to past and future results 314 References 316	8.	Applications	309
8.2. Closures 311 8.3. k-twisted DR cycles with targets 312 8.4. Proof of Theorem 0.8 313 8.5. Connections to past and future results 314 References 316		8.1. Proofs of Theorem 0.9 and Conjecture A	309
8.3. k-twisted DR cycles with targets 312 8.4. Proof of Theorem 0.8 313 8.5. Connections to past and future results 314 References 316		8.2. Closures	311
8.4. Proof of Theorem 0.8 313 8.5. Connections to past and future results 314 References 316		8.3. <i>k</i> -twisted DR cycles with targets	312
8.5. Connections to past and future results		8.4. Proof of Theorem 0.8	313
References		8.5. Connections to past and future results	314
	Re	- eferences	316

0. Introduction

0.1. Double ramification cycles

Let $A = (a_1, ..., a_n)$ be a vector of n integers satisfying

$$\sum_{i=1}^{n} a_i = 0.$$

In the moduli space $\mathcal{M}_{g,n}$ of non-singular curves of genus g with n marked points, consider the substack defined by the following classical condition:

$$\left\{ (C, p_1, ..., p_n) \in \mathcal{M}_{g,n} : \mathcal{O}_C \left(\sum_{i=1}^n a_i p_i \right) \simeq \mathcal{O}_C \right\}.$$
(0.1)

From the point of view of relative Gromov–Witten theory, the most natural compactification of the substack (0.1) is the space $\overline{\mathcal{M}}_{g,A}^{\sim}$ of stable maps to *rubber*: stable maps to \mathbb{CP}^1 relative to zero and ∞ modulo the \mathbb{C}^* -action on \mathbb{CP}^1 . The rubber moduli space carries a natural virtual fundamental class $[\overline{\mathcal{M}}_{g,A}^{\sim}]^{\text{vir}}$ of (complex) dimension 2g-3+n. The pushforward via the canonical morphism

$$\varepsilon: \overline{\mathcal{M}}_{g,A}^{\sim} \longrightarrow \overline{\mathcal{M}}_{g,n}$$

is the *double ramification cycle*

$$\varepsilon_*[\overline{\mathcal{M}}_{g,A}^{\sim}]^{\operatorname{vir}} = \mathsf{DR}_{g,A} \in \mathsf{CH}_{2g-3+n}(\overline{\mathcal{M}}_{g,n}).$$

Double ramification cycles have been studied intensively for the past two decades. Examples of early results can be found in [17], [18], [21], [25], [33], [35], [57]. A complete formula was conjectured by Pixton in 2014 and proven in [42]. For subsequent study and applications, see [4], [16], [20], [26], [28], [37]–[39], [59], [62], [66], [72], [75]. Essential for our work is the formula for double ramification cycles for target varieties in [43].

We refer the reader to [42, §0] and [64, §5] for introductions to the subject. For a classical perspective from the point of view of Abel–Jacobi theory, see [37].

0.2. Twisted double ramification cycles

We develop here a theory which extends the study of double ramification cycles from the moduli space of stable curves $\overline{\mathcal{M}}_{g,n}$ to the Picard stack of curves with line bundles $\mathfrak{Pic}_{g,n}$. An object of $\mathfrak{Pic}_{g,n}$ over \mathcal{S} is a flat family

$$\pi\colon \mathcal{C} \longrightarrow \mathcal{S}$$

of prestable(1) *n*-pointed genus-*g* curves together with a line bundle

$$\mathcal{L} \longrightarrow \mathcal{C}.$$

The Picard stack $\mathfrak{Pic}_{g,n}$ is an algebraic (Artin) stack which is locally of finite type; see §2.1 for a treatment of foundational issues.

Since the degree of a line bundle is constant in flat families, there is a disjoint union

$$\mathfrak{Pic}_{g,n} = \bigcup_{d \in \mathbb{Z}} \mathfrak{Pic}_{g,n,d},$$

where $\mathfrak{Pic}_{q,n,d}$ is the Picard stack of curves with degree-d line bundles. Let

$$A = (a_1, ..., a_n), \quad \sum_{i=1}^n a_i = d,$$

be a vector of integers. The first result of the paper is the construction of a *universal* twisted double ramification cycle in the operational Chow theory⁽²⁾ of $\mathfrak{Pic}_{a.n.d}$,

$$\mathsf{DR}_{g,A}^{\mathsf{op}} \in \mathsf{CH}_{\mathsf{op}}^g(\mathfrak{Pic}_{g,n,d}).$$

The geometric intuition behind the construction is simple. Let

$$\pi: \mathcal{C} \longrightarrow \mathcal{S}, \quad p_1, \dots, p_n: \mathcal{S} \longrightarrow \mathcal{C}, \quad \mathcal{L} \longrightarrow \mathcal{C}$$

be an object of $\mathfrak{Pic}_{g,n,d}$. The class $\mathsf{DR}_{g,A}^{\mathsf{op}}$ should operate as the locus in the base \mathcal{S} heuristically determined by the condition

$$\mathcal{O}_C\left(\sum_{i=1}^n a_i p_i\right) \simeq \mathcal{L}|_C.$$

To make the above idea precise, we do *not* use the virtual class of the moduli space of stable maps in Gromov–Witten theory, but rather an alternative approach by partially resolving the classical Abel–Jacobi map. The method follows the path of [37], [39] and may be viewed as a universal Abel–Jacobi construction over the Picard stack. Log geometry based on the stack of tropical divisors constructed in [56] plays a crucial role. Our construction is presented in §3.8.

The basic compatibility of our new operational class

$$\mathsf{DR}_{g,A}^{\mathsf{op}} \in \mathsf{CH}_{\mathsf{op}}^g(\mathfrak{Pic}_{g,n,d})$$

⁽¹⁾ A prestable *n*-pointed curve is a connected nodal curve with markings at distinct non-singular points. For the entire paper, we avoid the (g, n) = (1, 0) case because of non-affine stabilizers.

 $^(^2)$ All Chow theories in the paper will be taken with $\mathbb Q\text{-coefficients}.$

with the standard double ramification cycle is as follows. Let

$$A = (a_1, \dots, a_n), \quad \sum_{i=1}^n a_i = 0,$$

be given. The universal data

$$\pi: \mathcal{C}_{g,n} \longrightarrow \overline{\mathcal{M}}_{g,n}, \quad \mathcal{O} \longrightarrow \mathcal{C}_{g,n} \tag{0.2}$$

determine a map $\varphi_{\mathcal{O}}: \overline{\mathcal{M}}_{g,n} \to \mathfrak{Pic}_{g,n,0}$. The action of $\mathsf{DR}_{g,A}^{\mathsf{op}}$ on the fundamental class of $\overline{\mathcal{M}}_{g,n}$ corresponding to the family (0.2) then equals the previously defined double ramification cycle

$$\mathsf{DR}_{g,A}^{\mathsf{op}}(\varphi_{\mathcal{O}})([\overline{\mathcal{M}}_{g,n}]) = \mathsf{DR}_{g,A} \in \mathsf{CH}_{2g-3+n}(\overline{\mathcal{M}}_{g,n}).$$

More generally, for a vector $A = (a_1, ..., a_n)$ of integers satisfying

$$\sum_{i=1}^n a_i \,{=}\, k(2g {-}\, 2),$$

canonically twisted double ramification cycles,

$$\mathsf{DR}_{g,A,\omega^k} \in \mathsf{CH}_{2g-3+n}(\overline{\mathcal{M}}_{g,n}),$$

related to the classical loci

$$\left\{ (C, p_1, ..., p_n) \in \mathcal{M}_{g,n} : \mathcal{O}_C \left(\sum_{i=1}^n a_i p_i \right) \simeq \omega_C^k \right\},\$$

have been constructed in [34] for k=1 and in [37], [38], [56] for all $k \ge 1$. The universal data

$$\pi: \mathcal{C}_{g,n} \longrightarrow \overline{\mathcal{M}}_{g,n}, \quad \omega_{\pi}^k \longrightarrow \mathcal{C}_{g,n} \tag{0.3}$$

determine a map $\varphi_{\omega_{\pi}^{k}}: \overline{\mathcal{M}}_{g,n} \to \mathfrak{Pic}_{g,n,k(2g-2)}$. Here, ω_{π} is the relative dualizing sheaf of the morphism π .

The action of $\mathsf{DR}_{g,A}^{\mathsf{op}}$ on the fundamental class of $\overline{\mathcal{M}}_{g,n}$ corresponding to the family (0.3) is compatible with the constructions of [34], [37], [38], [56],

$$\mathsf{DR}_{g,A}^{\mathsf{op}}(\varphi_{\omega_\pi^k})([\overline{\mathcal{M}}_{g,n}]) \,{=}\, \mathsf{DR}_{g,A,\omega^k} \,{\in}\, \mathsf{CH}_{2g-3+n}(\overline{\mathcal{M}}_{g,n})$$

for all $k \ge 1$.

The above compatibilities of $\mathsf{DR}_{g,A}^{\mathsf{op}}$ with the standard and canonically twisted double ramification cycles are proven in §3.7.

THEOREM 0.1. Let $g \ge 0$ and $d \in \mathbb{Z}$. Let $A = (a_1, ..., a_n)$ be a vector of integers satisfying

$$\sum_{i=1}^{n} a_i = d.$$

Logarithmic compactification of the Abel–Jacobi map yields a universal twisted double ramification cycle

$$\mathsf{DR}_{g,A}^{\mathsf{op}} \in \mathsf{CH}_{\mathsf{op}}^g(\mathfrak{Pic}_{g,n,d})$$

which is compatible with the standard double ramification cycle

$$\mathsf{DR}_{g,A,\omega^k} \in \mathsf{CH}_{2g-3+n}(\overline{\mathcal{M}}_{g,n})$$

in case d = k(2g-2) for $k \ge 0$.

0.3. Pixton's formula

0.3.1. Prestable graphs

We define the set $\mathsf{G}_{g,n}$ of prestable graphs as follows. A *prestable graph* $\Gamma \in \mathsf{G}_{g,n}$ consists of the data

$$\Gamma = (V, H, L, g: V \to \mathbb{Z}_{\geq 0}, v: H \to V, \iota: H \to H)$$

satisfying the following properties:

(i) V is a vertex set with a genus function $g: V \to \mathbb{Z}_{\geq 0}$;

(ii) H is a half-edge set equipped with a vertex assignment $v: H \rightarrow V$ and an involution ι ;

(iii) E, the edge set, is defined by the 2-cycles of ι in H (self-edges at vertices are permitted);

(iv) L, the set of legs, is defined by the fixed points of ι and is placed in bijective correspondence with a set of n markings;

(v) the pair (V, E) defines a *connected* graph satisfying the genus condition

$$\sum_{v\in \mathcal{V}} \mathbf{g}(v) \! + \! h^1(\Gamma) \! = \! g.$$

To emphasize Γ , the notation V(Γ), H(Γ), L(Γ), and E(Γ) will also be used for the vertex, half-edges, legs, and edges of Γ , respectively.

An isomorphism between $\Gamma, \Gamma' \in \mathsf{G}_{g,n}$ consists of bijections $V \to V'$ and $H \to H'$ respecting the structures L, g, v, and ι . Let $\operatorname{Aut}(\Gamma)$ denote the automorphism group of Γ .

210

211

While the set of isomorphism classes of prestable graphs is infinite, the set of isomorphism classes of prestable graphs with prescribed bounds on the number of edges is finite.

Let $\mathfrak{M}_{g,n}$ be the algebraic (Artin) stack of prestable curves of genus g with n marked points. A prestable graph Γ determines an algebraic stack \mathfrak{M}_{Γ} of curves with degenerations forced by the graph,

$$\mathfrak{M}_{\Gamma} = \prod_{v \in \mathcal{V}} \mathfrak{M}_{g(v), \mathbf{n}(v)},$$

together with a canonical $(^3)$ map

$$j_{\Gamma}:\mathfrak{M}_{\Gamma}\longrightarrow\mathfrak{M}_{g,n}.$$

Since $\mathfrak{M}_{g,n}$ is smooth and the morphism j_{Γ} is proper, representable, and lci, we obtain an operational Chow class on the algebraic stack of curves,

$$j_{\Gamma*}[\mathfrak{M}_{\Gamma}] \in \mathsf{CH}^{|\mathrm{E}(\Gamma)|}_{\mathsf{op}}(\mathfrak{M}_{g,n}).$$

Via the morphism of algebraic stacks,

$$\varepsilon: \mathfrak{Pic}_{q,n,d} \longrightarrow \mathfrak{M}_{q,n},$$

 $j_{\Gamma*}[\mathfrak{M}_{\Gamma}]$ also defines an operational Chow class on the Picard stack,

$$arepsilon^* j_{\Gamma *} [\mathfrak{M}_{\Gamma}] \in \mathsf{CH}^{|\mathrm{E}(\Gamma)|}_{\mathsf{op}}(\mathfrak{Pic}_{g,n,d}).$$

0.3.2. Prestable graphs with degrees

We will require a refinement of the prestable graphs of 0.3.1 which includes degrees of line bundles.

We define the set $G_{g,n,d}$ of prestable graphs of degree d as follows:

$$\Gamma_{\delta} = (\Gamma, \delta) \in \mathsf{G}_{g,n,d}$$

consists of the data

- a prestable graph $\Gamma \in \mathsf{G}_{g,n}$;
- a function $\delta: \mathbf{V} \rightarrow \mathbb{Z}$ satisfying the degree condition

$$\sum_{v \in V} \delta(v) = d.$$

 $^(^{3})$ To define the map, we choose an ordering on the half-edges at each vertex.

The function δ is often called the *multidegree*.

An automorphism of $\Gamma_{\delta} \in \mathsf{G}_{g,n,d}$ consists of an automorphism of Γ leaving δ invariant. Let $\operatorname{Aut}(\Gamma_{\delta})$ denote the automorphism group of Γ_{δ} .

For $\Gamma_{\delta} \in \mathsf{G}_{g,n,k}$, let \mathfrak{M}_{Γ} be the algebraic stack of curves defined in §0.3.1 with respect to the underlying prestable graph Γ . Let $\mathfrak{Pic}_{\Gamma_{\delta}}$ be the Picard stack,

$$\varepsilon: \mathfrak{Pic}_{\Gamma_s} \longrightarrow \mathfrak{M}_{\Gamma},$$

parameterizing curves with degenerations forced by Γ and with line bundles which have degree $\delta(v)$ restriction to the components corresponding to the vertex $v \in V$. We have a canonical map

$$j_{\Gamma_{\delta}}: \mathfrak{Pic}_{\Gamma_{\delta}} \longrightarrow \mathfrak{Pic}_{g,n,d}.$$

Since $\mathfrak{Pic}_{g,n,d}$ is smooth and the morphism $j_{\Gamma_{\delta}}$ is proper, representable, and lci, we obtain an operational Chow class,

$$j_{\Gamma_{\delta}*}[\mathfrak{Pic}_{\Gamma_{\delta}}] \in \mathsf{CH}_{\mathsf{op}}^{|\mathrm{E}(\Gamma)|}(\mathfrak{Pic}_{q,n,d})$$

As operational Chow classes, the following formula holds:

$$\varepsilon^* j_{\Gamma*}[\mathfrak{M}_{\Gamma}] = \sum_{\delta} j_{\Gamma_{\delta}*}[\mathfrak{Pic}_{\Gamma_{\delta}}] \in \mathsf{CH}^{|\mathcal{E}(\Gamma)|}_{\mathsf{op}}(\mathfrak{Pic}_{g,n,d}), \tag{0.4}$$

where the sum⁽⁴⁾ is over all functions $\delta: V \to \mathbb{Z}$ satisfying the degree condition. Equivalently, we may write (0.4) as

$$\frac{1}{|\operatorname{Aut}(\Gamma)|} \, \varepsilon^* j_{\Gamma*}[\mathfrak{M}_{\Gamma}] = \sum_{\Gamma_{\delta} \in \mathsf{G}_{g,n,d}} \frac{1}{|\operatorname{Aut}(\Gamma_{\delta})|} \, j_{\Gamma_{\delta}*}[\mathfrak{Pic}_{\Gamma_{\delta}}] \in \mathsf{CH}_{\mathsf{op}}^{|\operatorname{E}(\Gamma)|}(\mathfrak{Pic}_{g,n,d}),$$

where the sum on the right-hand side is now over all isomorphism classes of prestable graphs of degree d with underlying prestable graph Γ .

0.3.3. Tautological ψ , ξ , and η classes

The universal curve

$$\pi: \mathfrak{C}_{g,n} \longrightarrow \mathfrak{Pic}_{g,n}$$

carries two natural line bundles: the relative dualizing sheaf ω_{π} and the universal line bundle

 $\mathfrak{L} \longrightarrow \mathfrak{C}_{g,n}.$

^{(&}lt;sup>4</sup>) The sum is infinite, but only finitely many terms are non-zero in any operation.

Let p_i be the *i*th section of the universal curve, let

$$\mathfrak{S}_i \subset \mathfrak{C}_{g,n}$$

be the corresponding divisor, and let

$$\omega_{\log} = \omega_{\pi} \bigg(\sum_{i=1}^{n} \mathfrak{S}_{i} \bigg)$$

be the relative log-canonical line bundle with first Chern class $c_1(\omega_{\log})$. Let

$$\xi = c_1(\mathfrak{L})$$

be the first Chern class of \mathfrak{L} .

Definition 0.2. The following operational classes on $\mathfrak{Pic}_{q,n}$ are obtained from the following universal structures:

- $$\begin{split} & \bullet \quad & \\ \bullet \quad & \\ \psi_i \!=\! c_1(p_i^* \omega_\pi) \!\in\! \mathsf{CH}^1_{\mathsf{op}}(\mathfrak{Pic}_{g,n}) \,; \\ & \bullet \quad & \\ \xi_i \!=\! c_1(p_i^* \mathfrak{L}) \!\in\! \mathsf{CH}^1_{\mathsf{op}}(\mathfrak{Pic}_{g,n}) \,; \\ & \bullet \quad & \\ \eta_{a,b} \!=\! \pi_*(c_1(\omega_{\mathrm{log}})^a \xi^b) \!\in\! \mathsf{CH}^{a+b-1}_{\mathsf{op}}(\mathfrak{Pic}_{g,n}). \end{split}$$

For simplicity in the formulas, we will use the notation

$$\eta = \eta_{0,2} = \pi_*(\xi^2).$$

The standard κ classes are defined by the π pushforwards of powers of $c_1(\omega_{\log})$,

$$\eta_{a,0} = \kappa_{a-1}.$$

Definition 0.3. A decorated prestable graph $[\Gamma_{\delta}, \gamma]$ of degree d is a prestable graph $\Gamma_{\delta} \in \mathsf{G}_{q,n,d}$ of degree d together with the following decoration data γ :

- each leg $i \in L$ is decorated with a monomial $\psi_i^a \xi_i^b$;
- each half-edge $h \in H \setminus L$ is decorated with a monomial ψ_h^a ;
- each edge $e \in E$ is decorated with a monomial ξ_e^a ;
- each vertex in V is decorated with a monomial in the variables $\{\eta_{a,b}\}_{a+b\geq 2}$.

In all four cases, the monomial may be be trivial.

Let $\mathsf{DG}_{q,n,d}$ be the set of decorated prestable graphs of degree d. To each decorated graph of degree d,

$$[\Gamma_{\delta}, \gamma] \in \mathsf{DG}_{g,n,d},$$

we assign the operational class

$$j_{\Gamma_{\delta}*}[\gamma] \in \mathsf{CH}^*_{\mathsf{op}}(\mathfrak{Pic}_{g,n,d})$$

obtained via the proper representable morphism

$$j_{\Gamma_{\delta}}:\mathfrak{Pic}_{\Gamma_{\delta}}\longrightarrow\mathfrak{Pic}_{g,n,d}$$

and the action of the decorations.

The action of decorations is described as follows. Given $\Gamma_{\delta} \in \mathsf{G}_{g,n,k}$, the stack $\mathfrak{Pic}_{\Gamma_{\delta}}$ admits a morphism⁽⁵⁾

$$\mathfrak{Pic}_{\Gamma_{\delta}} \longrightarrow \prod_{v \in \mathcal{V}(\Gamma_{\delta})} \mathfrak{Pic}_{g(v), n(v), \delta(v)}$$

which sends a line bundle \mathcal{L} on a prestable curve \mathcal{C} to its restrictions on the various components,

$$\mathcal{L}|_{\mathcal{C}_v}, \quad v \in \mathcal{V}(\Gamma_\delta).$$

For $v \in V(\Gamma_{\delta})$, we define the operational class $\eta(v)$ on $\mathfrak{Pic}_{\Gamma_{\delta}}$ as the pullback of the operational class η on the factor $\mathfrak{Pic}_{g(v),n(v),\delta(v)}$ above. The operational classes ψ at the markings and ξ at the half-edges are defined similarly.

Definition 0.4. The tautological classes in $\mathsf{CH}^*_{\mathsf{op}}(\mathfrak{Pic}_{g,n,d})$ consist of the \mathbb{Q} -linear span of the operational classes associated with all $[\Gamma_{\delta}, \gamma] \in \mathsf{DG}_{g,n,d}$.

By standard analysis [30], the tautological classes have a natural ring structure. Our formula for $\mathsf{DR}_{g,A}^{\mathsf{op}}$ will be a sum of operational classes determined by decorated prestable graphs of degree $d = \sum_{i=1}^{n} a_i$ (and hence will be tautological).

0.3.4. Weightings mod r

Let $\Gamma_{\delta} \in \mathsf{G}_{q,n,d}$ be a prestable graph of degree d, and let r be a positive integer.

Definition 0.5. A weighting mod r of Γ_{δ} is a function on the set of half-edges,

$$w: \operatorname{H}(\Gamma_{\delta}) \longrightarrow \{0, 1, ..., r-1\},\$$

which satisfies the following three properties:

(i) for all $i \in L(\Gamma_{\delta})$, corresponding to the marking $i \in \{1, ..., n\}$,

$$w(i) = a_i \mod r;$$

(ii) for all $e \in E(\Gamma_{\delta})$, corresponding to two half-edges $h, h' \in H(\Gamma_{\delta})$,

 $w(h) + w(h') = 0 \mod r;$

^{(&}lt;sup>5</sup>) The fibers of the map are torsors under the group $\mathbb{G}_m^{h^1(\Gamma)}$.

(iii) for all $v \in \mathcal{V}(\Gamma_{\delta})$,

$$\sum_{v(h)=v} w(h) = \delta(v) \mod r,$$

where the sum is taken over all n(v) half-edges incident to v.

We denote by $W_{\Gamma_{\delta},r}$ the finite set of all possible weightings mod r of Γ_{δ} . The set $W_{\Gamma_{\delta},r}$ has cardinality $r^{h^{1}(\Gamma_{\delta})}$. We view r as a *regularization parameter*.

0.3.5. Calculation of the twisted double ramification cycle

We denote by $\mathsf{P}_{g,A,d}^{c,r} \in \mathsf{CH}_{\mathsf{op}}^{c}(\mathfrak{Pic}_{g,n,d})$ the codimension- $c(^{6})$ component of the tautological operational class

$$\begin{split} \sum_{\substack{\Gamma_{\delta} \in \mathsf{G}_{g,n,d} \\ w \in \mathsf{W}_{\Gamma_{\delta},r}}} \frac{r^{-h^{1}(\Gamma_{\delta})}}{|\operatorname{Aut}(\Gamma_{\delta})|} j_{\Gamma_{\delta}*} \bigg[\prod_{i=1}^{n} \exp\left(\frac{1}{2}a_{i}^{2}\psi_{i} + a_{i}\xi_{i}\right) \prod_{v \in \mathcal{V}(\Gamma_{\delta})} \exp\left(-\frac{1}{2}\eta(v)\right) \\ \times \prod_{e=(h,h') \in \mathcal{E}(\Gamma_{\delta})} \frac{1}{\psi_{h} + \psi_{h'}} \bigg(1 - \exp\left(-\frac{w(h)w(h')}{2}(\psi_{h} + \psi_{h'})\right) \bigg) \bigg]. \end{split}$$

Several remarks about the formula are required.

(i) The sum is over all isomorphism classes of prestable graphs of degree d in the set $\mathsf{G}_{g,n,d}$. Only finitely many underlying prestable graphs $\Gamma \in \mathsf{G}_{g,n}$ can contribute in fixed codimension c. However, for each such prestable graph, the above formula has infinitely many summands corresponding to the infinitely many functions

$$\delta: V \longrightarrow \mathbb{Z}$$

which satisfy the degree condition. The operational Chow class $\mathsf{P}_{g,A,d}^{c,r}$ is nevertheless well defined, since only finitely many summands have non-vanishing operation on any given family of curves carrying a degree-*d* line bundle over a base \mathcal{S} of finite type.

(ii) Once the prestable graph Γ_{δ} is chosen, we sum over all $r^{h^1(\Gamma_{\delta})}$ different weightings $w \in W_{\Gamma_{\delta},r}$.

(iii) Inside the pushforward in the above formula, the first product

$$\prod_{i=1}^n \exp\!\left(\frac{1}{2}a_i^2\psi_{h_i}\!+\!a_i\xi_{h_i}\right)$$

is over $h \in L(\Gamma)$ via the correspondence of legs and markings.

 $^(^{6})$ Codimension here is usually called degree. But since we already have line bundle degrees, we use the term codimension for clarity.

- (iv) The class $\eta(v)$ is the $\eta_{0,2}$ class of Definition 0.2 associated with the vertex.
- (v) The third product is over all $e \in E(\Gamma_{\delta})$. The factor

$$\frac{1}{\psi_h + \psi_{h'}} \left(1 - \exp\left(-\frac{w(h)w(h')}{2}(\psi_h + \psi_{h'})\right) \right)$$

is well defined, since

- the denominator formally divides the numerator,
- the factor is symmetric in h and h'.

No edge orientation is necessary.

The following fundamental polynomiality property of $\mathsf{P}_{g,A,d}^{c,r}$ is parallel to Pixton's polynomiality in [42, Appendix] and is a consequence of [42, Proposition 3"].

PROPOSITION 0.6. For fixed g, A, d, and c and a decorated graph $[\Gamma_{\delta}, \gamma]$ of degree d, the coefficient of $j_{\Gamma_{\delta}*}[\gamma]$ in the tautological class

$$\mathsf{P}_{g,A,d}^{c,r} \in \mathsf{CH}_{\mathsf{op}}^c(\mathfrak{Pic}_{g,n,d})$$

is polynomial in r (for all sufficiently large r).

We denote by $\mathsf{P}_{g,A,d}^c$ the value at r=0 of the polynomial associated with $\mathsf{P}_{g,A,d}^{c,r}$ by Proposition 0.6. In other words, $\mathsf{P}_{g,A,d}^c$ is the *constant* term of the associated polynomial in r.

The main result of the paper is a formula for the universal twisted double ramification cycle in operational $\text{Chow.}(^{7})$

THEOREM 0.7. Let $g \ge 0$ and $d \in \mathbb{Z}$. Let $A = (a_1, ..., a_n)$ be a vector of integers with

$$\sum_{i=1}^{n} a_i = d$$

The universal twisted double ramification cycle is calculated by Pixton's formula:

$$\mathsf{DR}_{g,A}^{\mathsf{op}} = \mathsf{P}_{g,A,d}^g \in \mathsf{CH}_{\mathsf{op}}^g(\mathfrak{Pic}_{g,n,d}).$$

Theorem 0.7 is the most fundamental formulation of the relationship between Abel– Jacobi theory and Pixton's formula that we know. Certainly, Theorem 0.7 implies the double ramification cycle and X-valued double ramification cycle results of [42], [43]. But since we will use [43] in the proof of Theorem 0.7, we provide no new approach to these older results. However, the additional depth of Theorem 0.7 immediately allows new applications.

216

^{(&}lt;sup>7</sup>) Our handling of the prefactor 2^{-g} in [42, Theorem 1] differs here. The factors of 2 are now placed in the definition of $P_{g,A,d}^{c,r}$ as in [43].

217

0.4. Vanishing

From his original double ramification cycle formula, Pixton conjectured an associated vanishing property in the tautological ring of the moduli space of curves which was proven by Clader and Janda [21]. The parallel vanishing statement in the tautological ring of the moduli space of stable maps to X was proven in [4]. The most general vanishing statement is the following result.

THEOREM 0.8. Let $g \ge 0$ and $d \in \mathbb{Z}$. Let $A = (a_1, ..., a_n)$ be a vector of integers with

$$\sum_{i=1}^{n} a_i = d.$$

Pixton's vanishing holds in operational Chow:

$$\mathsf{P}^c_{g,A,d} = 0 \in \mathsf{CH}^c_{\mathsf{op}}(\mathfrak{Pic}_{g,n,d}) \quad for \ all \ c > g.$$

Theorem 0.8 may be viewed as providing relations among tautological classes in the operational Chow ring of the Picard stack—a new direction of study with many open questions.⁽⁸⁾ While Theorem 0.8 implies the vanishings of [4], [21], we will use these results in our proof.

0.5. Twisted holomorphic and meromorphic differentials

0.5.1. Fundamental classes

Let $A = (a_1, ..., a_n)$ be a vector of zero and pole multiplicities satisfying

$$\sum_{i=1}^{n} a_i = 2g - 2.$$

Let $\mathcal{H}_g(A) \subset \mathcal{M}_{g,n}$ be the quasi-projective locus of pointed curves $(C, p_1, ..., p_n)$ satisfying the condition

$$\mathcal{O}_C\left(\sum_{i=1}^n a_i p_i\right) \simeq \omega_C.$$

In other words, $\mathcal{H}_g(A)$ is the locus of meromorphic differentials⁽⁹⁾ with zero and pole multiplicities prescribed by A. In [28], a compact moduli space of twisted canonical divisors

$$\mathcal{H}_g(A) \subset \overline{\mathcal{M}}_{g,n}$$

 $^(^8)$ See [67], [77] for tautological relations on the Picard variety over the moduli space of smooth curves.

⁽⁹⁾ If all the parts of A are non-negative, then $\mathcal{H}_g(A)$ is the locus of holomorphic differentials.

is constructed which contains $\mathcal{H}_q(A)$ as an open set.

In the strictly meromorphic case, where A contains at least one strictly negative part, $\widetilde{\mathcal{H}}_g(A)$ is of pure codimension g in $\overline{\mathcal{M}}_{g,n}$ by [28, Theorem 3]. A weighted fundamental cycle of $\widetilde{\mathcal{H}}_g(A)$,

$$\mathsf{H}_{q,A} \in \mathsf{CH}_{2q-3+n}(\overline{\mathcal{M}}_{q,n}),\tag{0.5}$$

is constructed in [28, Appendix A] with explicit non-trivial weights on the irreducible components. In the strictly meromorphic case, $\mathcal{H}_g(A) \subset \overline{\mathcal{M}}_{g,n}$ is also of pure codimension g. The closure

$$\overline{\mathcal{H}}_g(A) \subset \overline{\mathcal{M}}_{g,n}$$

contributes to the fundamental class $H_{g,A}$ with multiplicity 1, but there are additional boundary contributions, see [28, Appendix A].

The universal family over the moduli space of stable curves together with the relative dualizing sheaf,

$$\pi: \mathcal{C}_{g,n} \longrightarrow \overline{\mathcal{M}}_{g,n}, \quad \omega_{\pi} \longrightarrow \mathcal{C}_{g,n}, \tag{0.6}$$

determine an object of $\mathfrak{Pic}_{g,n,2g-2}$. By [40] and the compatibility of Theorem 0.1, the action of $\mathsf{DR}_{g,A}^{\mathsf{op}}$ on the fundamental class of $\overline{\mathcal{M}}_{g,n}$ equals the weighted fundamental class of $\widetilde{\mathcal{H}}_g(A)$:

$$\mathsf{DR}_{g,A}^{\mathsf{op}}(\varphi_{\omega})([\overline{\mathcal{M}}_{g,n}]) = \mathsf{H}_{g,A} \in \mathsf{CH}_{2g-3+n}(\overline{\mathcal{M}}_{g,n}).$$

We can now apply Theorem 0.7 to prove the following result.

THEOREM 0.9. In the strictly meromorphic case,

$$\mathsf{H}_{g,A} = \mathsf{P}_{g,A,2g-2}^{g}[\overline{\mathcal{M}}_{g,n}]$$

for the universal family

$$\pi: \mathcal{C}_{g,n} \longrightarrow \overline{\mathcal{M}}_{g,n}, \quad \omega_{\pi} \longrightarrow \mathcal{C}_{g,n}.$$

Theorem 0.9 is exactly equivalent to Conjecture A in [28, Appendix A]. Since both the moduli space $\widetilde{\mathcal{H}}_g(A)$ and the weighted fundamental cycle $\mathsf{H}_{g,A}$ have explicit geometric definitions, the result provides a geometric representative of Pixton's cycle class in terms of twisted differentials. Theorem 0.9 is proven in §8.1, where the parallel conjectures [72] for higher differentials are also proven (by parallel arguments).

0.5.2. Closures

Let $A = (a_1, ..., a_n)$ be a vector of integers satisfying

$$\sum_{i=1}^n a_i = 2g - 2$$

219

A careful investigation of the closure

$$\mathcal{H}_g(A) \subset \overline{\mathcal{H}}_g(A) \subset \overline{\mathcal{M}}_{g,n}$$

is carried out in [8] in both the holomorphic and meromorphic cases. By a simple procedure presented in [28, Appendix], Theorem 0.9 determines the cycle classes of the closures

$$[\overline{\mathcal{H}}_g(A)] \in \mathsf{CH}_*(\overline{\mathcal{M}}_{g,n}).$$

for A in both the holomorphic and meromorphic cases.

A similar procedure determines the corresponding classes for k-differentials, see [72, §3.4] for an explanation. In particular, our results imply that the cycle classes of the closures are tautological⁽¹⁰⁾ for all k (as was previously known only for k=1, due to [70]).

In the case of holomorphic differentials, another approach to the class of the closure

$$\overline{\mathcal{H}}_g(A) \subset \overline{\mathcal{M}}_{g,n}$$

is provided by Conjecture A.1 in [65, Appendix] via a limit of Witten's r-spin class. A significant first step in the proof of [65, Conjecture A.1] by Chen, Janda, Ruan, and Sauvaget can be found in [19]. Further progress requires a virtual localization analysis for moduli spaces of stable log maps. An approach to Theorem 0.9 using log stable maps, virtual localization in the log context, and the strategy of [42] also appears possible (once the required moduli spaces and localization formulas are established).

0.6. Invariance properties

The universal twisted double ramification cycle has several basic invariance properties which play an important role in our study.

Recall that an object of $\mathfrak{Pic}_{g,n,d}$ over \mathcal{S} is a flat family of prestable *n*-pointed genus-*g* curves together with a line bundle of relative degree *d*,

$$\pi: \mathcal{C} \longrightarrow \mathcal{S}, \quad p_1, \dots, p_n: \mathcal{S} \longrightarrow \mathcal{C}, \quad \mathcal{L} \longrightarrow \mathcal{C}.$$
(0.7)

Let $\mathsf{DR}_{g,A,\mathcal{L}}^{\mathsf{op}} \in \mathsf{CH}_{\mathsf{op}}^g(\mathcal{S})$ be the twisted double ramification cycle associated with the above family (0.7) and the vector

$$A = (a_1, ..., a_n), \quad d = \sum_{i=1}^n a_i.$$

 $^(^{10})$ The precise statement is given in Corollary 8.2 of §8.2.

Invariance I. (Dualizing)

A new object of $\mathfrak{Pic}_{a,n,-d}$ over \mathcal{S} is obtained from (0.7) by dualizing \mathcal{L} :

$$\pi: \mathcal{C} \to \mathcal{S}, \quad p_1, \dots, p_n: \mathcal{S} \to \mathcal{C}, \quad \mathcal{L}^* \to \mathcal{C}.$$

$$(0.8)$$

Let $\mathsf{DR}_{g,-A,\mathcal{L}^*}^{\mathsf{op}} \in \mathsf{CH}_{\mathsf{op}}^g(\mathcal{S})$ be the twisted double ramification cycle associated with the new family (0.8) and the vector $-A = (-a_1, ..., -a_n)$. We have the invariance

$$\mathsf{DR}_{g,-A,\mathcal{L}^*}^{\mathsf{op}} = \varepsilon^* \mathsf{DR}_{g,A,\mathcal{L}}^{\mathsf{op}},$$

where $\varepsilon: \mathfrak{Pic}_{q,n,-d} \to \mathfrak{Pic}_{q,n,d}$ is the natural map obtained via dualizing the line bundle.

Invariance II. (Unweighted markings)

Assume that we have an additional section $p_{n+1}: S \to C$ of π which yields an object of $\mathfrak{Pic}_{q,n+1,d}:$

$$\pi: \mathcal{C} \to \mathcal{S}, \quad p_1, \dots, p_n, p_{n+1}: \mathcal{S} \to \mathcal{C}, \qquad \mathcal{L} \to \mathcal{C}.$$
 (0.9)

Let $A_0 \in \mathbb{Z}^{n+1}$ be the vector obtained by appending zero (as the last coefficient) to A. Let $\mathsf{DR}_{g,A_0,\mathcal{L}}^{\mathsf{op}} \in \mathsf{CH}_{\mathsf{op}}^g(\mathcal{S})$ be the twisted double ramification cycle associated with the new family (0.9) and the vector A_0 . We have the invariance

$$\mathsf{DR}^{\mathsf{op}}_{g,A_0,\mathcal{L}} = F^* \mathsf{DR}^{\mathsf{op}}_{g,A,\mathcal{L}},$$

where $F: \mathfrak{Pic}_{q,n+1,d} \to \mathfrak{Pic}_{q,n,d}$ is the map forgetting the last marking.

Invariance III. (Weight translation)

Let $B = (b_1, ..., b_n) \in \mathbb{Z}^n$ satisfy

$$\sum_{i=1}^{n} b_i = e.$$

Then, the family

$$\pi: \mathcal{C} \longrightarrow \mathcal{S}, \quad p_1, \dots, p_n: \mathcal{S} \longrightarrow \mathcal{C}, \quad \mathcal{L}\left(\sum_{i=1}^n b_i p_i\right) \longrightarrow \mathcal{C}.$$
 (0.10)

defines an object of $\mathfrak{Pic}_{g,n,d+e}$. Let $\mathsf{DR}_{g,A+B,\mathcal{L}(\sum_i b_i p_i)}^{\mathsf{op}} \in \mathsf{CH}_{\mathsf{op}}^g(\mathcal{S})$ be the twisted double ramification cycle associated with the new family (0.10) and the vector A+B. We have the invariance

$$\mathsf{DR}_{g,A+B,\mathcal{L}(\sum_i b_i p_i)}^{\mathsf{op}} = \mathsf{DR}_{g,A,\mathcal{L}}^{\mathsf{op}}$$

220

Invariance IV. (Twisting by pullback)

Let $\mathcal{B} \to \mathcal{S}$ be any line bundle on the base. By tensoring (0.7) with $\pi^* \mathcal{B}$, we obtain a new object of $\mathfrak{Pic}_{q,n,d}$ over \mathcal{S} :

$$\pi: \mathcal{C} \longrightarrow \mathcal{S}, \quad p_1, \dots, p_n: \mathcal{S} \longrightarrow \mathcal{C}, \quad \mathcal{L} \otimes \pi^* \mathcal{B} \longrightarrow \mathcal{C}.$$
(0.11)

Let $\mathsf{DR}_{g,A,\mathcal{L}\otimes\pi^*\mathcal{B}}^{\mathsf{op}} \in \mathsf{CH}_{\mathsf{op}}^g(\mathcal{S})$ be the twisted double ramification cycle associated with the new family (0.11) and the vector A. We have the invariance

$$\mathsf{DR}_{g,A,\mathcal{L}\otimes\pi^*\mathcal{B}}^{\mathsf{op}} = \mathsf{DR}_{g,A,\mathcal{L}}^{\mathsf{op}}$$

Invariance V. (Vertical twisting)

Consider a partition of the genus, marking, and degree data:

$$g_1 + g_2 = g, \quad N_1 \sqcup N_2 = \{1, ..., n\}, \quad d_1 + d_2 = d,$$
 (0.12)

which is *not* symmetric. $(^{11})$ Such a partition defines a divisor

$$\Delta_1 \in \mathsf{CH}^1(\mathfrak{C}_{g,n,d})$$

in the universal curve over $\mathfrak{Pic}_{g,n,d}$ by twisting by the (g_1, N_1, d_1) -component of a curve with a separating node with separating data (0.12).

By tensoring (0.7) with $\mathcal{O}_{\mathcal{C}}(\Delta_1)$, we obtain a new object of $\mathfrak{Pic}_{q,n,d}$ over \mathcal{S} :

$$\pi: \mathcal{C} \longrightarrow \mathcal{S}, \quad p_1, ..., p_n: \mathcal{S} \longrightarrow \mathcal{C}, \quad \mathcal{L}(\Delta_1) \longrightarrow \mathcal{C}.$$
 (0.13)

Let

$$\mathsf{DR}^{\mathsf{op}}_{g,A,\mathcal{L}(\Delta_1)} \! \in \! \mathsf{CH}^g_{\mathsf{op}}(\mathcal{S})$$

be the twisted double ramification cycle associated with the new family (0.13) and the vector A. We have the invariance

$$\mathsf{DR}_{g,A,\mathcal{L}(\Delta_1)}^{\mathsf{op}} = \mathsf{DR}_{g,A,\mathcal{L}}^{\mathsf{op}}.$$
 (0.14)

For symmetric separating data (0.12), equality (0.14) holds with $\Delta_1 \subset \mathfrak{C}_{g,n,d}$ defined as the full preimage of the locus $\Delta \subset \mathfrak{Pic}_{g,n,d}$ of curves with a separating node (0.12). Then, equality (0.14) follows from Invariance IV with $\mathcal{B}=\mathcal{O}(\Delta)$.

^{(&}lt;sup>11</sup>) We require $(g_1, N_1, d_1) \neq (g_2, N_2, d_2)$ so that the two sides of a separating node with separating data (0.12) can be distinguished.

Invariance VI. (Partial stabilization)

Consider a second family of prestable *n*-pointed genus-*g* curves over S,

$$\pi': \mathcal{C}' \longrightarrow \mathcal{S}, \quad p_1', \dots, p_n': \mathcal{S} \longrightarrow \mathcal{C}',$$

together with a birational \mathcal{S} -morphism

$$f: \mathcal{C}' \longrightarrow \mathcal{C}, \quad f \circ p'_i = p_i.$$

A line bundle of relative degree d is defined on \mathcal{C}' by

$$f^*\mathcal{L} \longrightarrow \mathcal{C}'.$$

We require the following property to hold:

If the section p'_i meets the exceptional locus of f, then $a_i = 0$.

Let $\mathsf{DR}_{g,A,f^*\mathcal{L}}^{\mathsf{op}} \in \mathsf{CH}_{\mathsf{op}}^g(\mathcal{S})$ be the twisted double ramification cycle associated with the new family

$$\pi': \mathcal{C}' \longrightarrow \mathcal{S}, \quad p_1', ..., p_n': \mathcal{S} \longrightarrow \mathcal{C}', \quad f^* \mathcal{L} \longrightarrow \mathcal{C}'$$

$$(0.15)$$

and the vector A. We have the invariance

$$\mathsf{DR}_{g,A,f^*\mathcal{L}}^{\mathsf{op}} = \mathsf{DR}_{g,A,\mathcal{L}}^{\mathsf{op}}.$$

Theorem 0.7 provides two paths to viewing the above invariance properties. The invariances can be seen either from formal properties of the geometric construction of the universal twisted double ramification cycle or from formal symmetries of Pixton's formula. In fact, all invariances hold not only for the codimension-g part $\mathsf{P}_{g,A,d}^g$ which computes the double ramification cycle, but for the full mixed-degree class $\mathsf{P}_{g,A,d}^{\mathfrak{g}}$.

For example, Invariance VI on the formula side says that for the maps

$$\varphi_{\mathcal{L}}, \varphi_{f^*\mathcal{L}} \colon \mathcal{S} \longrightarrow \mathfrak{Pic}_{g,n,d}$$

obtained from the families (0.7) and (0.15), we have

$$\varphi_{f^*\mathcal{L}}^*\mathsf{P}^g_{g,A,d} \!=\! \varphi_{\mathcal{L}}^*\mathsf{P}^g_{g,A,d} \!\in\! \mathsf{CH}^g_{\mathsf{op}}(\mathcal{S})$$

for $\mathsf{P}_{q,A,d}^g \in \mathsf{CH}_{\mathsf{op}}^g(\mathfrak{Pic}_{g,n,d})$.

Proofs of all of the invariances will be presented in §7. The above invariances (together with geometric definitions when transversality to the Abel–Jacobi map holds) do not characterize⁽¹²⁾ $\mathsf{DR}^{\mathsf{op}}$.

 $^(^{12})$ Further geometric properties are required, see [41, §1.6] for a discussion.

223

0.7. Universal formula in degree zero

The most efficient statement of the double ramification cycle formula on the Picard stack of curves occurs in the degree d=0 case with *no* markings. In order to avoid⁽¹³⁾ the unpointed genus 1 case, let $g \neq 1$.

The specialization of Theorem 0.7 to $d{=}0$ calculates $\mathsf{DR}^{\mathsf{op}}_{g,\varnothing}$ as the value at $r{=}0$ of the degree- g part of

$$\exp\left(-\frac{1}{2}\eta\right) \sum_{\substack{\Gamma_{\delta} \in \mathsf{G}_{g,0,0} \\ w \in \mathsf{W}_{\Gamma_{\delta},r}}} \frac{r^{-h^{1}(\Gamma_{\delta})}}{|\operatorname{Aut}(\Gamma_{\delta})|} j_{\Gamma_{\delta}*} \left[\prod_{e=(h,h')\in \mathsf{E}(\Gamma_{\delta})} \frac{1}{\psi_{h} + \psi_{h'}} \times \left(1 - \exp\left(-\frac{w(h)w(h')}{2}(\psi_{h} + \psi_{h'})\right)\right)\right]$$

as an operational Chow class on $\mathfrak{Pic}_g = \mathfrak{Pic}_{g,0,0}$. The full statement of Theorem 0.7 can be recovered from the above d=0 specialization via pullback under the map

$$\mathfrak{Pic}_{g,n,d} \longrightarrow \mathfrak{Pic}_{g},$$
$$(C, p_1, ..., p_n, \mathcal{L}) \longmapsto \left(C, \mathcal{L}\left(-\sum_{i=1}^n a_i p_i\right)\right).$$

Indeed, the above map is the composition of the morphism

$$\tau_{-A}:\mathfrak{Pic}_{g,n,d}\longrightarrow \mathfrak{Pic}_{g,n,0},$$
$$(C, p_1, ..., p_n, \mathcal{L})\longmapsto \left(C, p_1, ..., p_n, \mathcal{L}\left(-\sum_{i=1}^n a_i p_i\right)\right),$$

with the morphism

$$F:\mathfrak{Pic}_{g,n,0}\longrightarrow\mathfrak{Pic}_{g}$$

forgetting the markings $p_1, ..., p_n$.

• For the $\mathsf{DR}_{q,A}^{\mathsf{op}}$ side of Theorem 0.7, Invariance II implies

$$F^*\mathsf{DR}_{q,\varnothing}^{\mathsf{op}} = \mathsf{DR}_{q,\mathbf{0}}^{\mathsf{op}}$$

for the zero vector $\mathbf{0} \in \mathbb{Z}^n$. Furthermore, Invariance III implies

$$\tau_{-A}^*\mathsf{DR}^{\mathsf{op}}_{g,\mathbf{0}} \,{=}\, \mathsf{DR}^{\mathsf{op}}_{g,A}.$$

• For the $\mathsf{P}_{g,A,d}^g$ side of Theorem 0.7, the corresponding invariance properties of Pixton's formula (discussed in §7) yield the parallel transformation

$$\tau_{-A}^* F^* \mathsf{P}_{g,\varnothing,0}^g = \mathsf{P}_{g,A,d}^g$$

 $^(^{13})$ For g=1, a parallel formula holds for n=1 and A=(0).

Therefore, the equality in Theorem 0.7 for general A and d follows from the specialization to $A=\emptyset$ and d=0. For certain steps in our proof of Theorem 0.7, the $A=\emptyset$ and d=0 geometry is advantageous and is used.

By restricting to suitable open subsets of $\mathfrak{Pic}_g,$ we can simplify the $d{=}0$ formula even further. Let

$$\mathfrak{Pic}_q^{\mathsf{ct}} \subset \mathfrak{Pic}_q$$

be the locus where the curve C is of compact type. We obtain

$$\mathsf{DR}_{g,\varnothing}^{\mathsf{op}}|_{\mathfrak{Pic}_{g}^{\mathsf{ct}}} = \frac{\theta^{g}}{g!}, \quad \text{for } \theta = -\frac{1}{2} \bigg(\eta + \sum_{\Delta} d_{\Delta}^{2}[\Delta] \bigg), \tag{0.16}$$

where the sum is over the boundary divisors $\Delta \subset \mathfrak{Pic}_g$ on which generically the curve splits into two components carrying line bundles of degrees d_{Δ} and $-d_{\Delta}$. The class θ here may be viewed as a universal theta divisor on $\mathfrak{Pic}_q^{\mathsf{ct}}$.

Formula (0.16) was first written on the moduli space of stable curves of compact type in [33], [35]. The operational Chow class $\mathsf{DR}_{g,\varnothing}^{\mathsf{op}}$ on \mathfrak{Pic}_g , however, is *not* the power of a divisor.

0.8. Acknowledgements

We thank D. Chen, A. Chiodo, F. Janda, G. Farkas, T. Graber, S. Grushevsky, A. Kresch, S. Molcho, M. Möller, A. Pixton, D. Ranganathan, A. Sauvaget, H.-H. Tseng, J. Wise, and D. Zvonkine for many discussions about Abel–Jacobi theory, double ramification cycles, and meromorphic differentials. The AIM workshop on *Double ramification cycles and integrable systems* played a key role at the start of the paper. We are grateful to the organisers A. Buryak, R. Cavalieri, E. Clader, and P. Rossi. We thank the referees for several improvements in the presentation.

Y.B. was supported by ERC-2017-AdG-786580-MACI and the Korea Foundation for Advanced Studies. D. H. was partially supported by NWO grant 613.009.103. R. P. was partially supported by SNF-200020-182181, SwissMAP, and the Einstein Stiftung. J. S. was supported by the SNF Early Postdoc.Mobility grant 184245 and thanks the Max Planck Institute for Mathematics in Bonn for its hospitality. R. S. was supported by NWO grant 613.009.113.

The project has received funding from the European Research Council (ERC) under the European Union Horizon 2020 research and innovation program (grant agreement No. 786580).

1. Notation, conventions, and the plan

1.1. Ground field

In the introduction, the ground field was the complex numbers \mathbb{C} . However, for the remainder of the paper, we will work more generally over a field K of characteristic zero. We will make essential use of the results of [43] which are stated over \mathbb{C} , but also hold over \mathbb{Q} by the following standard argument:

(i) Both the DR cycle (via the b-Chow approach [37]) and Pixton's class are defined over \mathbb{Q} .

(ii) Rational equivalence of cycles uses finitely many subschemes and rational functions and hence descends to a finitely generated \mathbb{Q} -subalgebra of \mathbb{C} . A non-empty scheme of finite type over \mathbb{Q} has points over some finite extension, and hence the rational equivalence descends to a finite extension of \mathbb{Q} .

(iii) The rational equivalence descends (via a Galois argument) further to \mathbb{Q} , since we work in Chow with rational coefficients.

By similar arguments, our results are in fact true over $\mathbb{Z}[1/N]$ for a positive integer N depending on the ramification data. Understanding what happens at small primes (or integral Chow groups) is an interesting question.

1.2. Basics

Let K be a ground field of characteristic zero. When we work in the logarithmic category, we assume Spec K to be equipped with the trivial log structure.

We write $\overline{\mathcal{M}}f$ for the stack of all stable (ordered) marked curves over K and \mathfrak{M} for the stack of all prestable curves with ordered marked points. Both come with natural log structures, and the universal marked curves over these spaces are naturally log curves. For $\overline{\mathcal{M}}$, the log structure is described in [46]. The same construction applies unchanged to \mathfrak{M} , see [32, Appendix A]. The natural open immersion

$$\overline{\mathcal{M}} \longrightarrow \mathfrak{M}$$

is strict (though, in contrast to [32], [46], we order the markings of our log curves). We use subscripts g and n to fix the genus and number of markings when necessary.

Let \mathfrak{C} be the universal prestable curve over \mathfrak{M} . For efficiency of notation, we will also denote by \mathfrak{C} the universal curve over the various other moduli stacks of curves with additional structure which will appear in the paper. These universal curves are always obtained by pulling-back \mathfrak{C} over \mathfrak{M} .

For the convenience of the reader, we provide the key notation in Table 1.

stack of prestable marked curves of genus g with n markings
stack of stable marked curves of genus g with n markings
universal prestable curve over \mathfrak{M}
$A \in \mathbb{Z}^n$ with $\sum_{i=1}^n a_i$ denoted by d
Picard stack (§2.1)
relative universal Picard stack $(\S2.1)$
operational Chow group $(\S2.2)$
tautological class on $\mathfrak{Pic}_{g,n}$ (§0.3.3)
tautological class on $\mathfrak{Pic}_{g,n}$ (§0.3.3)
tautological class on $\mathfrak{Pic}_{g,n}$ (§0.3.3)
Pixton's cycle $(\S0.3.5)$
homomorphisms $c(\varphi)$: $CH_*(B) \to CH_{*-p}(B)$ given by an operational
class $c \in CH^p_{op}(\mathfrak{X})$ and $\varphi: B \to \mathfrak{X}$ with B finite-type scheme (§2.2)
operational DR cycle ($\S3$)

Table 1. Key notation.

1.3. Plan of the paper

In §2 we treat several technical issues related to operational Chow groups of the Picard stack \mathfrak{Pic} . In fact, we develop a general theory of operational Chow groups of algebraic stacks which are locally of finite type over K. The theory is certainly known to experts, but for our later results, we will require the precise definitions. In particular, with a proper representable morphism of algebraic stacks, we associate an operational class, which will be the key to constructing the operational double ramification cycle.

The core of the paper starts in §3 where we give three equivalent definitions of the universal double ramification cycle on \mathfrak{Pic} . Our first definition is by taking a closure in the spirit of [38] which is simple, but rather difficult to work with. The second is via logarithmic geometry following [56]. The third is a b-Chow definition along the path of [39]. In §3.5, we give an explicit description of the set-theoretic image of the double ramification cycle in \mathfrak{Pic} . We prove Theorem 0.1 in §3.7.

In §4, we discuss properties of Pixton's cycle $\mathsf{P}_{g,A,d}^c$ defined in §0.3.5. In particular, formal properties of Pixton's cycle parallel to the invariances of the double ramification cycles are proven. Compatibilities with definitions in previously studied cases are also proven.

In §5, we prove Theorem 0.7, the main result of the paper, by an eventual reduction to the formula of [43] in the case of target \mathbb{P}^n for large n. A crucial step in the proof is the matching of the double ramification cycle defined in [43] via rubber maps with our new universal definition on \mathfrak{Pic} in a suitable sense when the target is \mathbb{P}^n . The matching is verified in §6, where we follow the pattern of the proof given in [56] in case the target is a point.

In $\S7$, we prove the invariance properties of \$0.6. Theorems 0.8 and 0.9 are proven as a consequence of Theorem 0.7 in \$8. The connections between the vanishing result of Theorem 0.8 and past (and future) work is discussed in \$8.5.

2. Picard stacks and operational Chow

2.1. The Picard stack and relative Picard space

Our stacks will be with respect to the fppf topology ([63, Definition 9.1.1]). We define the Picard stack $\mathfrak{Pic}_{g,n}$ as the fibered category over $\mathfrak{M}_{g,n}$ whose fiber over a scheme $T \to \mathfrak{M}_{g,n}$ is the groupoid of line bundles on $\mathfrak{C}_{g,n} \times_{\mathfrak{M}_{g,n}} T$ with morphisms given by isomorphisms of line bundles, see [51, Example 14.4.7]. We define the relative Picard space $\mathfrak{Pic}_{g,n}^{\mathrm{rel}}/\mathfrak{M}_{g,n}$ to be the quotient of $\mathfrak{Pic}_{g,n}$ by its relative inertia over $\mathfrak{M}_{g,n}$. Equivalently, $\mathfrak{Pic}_{g,n}^{\mathrm{rel}}$ is the fppf-sheaffication of the fibered category of *isomorphism classes* of line bundles on $\mathfrak{C}_{g,n} \times_{\mathfrak{M}_{g,n}} T$, see [14, Chapter 8] and [27, Chapter 9].

Relative representability of $\mathfrak{Pic}_{g,n}^{\mathrm{rel}}/\mathfrak{M}_{g,n}$ by smooth algebraic spaces can be checked locally on $\mathfrak{M}_{g,n}$. It then follows from [3, Appendix], as the curve

$$\mathfrak{C}_{g,n} \longrightarrow \mathfrak{M}_{g,n}$$

is flat, proper, relatively representable by algebraic spaces, and cohomologically flat in dimension zero (reduced and connected geometric fibers). The Picard stack $\mathfrak{Pic}_{g,n}$ is a \mathbb{G}_m -gerbe over $\mathfrak{Pic}_{g,n}^{\mathrm{rel}}$, and hence is a (smooth) algebraic stack. In particular, $\mathfrak{Pic}_{g,n}$ is smooth over K of pure dimension 4g-4+n, and $\mathfrak{Pic}_{g,n}^{\mathrm{rel}}$ is smooth over K of pure dimension 4g-4+n.

Remark 2.1. We will moreover assume $(g, n) \neq (1, 0)$. Then $\mathfrak{M}_{g,n}$, and hence $\mathfrak{Pic}^{\mathrm{rel}}$, \mathfrak{Pic} , and anything of Deligne–Mumford type over them, has affine stabilizers, and so is therefore stratified by global quotient stacks in the sense of [49]. The latter property will be important for some intersection-theoretic computations, in particular the proof of Proposition 2.16.

2.2. Operational Chow groups of algebraic stacks

Our goal here is to define the operational Chow group of $\mathfrak{Pic}_{g,n}$, following [29, Chapter 17]. In fact, we construct an operational Chow group for any algebraic stack locally of finite type over a field. The definition is a simple generalization of [29].

Definition 2.2. Let \mathfrak{Y} be a locally finite type algebraic stack over K. Let p be an integer. A bivariant class c in the pth operational Chow group $\mathsf{CH}^p_{\mathsf{op}}(\mathfrak{Y})$ is a collection of homomorphisms

$$c(\varphi)^m : \mathsf{CH}_m(B) \longrightarrow \mathsf{CH}_{m-p}(B)$$

for all maps $\varphi: B \to \mathfrak{Y}$ where B is a scheme of finite type over K, and for all integers m, compatible with proper pushforward, flat pullback, and Gysin homomorphisms for regular embeddings (conditions (C1)–(C3) in [29, §17.1]).

For a class $\alpha \in CH_m(B)$, we will sometimes write $c(\alpha)$ in place of $c(\varphi)^m(\alpha)$, if the morphism φ is clear.

Such a definition for the operational Chow group of a Deligne–Mumford stack is given in [24]. To be able to use Chow groups on algebraic stacks as defined in [49] for algebraic stacks of finite type over a field, we will use the following result.

LEMMA 2.3. Let $f: \mathfrak{A} \to \mathfrak{B}$ be a morphism over a field K, where \mathfrak{B} is an algebraic stack locally of finite type over K, and \mathfrak{A} is an algebraic stack of finite type over K. Then, there exists a factorization of f via a commutative diagram



where $\mathfrak{B}' \to \mathfrak{B}$ is an open immersion and \mathfrak{B}' is quasi-compact (and hence of finite type).

Proof. We cover \mathfrak{B} by affine flat finite presentation morphisms $\{V_i \to \mathfrak{B}\}_{i \in I}$. Let U_i be the image of V_i in \mathfrak{B} , for all i. The U_i are open, and the $f^{-1}(U_i)$ cover \mathfrak{A} . As \mathfrak{A} is quasi-compact, there is a finite subset $J \subset I$ such that $\{f^{-1}(U_i)\}_{i \in J}$ covers \mathfrak{A} . Then,

$$\mathfrak{B}' = \bigcup_{i \in J} U_i$$

defines the required factorization.

For representable morphisms (representable by algebraic spaces) $f: \mathfrak{X} \to \mathfrak{Y}$, we can also define the operational Chow group $CH^p_{op}(\mathfrak{X} \to \mathfrak{Y})$ as a collection of morphisms

$$c(\varphi)^m : \operatorname{CH}_m(B) \longrightarrow \operatorname{CH}_{m-p}(B \times_{\mathfrak{Y}} \mathfrak{X})$$

for all maps $\varphi: B \to \mathfrak{Y}$, where B is a scheme of finite type over K, and for all integers m, compatible with proper pushforward, flat pullback, and Gysin homomorphisms. We have

$$\mathsf{CH}^p_{\mathsf{op}}(\mathfrak{X}) = \mathsf{CH}^p_{\mathsf{op}}(\mathrm{id}: \mathfrak{X} \to \mathfrak{X}).$$

We have products, pullbacks, and proper representable pushforwards on these operational Chow groups of algebraic stacks as described in [29, §17.2] satisfying the properties described there.

Remark 2.4. Even for B a scheme and $\pi: \mathfrak{X} \to \mathfrak{Y}$ representable, the fiber product $B \times_{\mathfrak{Y}} \mathfrak{X}$ can be an algebraic space. Therefore, some care is needed when generalizing classical constructions such as the product

$$\mathsf{CH}^a_{\mathsf{op}}(\mathfrak{X}) \!\times\! \mathsf{CH}^b_{\mathsf{op}}(\pi \colon\! \mathfrak{X} \!\rightarrow\! \mathfrak{Y}) \longrightarrow \mathsf{CH}^{a+b}_{\mathsf{op}}(\pi \colon\! \mathfrak{X} \!\rightarrow\! \mathfrak{Y}).$$

Indeed, for $c \in \mathsf{CH}^a_{\mathsf{op}}(\mathfrak{X})$ and $d \in \mathsf{CH}^b_{\mathsf{op}}(\pi: \mathfrak{X} \to \mathfrak{Y})$, we want to define

$$c \cdot d \in \mathsf{CH}^{a+b}_{\mathsf{op}}(\pi: \mathfrak{X} \to \mathfrak{Y}).$$

For a map $\varphi: B \to \mathfrak{Y}$ with B a finite-type scheme fitting in a pullback diagram

we want to define the induced map

$$(c \cdot d)(\varphi)^m \colon \mathsf{CH}_m(B) \longrightarrow \mathsf{CH}_{m-a-b}(B \times_\mathfrak{Y} \mathfrak{X})$$

as the composition

$$\mathsf{CH}_m(B) \xrightarrow{d(\varphi)^m} \mathsf{CH}_{m-b}(B \times_{\mathfrak{Y}} \mathfrak{X}) \xrightarrow{c(\varphi')^{m-b}} \mathsf{CH}_{m-a-b}(B \times_{\mathfrak{Y}} \mathfrak{X}).$$

But, a priori, the map $c(\varphi')^{m-b}$ does not make sense, since the domain $B \times_{\mathfrak{Y}} \mathfrak{X}$ of φ' is an algebraic space. However, given a collection

$$c = c(\varphi')^n : \operatorname{CH}_n(B') \longrightarrow \operatorname{CH}_{n-a}(B')$$

for all finite-type schemes B' with $\varphi': B' \to \mathfrak{X}$, we can construct a collection of maps

$$c(\varphi')^n : \operatorname{CH}_n(B') \longrightarrow \operatorname{CH}_{n-a}(B')$$

for $\varphi': B' \to \mathfrak{X}$, with B' a finite-type algebraic space via [76, §5.1]. Indeed, for each integral closed substack $Z \subset B'$, [76, §5.1] defines an action of c on [Z] which is independent of the chosen cover of the algebraic space by a scheme. This action induces a map

$$\mathsf{CH}_n(B') \longrightarrow \mathsf{CH}_{n-a}(B'),$$

which commutes with proper pushforward and flat morphisms via [76, Lemma 5.3] and is compatible with Gysin homomorphisms. Applying this to

$$\varphi' \colon B' = B \times_{\mathfrak{Y}} \mathfrak{X} \longrightarrow \mathfrak{X}$$

with n=m-b gives the desired map $c(\varphi')^{m-b}$.

For a proper representable morphism $\pi: \mathfrak{X} \to \mathfrak{Y}$ which is flat of relative dimension q, the pushforward

$$\mathsf{CH}^*_{\mathsf{op}}(\pi \colon \mathfrak{X} \to \mathfrak{Y}) \longrightarrow \mathsf{CH}^*_{\mathsf{op}}(\mathfrak{Y})$$

can be extended to a pushforward

$$\mathrm{CH}^p_{\mathrm{op}}(\mathfrak{X}) \mathop{\longrightarrow} \mathrm{CH}^{p-q}_{\mathrm{op}}(\mathfrak{Y})$$

as follows. Because π is flat, the pullback π^* gives a natural element in

$$\mathsf{CH}_{\mathsf{op}}^{-q}(\pi {:}\, \mathfrak{X} \,{\to}\, \mathfrak{Y}),$$

and then we can compose

$$\mathsf{CH}^p_{\mathsf{op}}(\mathfrak{X}) \longrightarrow \mathsf{CH}^p_{\mathsf{op}}(\mathfrak{X}) \!\times\! \mathsf{CH}^{-q}_{\mathsf{op}}(\pi \!:\! \mathfrak{X} \!\rightarrow\! \mathfrak{Y})$$

given by $c \mapsto (c, \pi^*)$ with the product and the pushforward maps

$$\mathsf{CH}^p_{\mathsf{op}}(\mathfrak{X}) \times \mathsf{CH}^{-q}_{\mathsf{op}}(\pi : \mathfrak{X} \to \mathfrak{Y}) \longrightarrow \mathsf{CH}^{p-q}_{\mathsf{op}}(\pi : \mathfrak{X} \to \mathfrak{Y}) \longrightarrow \mathsf{CH}^{p-q}_{\mathsf{op}}(\mathfrak{Y}),$$

yielding the desired pushforward map

$$\pi_*: \mathrm{CH}^p_{\mathrm{op}}(\mathfrak{X}) \longrightarrow \mathrm{CH}^{p-q}_{\mathrm{op}}(\mathfrak{Y}).$$

This may for example be applied to the universal curve $\pi: \mathfrak{C}_{g,n} \to \mathfrak{Pic}_{g,n}$. A similar construction also works for π proper representable and lci. This pushforward map commutes with pullback of operational classes.

2.3. Relationship to usual Chow groups

Let \mathfrak{Y} be a locally finite type algebraic stack over K, and $(\mathfrak{U}_i)_{i \in \mathbb{N}}$ an increasing sequence of finite-type open substacks of \mathfrak{Y} with

$$\bigcup_{i}\mathfrak{U}_{i}=\mathfrak{Y}.$$

In particular, for a finite-type scheme B/K, every map $B \to \mathfrak{Y}$ factors via some \mathfrak{U}_i . We have pullback maps

$$\operatorname{CH}^*_{\operatorname{op}}(\mathfrak{Y}) \longrightarrow \operatorname{CH}^*_{\operatorname{op}}(\mathfrak{U}_i)$$
 (2.1)

which induce a map

$$\Phi: \mathsf{CH}^*_{\mathsf{op}}(\mathfrak{Y}) \longrightarrow \lim_{i} \mathsf{CH}^*_{\mathsf{op}}(\mathfrak{U}_i)$$

$$(2.2)$$

to the inverse limit of the $\mathsf{CH}^*_{\mathsf{op}}(\mathfrak{U}_i)$, with transition maps given by pullback of operational classes along open immersions.

LEMMA 2.5. The map

$$\Phi: \mathsf{CH}^*_{\mathsf{op}}(\mathfrak{Y}) \longrightarrow \lim_{i} \mathsf{CH}^*_{\mathsf{op}}(\mathfrak{U}_i)$$

is an isomorphism of abelian groups.

Proof. We first show injectivity. Let c in $\mathsf{CH}^*_{\mathsf{op}}(\mathfrak{Y})$ with $\Phi(c)=0$. For every $B \to \mathfrak{Y}$ with B/K of finite type, we get a map

$$c(B/\mathfrak{Y}): \mathsf{CH}_*(B) \longrightarrow \mathsf{CH}_*(B).$$

There exists i such that the map $B \rightarrow \mathfrak{Y}$ factors via \mathfrak{U}_i . Then, $\Phi(c) = 0$ implies that

$$c(B/\mathfrak{U}_i): \mathsf{CH}_*(B) \longrightarrow \mathsf{CH}_*(B)$$

is the zero map. By definition of the pullback,

$$c(B/\mathfrak{Y}) = c(B/\mathfrak{U}_i).$$

Next, we show surjectivity. Suppose that we have a compatible collection

$$c_i \in \mathsf{CH}^*_{\mathsf{op}}(\mathfrak{U}_i).$$

We will build $c \in \mathsf{CH}^*_{op}(\mathfrak{Y})$ as follows. Let $B \to \mathfrak{Y}$ with B/K of finite type. There exists N such that, for all $i \ge N$, the map $B \to \mathfrak{Y}$ factors via \mathfrak{U}_i . Then, for all $i \ge N$, we have maps

$$c(B/\mathfrak{U}_i)$$
: $CH_*(B) \longrightarrow CH_*(B),$

and the compatibly means $c_i(B/\mathfrak{U}_i)=c_j(B/\mathfrak{U}_i)$ for all $i,j \ge N$. We define $c=c_N$, which clearly is sent by Φ to the c_i .

To conclude, we must check that c satisfies the axioms of an operational class. This follows easily from the fact that each $B \rightarrow \mathfrak{Y}$ factors via some \mathfrak{U}_i .

LEMMA 2.6. ([76, Proposition 5.6]) Let \mathfrak{Y} be a smooth finite-type Deligne-Mumford stack over K of pure dimension d, and let $\iota: \mathfrak{Y} \to \mathfrak{Y}$ be the identity. Then, for $m \ge 0$, the map

$$\mathsf{CH}^{m}_{\mathsf{op}}(\mathfrak{Y}) \longrightarrow \mathsf{CH}_{d-m}(\mathfrak{Y}),$$

$$\alpha \longmapsto \alpha(\iota)([\mathfrak{Y}]),$$
(2.3)

is an isomorphism.

232

Combining Lemmas 2.5 and 2.6, we immediately obtain the following result.

COROLLARY 2.7. Let \mathfrak{Y} be a smooth Deligne–Mumford stack over K of pure dimension d, and let \mathfrak{U}_i be a sequence of finite-type open substacks with $\bigcup_{i \in I} \mathfrak{U}_i = \mathfrak{Y}$. Then, the natural map

$$\Phi: \mathsf{CH}^m_{\mathsf{op}}(\mathfrak{Y}) \longrightarrow \lim_{i \in I} \mathsf{CH}_{d-m}(\mathfrak{U}_i)$$
(2.4)

obtained by combining (2.2) and (2.3) is an isomorphism.

As a final remark, $(^{14})$ we note that there exists a map of Chow groups in the opposite direction of (2.3) in greater generality. Let \mathfrak{Y} be a smooth algebraic stack of finite type over K and of pure dimension d which has a stratification $(^{15})$ by quotient stacks. Then, there exists a map

$$\Psi: \mathsf{CH}_{d-m}(\mathfrak{Y}) \longrightarrow \mathsf{CH}^m_{\mathsf{op}}(\mathfrak{Y})$$

from the Chow group $\mathsf{CH}_*(\mathfrak{Y})$ constructed in [49] to the operational Chow group of \mathfrak{Y} defined as follows. Given $\varphi: B \to \mathfrak{Y}$ with B a finite-type scheme, let

$$\varphi_B: B \longrightarrow B \times \mathfrak{Y}$$

be the diagonal morphism. Since \mathfrak{Y} is smooth, φ_B is representable and is a local complete intersection of codimension d. For $\beta \in CH_{d-m}(\mathfrak{Y})$, we define

$$\begin{split} \Psi(\beta)(\varphi) \colon \mathsf{CH}_*(B) &\longrightarrow \mathsf{CH}_{*-m}(B), \\ \alpha &\longmapsto \varphi_B^!(\alpha \times \beta), \end{split}$$

where $\alpha \times \beta \in \mathsf{CH}_{*+d-m}(B \times \mathfrak{Y})$ is the exterior product of α and β as defined in [49, §3.2]. The collection of maps $\Psi(\beta)(\varphi)$ defines an element

$$\Psi(\beta) \in \mathsf{CH}^m_{\mathsf{op}}(\mathfrak{Y}).$$

For \mathfrak{Y} a Deligne–Mumford stack, the map Ψ is the inverse of the map (2.3). However, for an arbitrary smooth algebraic stack \mathfrak{Y} , we do *not* know whether Ψ is injective or surjective.

^{(&}lt;sup>14</sup>) We thank A. Kresch for related discussion.

^{(&}lt;sup>15</sup>) See [49] for the precise definition. The property is always satisfied for Deligne–Mumford stacks.

2.4. Constructing an operational Chow class

2.4.1. Overview

Given a vector $A \in \mathbb{Z}^n$ of ramification data satisfying

$$\sum_{i=1}^{n} a_i = d,$$

we will construct in §3.3 a stack $\mathbf{Div}_{g,A}$ together with a proper representable Abel–Jacobi map

$$\operatorname{Div}_{g,A} \longrightarrow \operatorname{\mathfrak{Pic}}_{g,n,d}.$$

We wish to define the twisted universal double ramification cycle $\mathsf{DR}_{g,A}^{\mathsf{op}}$ as the pushforward of the fundamental class of $\mathbf{Div}_{g,A}$ to $\mathfrak{Pic}_{g,n,d}$. However, two basic issues must be settled to carry out the construction:

• The stack $\operatorname{Div}_{g,A}$ is not Deligne–Mumford and is not quasi-compact, so the existence of a well-behaved fundamental class in the operational Chow group is not clear.

• The proper representable pushforward of [6, Appendix B] is only defined between finite-type-stacks, and so cannot be applied directly.⁽¹⁶⁾

To solve these problems, we provide here a very general construction which associates with a suitable proper morphism

$$a: \mathfrak{X} \longrightarrow \mathfrak{Y}$$

an operational Chow class on \mathfrak{Y} which plays the role of the pushforward of the fundamental class of \mathfrak{X} . In §3.3, we will apply the result to construct the universal twisted double ramification cycle $\mathsf{DR}_{g,A}^{\mathsf{op}}$. We also verify certain basic properties such as invariance of the class under proper birational maps which will be important in §6.

2.4.2. Construction

Let \mathfrak{X} and \mathfrak{Y} be algebraic stacks locally of finite type over a field K and suppose we have a proper morphism

$$a: \mathfrak{X} \longrightarrow \mathfrak{Y}$$

of Deligne–Mumford type. Suppose further that \mathfrak{Y} is smooth of pure dimension dim \mathfrak{Y} over the field, and \mathfrak{X} is of pure dimension dim \mathfrak{X} . Let

$$e = \dim \mathfrak{Y} - \dim \mathfrak{X}.$$

We will construct an operational Chow class associated with \mathfrak{X} in $\mathsf{CH}^{e}_{\mathsf{op}}(\mathfrak{Y})$.

^{(&}lt;sup>16</sup>) Chow theory of non-finite-type algebraic stacks will be developed in [6, Appendix A].

For all finite-type schemes B with a morphism $\varphi: B \to \mathfrak{Y}$, and for all integers m, we must define maps

$$c(\varphi)^m : \mathsf{CH}_m(B) \longrightarrow \mathsf{CH}_{m-e}(B),$$

which are compatible under proper pushforward and flat pullback and satisfy commutativity (properties (C1)-(C3) of $[29, \S17.1]$).

Let $[V] \in CH_m(B)$ be an irreducible cycle, and let $i_V: V \to B$ be the inclusion. Let $V \to B \to \mathfrak{Y}$ be factored as in Lemma 2.3 into $V \to \mathfrak{Y}' \to \mathfrak{Y}$, where \mathfrak{Y}' is of finite type and $\mathfrak{Y}' \to \mathfrak{Y}$ is an open immersion.

We form the diagram

$$\begin{aligned}
\mathfrak{X}' \times_{\mathfrak{Y}'} V &\longrightarrow \mathfrak{X}' \times V \\
\downarrow_{\psi_V} & \downarrow_{a \times \mathrm{id}} \\
V & \stackrel{\varphi_V}{\longrightarrow} \mathfrak{Y}' \times V,
\end{aligned}$$
(2.5)

where \mathfrak{X}' is the inverse image of \mathfrak{Y}' under *a*. Each stack in this diagram is of finite type, and therefore has a Chow group in the sense of [49]. Since \mathfrak{Y} is smooth over *K*, the map φ_V is a regular embedding of codimension dim \mathfrak{Y} . Also φ_V is unramified, and hence a regular local embedding, so Kresch's contruction yields a map

$$\varphi_V^!: \mathsf{CH}_m(\mathfrak{X}' \times V) \longrightarrow \mathsf{CH}_{m-\dim \mathfrak{Y}}(\mathfrak{X}' \times_{\mathfrak{Y}'} V).$$

In particular, $[\mathfrak{X}' \times V]$ is a class in dimension dim $\mathfrak{X}+m$, so the class $\varphi_V^!([\mathfrak{X} \times V])$ lies in $\mathsf{CH}_{m-e}(\mathfrak{X} \times \mathfrak{Y})$. The morphism ψ_V is proper and of Deligne–Mumford type, and so by [6, Appendix B] we have a pushforward ψ_{V*} .

Definition 2.8. We define a class $a_{op}[\mathfrak{X}] \in \mathsf{CH}_{op}(\mathfrak{Y})$ via the formula

$$c(\varphi)^{m}: Z_{m}(B) \longrightarrow \mathsf{CH}_{m-e}(B),$$
$$[V] \longmapsto i_{V*} \psi_{V*} \varphi_{V}^{!}([\mathfrak{X}' \times V]).$$

We must verify that this construction passes to rational equivalence, is independent of the choices made, and satisfies the properties (C1)-(C3). After verifying in Lemma 2.9 independence on the choice of factorization, we follow the logic in [29]: we verify in Lemmas 2.11–2.13 of §2.4.3 that the properties (C1)-(C3) hold on the level of cycles, and finally in Lemma 2.15 we use these to show that the construction passes to rational equivalence.

LEMMA 2.9. The class $c(\varphi)^m([V])$ defined above is independent of the chosen factorization $V \to \mathfrak{Y}' \to \mathfrak{Y}$. *Proof.* Let $V \to \mathfrak{Y}' \to \mathfrak{Y}$ and $V \to \mathfrak{Y}'' \to \mathfrak{Y}$ be two such factorizations. By considering $\mathfrak{Y}' \cap \mathfrak{Y}''$ inside \mathfrak{Y} , we may restrict to the case where one is contained in the other. So we suppose there is an open immersion

$$\iota: \mathfrak{Y}'' \longrightarrow \mathfrak{Y}'.$$

Let $j: \mathfrak{X}'' \to \mathfrak{X}'$ be the induced map. Consider the diagram

where $\mathfrak{X}'' \times_{\mathfrak{Y}'} V \to \mathfrak{X}' \times_{\mathfrak{Y}'} V$ is an isomorphism. We must show that

$$\psi'_{V*}(\varphi'_V)^!([\mathfrak{X}' \times V]) = \psi''_{V*}(\varphi''_V)^!([\mathfrak{X}'' \times V]).$$

Because φ_V'' and φ_V' are both regular embeddings of the same codimension and the front square commutes, by the same proof as for [29, Theorem 6.2 (c)], we obtain

$$(\varphi_V'')!([\mathfrak{X}'' \times V]) = (\varphi_V')!([\mathfrak{X}'' \times V]).$$

Therefore,

$$\psi_{V*}''(\varphi_V'')!([\mathfrak{X}'' \times V]) = \psi_{V*}''(\varphi_V')!([\mathfrak{X}'' \times V]) = \psi_{V*}''(\varphi_V')!((j \times \mathrm{id})^*[\mathfrak{X}' \times V]),$$
(2.7)

since $j \times id$ is an open immersion, so in particular flat, and the flat pullback of the fundamental class is the fundamental class itself. By the compatibility of the flat pullback and Gysin maps [49, §3.1], we obtain that (2.7) is equal to

$$\psi_{V*}''(j \times \mathrm{id})^*(\varphi_V')!([\mathfrak{X}' \times V]).$$

By commutativity of the left side of the cube and because of the pullback pushforward formula, we obtain

$$\psi_{V*}''(\varphi_V'')^!([\mathfrak{X}'' \times V]) = \psi_{V*}''(j \times \mathrm{id})^*(\varphi_V')^!([\mathfrak{X}' \times V]) = (\psi_V')_*(\varphi_V')^!([\mathfrak{X}' \times V])$$

as required.

236 Y. BAE, D. HOLMES, R. PANDHARIPANDE, J. SCHMITT AND R. SCHWARZ

2.4.3. Compatibility

We will now check that the maps $c(\varphi)^m$ defined in §2.4.2 are compatible under proper pushforward and flat pullback, and satisfy commutativity (properties (C1)–(C3) of [29, §17.1]).

The proper pushforward along DM-type maps between finite-type algebraic stacks over a field which are stratified by global quotient stacks is defined in [6, Appendix B]. We cannot use the results of [49] for compatibility with the proper pushforward, since Kresch discusses only projective pushforward. Nevertheless, we now show that the proper pushforward for DM-type maps of finite-type algebraic stacks over a field as defined in [6, Appendix B] is compatible with the Gysin maps of [49].

PROPOSITION 2.10. For a pullback diagram of algebraic stacks of finite type over K,



where *i* is a regular local embedding of codimension *d*, *p* is a proper DM-type morphism, and \mathfrak{Y}' is stratified by global quotient stacks, we have

$$i^! p_*(\alpha) = q_*(i^!\alpha)$$

for all $\alpha \in CH_*(\mathfrak{Y}'')$.

Proof. A class α on \mathfrak{Y}'' is represented by a projective map $z'': Z'' \to \mathfrak{Y}''$, a vector bundle $E'' \to Z''$ and a class [V] in the naive Chow group of E'' represented by $V \subset E$.

We want to push α forward via the construction of [6, Appendix B]: by [6, Theorem B.17], it suffices to treat the case where $E'' \rightarrow Z'' \rightarrow \mathfrak{Y}''$ fits in a pullback diagram of the form

where z' is projective. Then, the pushforward is defined by simply pushing forward on the level of bundles, so

$$p_*(z'', [V]) = (z', p''_*([V])).$$

237

If W = p''(V), then $p''_{*}([V]) = \deg(V/W)[W]$.

Let N be the normal bundle $N_{\mathfrak{X}}\mathfrak{Y}$. The Gysin maps constructed in [49, §3.1] are described explicitly on level of representatives as follows: i'(z', [W]) is represented by $[C_{W \times \mathfrak{Y}}\mathfrak{X}W]$ as a class on the bundle

$$N|_{Z' \times_{\mathfrak{Y}} \mathfrak{X}} \oplus E'|_{Z' \times_{\mathfrak{Y}} \mathfrak{X}}$$

with the projective map $\tilde{z}': Z' \times_{\mathfrak{Y}} \mathfrak{X} \to \mathfrak{X}'$ induced by z'. Hence,

$$i^! p_*(z'', [V]) = (\tilde{z}', \deg(V/W)[C_{W \times_{\mathfrak{Y}} \mathfrak{X}} W]).$$

Next, we study $q_*i^!(z'', [V])$. To start, $i^!(z'', [V])$ is represented by $[C_{V \times_{\mathfrak{Y}} \mathfrak{X}}V]$ as class on the bundle

$$N|_{Z'' \times_{\mathfrak{Y}} \mathfrak{X}} \oplus E''|_{Z'' \times_{\mathfrak{Y}} \mathfrak{X}}$$

with the projective map $Z'' \times_{\mathfrak{Y}} \mathfrak{X} \to \mathfrak{X}''$ induced by z''. We push $i^!(z'', [V])$ forward via the construction of [6, Appendix B]. We complete the diagram

to a pullback diagram so that we can pushforward on the levels of bundles:



The map $p': Z'' \to Z'$ induces a map $\tilde{p}': Z'' \times_{\mathfrak{Y}} \mathfrak{X} \to Z' \times_{\mathfrak{Y}} \mathfrak{X}$, and the square with $q, \tilde{p}', \tilde{z}'$, and \tilde{z}'' is a pullback (as the pullback of a pullback square).

There is a map

$$q'': N|_{Z'' \times_{\mathfrak{Y}} \mathfrak{X}} \oplus E''|_{Z'' \times_{\mathfrak{Y}} \mathfrak{X}} \longrightarrow N|_{Z' \times_{\mathfrak{Y}} \mathfrak{X}} \oplus E'|_{Z' \times_{\mathfrak{Y}} \mathfrak{X}}$$

induced by p'' such that it forms a pullback square, and so the pushforward along q is simply

$$q_*(i^!(z'', [V])) = q_*(\tilde{z}'', [C_{V \times_{\mathfrak{Y}} \mathfrak{X}}V]) = (\tilde{z}', q_*''([C_{V \times_{\mathfrak{Y}} \mathfrak{X}}V]))$$

The proof then reduces to comparing $q''_*([C_{V \times_{\mathfrak{Y}}\mathfrak{X}}V])$ and $\deg(V/W)([C_{W \times_{\mathfrak{Y}}\mathfrak{X}}W])$, which follows from [76, Lemma 3.15]. The result is a completely local statement and therefore extends from the setting of schemes to the setting of Deligne–Mumford stacks which we need here.

Let $\varphi: B \to \mathfrak{Y}$ and $\varphi': B' \to \mathfrak{Y}$ be morphisms from finite-type schemes, and let

$$h: B' \longrightarrow B$$

be a \mathfrak{Y} -morphism. By Definition 2.8, we have morphisms

$$c(\varphi)^m : Z_m(B) \longrightarrow \mathsf{CH}_{m-e}(B) \text{ and } c(\varphi')^m : Z_m(B') \longrightarrow \mathsf{CH}_{m-e}(B').$$

If h is proper, we have a pushforward map $\binom{17}{1}$

$$h_*: Z_m(B') \longrightarrow Z_m(B)$$

which descends to Chow.

LEMMA 2.11. If h is proper, then

$$c(\varphi)^m \circ h_* = h_* \circ c(\varphi h)^m.$$

Proof. Let $[V'] \in Z_m(B')$ for an irreducible cycle V'. Let V = h(V'). By definition,

 $h_*([V']) = \deg(V'/V)[V].$

Let \mathfrak{Y}' be a factorization of $V' \rightarrow V \rightarrow \mathfrak{Y}$. We have

$$(\operatorname{id} \times h)_*([\mathfrak{X}' \times V']) = \operatorname{deg}(V'/V)[\mathfrak{X}' \times V].$$

Via the commutative diagram



 $(^{17})$ We use the same notation for the proper pushforward on the Chow groups.

we compute

$$c(\varphi)^{m}(h_{*}(\alpha)) = c(\varphi)^{m}(\deg(V'/V)[V])$$

$$= i_{V*}\psi_{V*}\varphi_{V}^{!}(\deg(V'/V)[\mathfrak{X}'\times V])$$

$$= i_{V*}\psi_{V*}\varphi_{V}^{!}((\operatorname{id} \times h)_{*}[\mathfrak{X}'\times V'])$$

$$= h_{*}i_{V'*}\psi_{V'*}\varphi_{V'}^{!}([\mathfrak{X}'\times V'])$$

$$= h_{*}c(\varphi h)^{m}(\alpha),$$

where the final line follows from compatibility of the Gysin map with proper representable pushforward (Proposition 2.10). $\hfill \Box$

If $h: B' \to B$ is flat of relative dimension n, we have a pullback map

$$h^*: Z_m(B) \longrightarrow Z_{m+n}(B')$$

which descends to Chow.

LEMMA 2.12. If h is flat of relative dimension n,

$$c(\varphi h)^{m+n} \circ h^* = h^* \circ c(\varphi)^m.$$

Proof. Let $[V] \in Z_m(B)$ for an irreducible cycle V. Let

$$V' = h^{-1}(V),$$

so $[V'] = h^*([V])$. Let \mathfrak{Y}' be a factorization of $V' \to V \to \mathfrak{Y}$. We have

$$(\operatorname{id} \times h)^*([\mathfrak{X}' \times V]) = [\mathfrak{X}' \times V']$$

Via the commutative diagram (2.9), we compute

$$\begin{aligned} c(\varphi h)^{m+n}([V']) &= i_{V'*}\psi_{V'*}\varphi_{V'}^!([\mathfrak{X}' \times V']) \\ &= i_{V'*}\psi_{V'*}\varphi_{V'}^!((\mathrm{id} \times h)^*[\mathfrak{X}' \times V]) \\ &= h^*i_{V*}\psi_{V*}\varphi_V^!([\mathfrak{X}' \times V]) \\ &= h^*c(\varphi)^m([V]), \end{aligned}$$

where the final equality follows from the compatibility of the Gysin map with flat pullback for a morphism of finite-type algebraic stacks ([49, $\S3.1$]) and the pullback and pushforward formulas.
Let $\varphi: B \to \mathfrak{Y}$ be a morphism from a finite-type scheme as above. Let

$$g: B \longrightarrow Z$$

be a morphism of finite-type schemes, and let $i: Z' \to Z$ be a regular embedding of codimension f. Form the fiber square



Let V be an irreducible cycle in B with inverse image $V' = (i')^{-1}(V)$. We choose a representative $i^![V] = \sum_j n_j[V'_j]$ in $Z_{m-f}(V')$.

LEMMA 2.13. We have

$$i^{!}c(\varphi)^{m}([V]) = c(\varphi i')^{m-f}\left(\sum_{j} n_{j}[V'_{j}]\right).$$

Remark 2.14. In particular, once we have shown that the maps $c(\varphi)^m$ pass to rational equivalence, Lemma 2.13 will imply

$$i^{!}c(\varphi)^{m}(\alpha) = c(\varphi i')^{m-f}(i^{!}\alpha)$$

for $\alpha \in \mathsf{CH}_m(B)$.

Proof. From the equality $i'[V] = \sum n_j[V'_j]$, we deduce that

$$i^! [\mathfrak{X}' \times V] = \sum n_j [\mathfrak{X}' \times V'_j].$$

Via the commutative diagram $(^{18})$



 $(^{18})$ We should also add another layer of the diagram for the $V_j^\prime.$

we compute

$$i^! c(\varphi)^m([V]) = i^! i_{V*} \psi_{V*} \varphi_V^!([\mathfrak{X}' \times V])$$

and

$$c(\varphi i')^{m-f}\left(\sum_{j} n_{j}[V'_{j}]\right) = \sum n_{j} i_{V'_{j}} \psi_{V'_{j}} \varphi^{!}_{V'_{j}}([\mathfrak{X}' \times V'_{j}]) = i_{V'*} \psi_{V'*} \varphi^{!}_{V'}(i^{!}[\mathfrak{X}' \times V]).$$

We deduce equality of these expressions by using the compatibility of Gysin maps with proper pushforward (Proposition 2.10) and then the commutativity of Gysin maps from [49] to obtain $\varphi_{V'}^! i^! = i^! \varphi_{V}^!$.

LEMMA 2.15. The morphisms $c(\varphi)^m$ from Definition 2.8 pass to rational equivalence,

$$c(\varphi)^m \colon \mathsf{CH}_m(B) \longrightarrow \mathsf{CH}_{m-e}(B).$$

Proof. The proof is now completely analogous to [29, Theorem 17.1].

2.4.4. Properties

The class constructed in Definition 2.8 is invariant under proper birational maps in the following sense.

PROPOSITION 2.16. Let $f: \mathfrak{W} \to \mathfrak{X}$ be a proper DM-type birational morphism of locally-finite-type algebraic stacks over K of pure dimension, with \mathfrak{X} stratified by global quotient stacks. Then,

$$(a \circ f)_{\mathrm{op}}[\mathfrak{W}] = a_{\mathrm{op}}[\mathfrak{X}] \in \mathsf{CH}^{e}_{\mathsf{op}}(\mathfrak{Y})$$

where $(a \circ f)_{\text{op}}[\mathfrak{W}]$ and $a_{\text{op}}[\mathfrak{X}]$ are the operational classes constructed in Definition 2.8 with respect to $a \circ f: \mathfrak{W} \rightarrow \mathfrak{Y}$ and $a: \mathfrak{X} \rightarrow \mathfrak{Y}$.

Proof. The proper pushforward of the fundamental class along f is the fundamental class, as the map f is birational and hence of degree 1.

Choose a factorization $V \to \mathfrak{Y}' \to \mathfrak{Y}$. Denote by \mathfrak{X}' the pullback of \mathfrak{X} along a and by \mathfrak{W}' the the pullback of \mathfrak{X}' along f. As in (2.5), we form a pullback diagram

$$\mathfrak{W}' \times_{\mathfrak{Y}'} V \longrightarrow \mathfrak{W}' \times V \\
\downarrow_{\tilde{f}} \qquad \qquad \downarrow_{f \times \mathrm{id}} \\
\mathfrak{X}' \times_{\mathfrak{Y}'} V \longrightarrow \mathfrak{X}' \times V \\
\downarrow_{\psi_{V}} \qquad \qquad \downarrow_{a \times \mathrm{id}} \\
V \xrightarrow{\varphi_{V}} \mathfrak{Y}' \times V.$$
(2.11)

241

Proposition 2.10 then yields

$$\begin{split} \psi_{V*}\varphi_{V}^{!}([\mathfrak{X}' \times V]) &= \psi_{V*}\varphi_{V}^{!}((f \times \mathrm{id})_{*}[\mathfrak{W}' \times V]) \\ &= \psi_{V*}\tilde{f}_{*}\varphi_{V}^{!}([\mathfrak{W}' \times V]) \\ &= (\psi_{V}\tilde{f})_{*}\varphi_{V}^{!}([\mathfrak{W}' \times V]), \end{split}$$

which is the required equality.

We will also require a flat pullback property. Let \mathfrak{X} , \mathfrak{Y} , and \mathfrak{Z} be pure-dimensional algebraic stacks of locally finite type over K, with \mathfrak{Y} and \mathfrak{Z} smooth. Suppose that we have a fiber diagram

$$\begin{array}{c} \mathfrak{X} \times_{\mathfrak{Y}} \mathfrak{Z} \overset{\tilde{a}}{\longrightarrow} \mathfrak{Z} \\ \downarrow_{\tilde{f}} & \downarrow_{f} \\ \mathfrak{X} \overset{a}{\longrightarrow} \mathfrak{Y}, \end{array}$$

where $a: \mathfrak{X} \to \mathfrak{Y}$ is a proper DM-type morphism, and $f: \mathfrak{Z} \to \mathfrak{Y}$ is flat and $\mathrm{lci}_{(1^9)}$ with \mathfrak{Z} stratified by global quotient stacks.

LEMMA 2.17. In $CH_{op}(\mathfrak{Z})$, we have

$$\tilde{a}_{\mathrm{op}}[\mathfrak{X} \times_{\mathfrak{Y}} \mathfrak{Z}] = f^* a_{\mathrm{op}}[\mathfrak{X}].$$

Proof. Let $\mathfrak{X}', \mathfrak{Y}'$, and \mathfrak{Z}' denote appropriate finite-type factorizations as in Definition 2.8. Then, we compare the two operational classes via the following diagram:



By definition,

$$\tilde{a}_{\rm op}[\mathfrak{X} \times_{\mathfrak{Y}} \mathfrak{Z}](\varphi)([V]) = i_{V*} \psi_{V*} \varphi_V^! ([\mathfrak{X}' \times_{\mathfrak{Y}'} \mathfrak{Z}' \times V])$$

and

$$f^*a_{\mathrm{op}}[\mathfrak{X}](\varphi)([V]) = a_{\mathrm{op}}[\mathfrak{X}](f \circ \varphi)([V]) = i_{V*}\psi_{V*}(f \circ \varphi)_V^!([\mathfrak{X}' \times V]).$$

 $^(^{19})$ A flat and lci map is called *syntomic*.

Since f is lci, the above expression is equal to

$$i_{V*}\psi_{V*}\varphi_V^!(f\times \mathrm{id})^!([\mathfrak{X}'\times V]).$$

Because f is also flat, we see, as in [29, Proposition 6.6 (b)] that we obtain

$$i_{V*}\psi_{V*}\varphi_V^!(\tilde{f}\times \mathrm{id})^*([\mathfrak{X}'\times V]) = i_{V*}\psi_{V*}\varphi_V^!([\mathfrak{X}'\times_{\mathfrak{Y}'}\mathfrak{Z}'\times V]),$$

which yields the required equality.

3. The universal double ramification cycle

3.1. Overview

We fix a genus g, a number of markings n, and a vector $A = (a_1, ..., a_n) \in \mathbb{Z}^n$ of ramification data satisfying

$$\sum_{i=1}^{n} a_i = d.$$

We define here the associated universal twisted double ramification cycle class in the operational Chow group of the universal Picard stack $\mathfrak{Pic}_{q,n,d}$. The operational class is the class associated with a certain proper representable morphism

$$\operatorname{Div}_{g,A} \longrightarrow \operatorname{\mathfrak{Pic}}_{g,n,d},$$

using the theory of §2.4. Our goal here is to define the stack $\operatorname{Div}_{g,A}$ over $\operatorname{\mathfrak{Pic}}_{g,n,d}$.

We will present in \$\$.2-3.4 three essentially equivalent definitions of the universal twisted double ramification cycle, which yield the same operational class:

• a definition in $\S3.2$ by closing the Abel–Jacobi section, which is simple to state but difficult to handle;

• an intrinsic logarithmic definition in §3.3 following Marcus–Wise [56];

• a slight variation of the log definition in $\S3.4$, which facilitates comparison to the spaces of rubber maps.

After analyzing the set-theoretic closure of the Abel–Jacobi section in §3.5, the equality of the three resulting classes will be shown in $\S3.6$. In $\S3.8$, we briefly discuss the lift of universal twisted double ramification cycle to operational b-Chow.

243

3.2. $\mathsf{DR}_{q,A}^{\mathsf{op}}$ by closure

We define the Abel–Jacobi section σ of $\mathfrak{Pic}_{q,n,d} \rightarrow \mathfrak{M}_{g,n}$ by

$$\sigma: \mathfrak{M}_{g,n} \longrightarrow \mathfrak{Pic}_{g,n,d},$$

$$(C, p_1, \dots, \longmapsto \mathcal{O}_C \bigg(\sum_{i=1}^n a_i p_i \bigg).$$
(3.1)

The section σ is not a closed immersion (both because of the \mathbb{G}_m -automorphism groups of line bundles and because the image is not closed). However, σ is quasi-compact and relatively representable by schemes, and hence admits a well-defined *schematic image* (we use that the formation of the schematic image is compatible with flat base change, see [73, Lemma 0811]). The schematic image is the smallest closed reduced substack through which σ factors.

Since the schematic image $\bar{\sigma}$ is a closed substack of pure dimension,

$$\iota: \bar{\sigma} \longrightarrow \mathfrak{Pic}_{g,n,d}$$

we obtain an operational class $\iota_{op}[\bar{\sigma}]$ by Definition 2.8. Our first definition of the universal twisted double ramification cycles is via the schematic image of σ :

$$\mathsf{DR}_{g,A}^{\mathsf{op}} = \iota_{\mathsf{op}}[\bar{\sigma}] \in \mathsf{CH}_{\mathsf{op}}^g(\mathfrak{Pic}_{g,n,d}). \tag{3.2}$$

Let

$$\mathfrak{Pic}_0 \hookrightarrow \mathfrak{Pic}$$

be the open substack consisting of line bundles having degree zero on every irreducible component of every geometric fiber (*multidegree* $\underline{0}$), and let

$$\mathfrak{Pic}_0^{\mathrm{rel}} \hookrightarrow \mathfrak{Pic}^{\mathrm{rel}}$$

be defined analogously. We have a commutative diagram in which all squares are pullbacks:



Let $(C/B, p_1, ..., p_n)$ be a prestable curve over a scheme B of finite type over K. Let L be a line bundle on C such that

$$L\left(-\sum_{i=1}^n a_i p_i\right)$$

is of multidegree $\underline{0}$ for $A = (a_1, ..., a_n) \in \mathbb{Z}^n$. The data

$$C \longrightarrow B, \quad L\left(-\sum_{i=1}^{n} a_i p_i\right) \longrightarrow C$$

determine a map

$$\varphi: B \longrightarrow \mathfrak{Pic}_{g,n,\underline{0}},$$

and we form a pullback diagram

$$B' \longrightarrow \mathfrak{M}_{g,n} \\ \downarrow^{\psi} \qquad \downarrow^{e} \\ B \xrightarrow{\varphi^{\mathsf{rel}}} \mathfrak{Pic}_{g,n,\underline{0}}^{\mathsf{rel}}.$$

$$(3.4)$$

Since $\mathfrak{M}_{g,n}$ is smooth and $\mathfrak{Pic}_{g,n,\underline{0}}^{\mathrm{rel}}$ is separated, the map e is a regular embedding.

LEMMA 3.1. In the multidegree $\underline{0}$ case, we have

$$\mathsf{DR}_{g,A}^{\mathsf{op}}(\varphi)([B]) = \psi_* e^! [B]. \tag{3.5}$$

Proof. We begin by expanding the diagram (3.4) to

$$B' \longrightarrow \mathfrak{M}_{g,n} \times B \xrightarrow{f'} \mathfrak{M}_{g,n}$$

$$\downarrow^{\psi} \qquad \qquad \qquad \downarrow^{e}$$

$$B \xrightarrow{\varphi'} \mathfrak{Pic}_{g,n,\underline{0}}^{\mathrm{rel}} \times B \xrightarrow{f} \mathfrak{Pic}_{g,n,\underline{0}}^{\mathrm{rel}}.$$

$$(3.6)$$

Since $\mathfrak{Pic}_{g,n} \to \mathfrak{Pic}_{g,n}^{\mathrm{rel}}$ is smooth, we deduce from Lemma 2.17 and diagram (3.3) that

$$\mathsf{DR}_{g,A}^{\mathsf{op}}(\varphi)([B]) = \psi_* \varphi'^! [\mathfrak{M}_{g,n} \times B].$$
(3.7)

We then compute

$$\begin{aligned} \mathsf{DR}_{g,A}^{\mathsf{op}}(\varphi)([B]) &= \psi_* \varphi'^! [\mathfrak{M}_{g,n} \times B] \\ &= \psi_* \varphi'^! f'^* [\mathfrak{M}_{g,n}] \\ &= \psi_* \varphi'^! f'^* e^! [\mathfrak{Pic}_{g,n,\underline{0}}^{\mathrm{rel}}] \\ &= \psi_* \varphi'^! e^! f^* [\mathfrak{Pic}_{g,n,\underline{0}}^{\mathrm{rel}}] \\ &= \psi_* \varphi'^! e^! [\mathfrak{Pic}_{g,n,\underline{0}}^{\mathrm{rel}} \times B] \\ &= \psi_* e^! \varphi'^! [\mathfrak{Pic}_{g,n,\underline{0}}^{\mathrm{rel}} \times B] \\ &= \psi_* e^! [B]. \end{aligned}$$

245

In particular, if the intersection of B with the unit section in $\mathfrak{Pic}_{g,n,\underline{0}}^{\mathrm{rel}}$ is transversal, then we simply take the naive intersection in $\mathfrak{Pic}_{a,n,0}^{\mathrm{rel}}$ and push it down to B.

3.3. Logarithmic definition of DR^{op}

3.3.1. Overview of log divisors

We begin by recalling various results and definitions from log geometry. We refer the reader to [47] for basics on log geometry and [56] for the details of what we do here. While log geometry will not play a substantial role elsewhere in the paper, it will reappear in §6.

Given a log scheme $S = (S, M_S)$, we write

$$\mathbb{G}^{\mathsf{log}}_m(S) = \Gamma(S, M^{\mathsf{gp}}_S) \quad \text{and} \quad \mathbb{G}^{\mathsf{trop}}_m(S) = \Gamma(S, \overline{M}^{\mathsf{gp}}_S),$$

which we call the logarithmic and tropical multiplicative groups. Both can naturally be extended to presheaves on the category \mathbf{LSch}_S of log schemes over S, and both admit log smooth covers by log schemes (with subdivisions \mathbb{P}^1 and $[\mathbb{P}^1/\mathbb{G}_m]$, respectively). A log (tropical) line on S is a \mathbb{G}_m^{\log} ($\mathbb{G}_m^{\mathrm{trop}}$) torsor on S, for the strict étale topology.

Definition 3.2. (See [56, Definition 4.6]) Let C be a logarithmic curve over a logarithmic scheme S. A logarithmic divisor on C is a tropical line P over S and an S-morphism $C \rightarrow P$.

Let $\mathbf{Div}_{g}^{\mathsf{rel}}$ be the stack $(^{20})$ in the strict étale topology on logarithmic schemes whose S-points are triples (C, P, α) , where C is a logarithmic curve of genus g over S, P is a tropical line over S, and

$$\alpha \colon C \longrightarrow P$$

is an S-morphism.

If S is a geometric log point and C/S is a log curve, then the set of isomorphism classes of $\mathbf{Div}_{g}^{\mathsf{rel}}(S)$ is given by $\pi_{*}\overline{M}_{C}^{\mathsf{gp}}/\overline{M}_{S}^{\mathsf{gp}}$. At the markings, an element of $\pi_{*}\overline{M}_{C}^{\mathsf{gp}}/\overline{M}_{S}^{\mathsf{gp}}$ determines an element of the groupified relative characteristic monoid \mathbb{Z} (for those who prefer a tropical perspective, this can be viewed as the outgoing slope at the marking).

Definition 3.3. Let $\mathbf{Div}_{g,A}^{\mathsf{rel}}$ be the (open and closed) substack of $\mathbf{Div}_{g}^{\mathsf{rel}}$ consisting of those triples where the curve carries exactly n markings and where on each geometric fiber the outgoing slopes at the markings correspond to A (our log curves come with an ordering of their markings as explained in §1.2).

 $[\]binom{20}{g}$ The stack $\mathbf{Div}_{g}^{\mathsf{rel}}$ was denoted \mathbf{Div}_{g} in [56], but we wish to reserve the latter notation for a certain \mathbb{G}_{m} -gerbe over $\mathbf{Div}_{g}^{\mathsf{rel}}$ which will play a much more prominent role in our paper.

Remark 3.4. It is natural to ask for a description of the functor of points of the underlying (non-logarithmic) stack of $\mathbf{Div}_{g,A}^{\mathsf{rel}}$ as a fibered category over $\mathfrak{M}_{g,n}$. However, we expect that such a description will not be simple. A closely related problem is solved in [12], and the intricacy of the resulting definition suggests that the path will not be easy.

3.3.2. Abel–Jacobi map

Given a log curve $\pi: C \to S$ of genus g, the right-derived pushforward to S of the standard exact sequence

$$1 \longrightarrow \mathcal{O}_C^{\times} \longrightarrow M_C^{gp} \longrightarrow \overline{M}_C^{gp} \longrightarrow 1$$
(3.8)

yields a natural map

$$\pi_* \overline{M}_C^{\mathsf{gp}} \longrightarrow R^1 \pi_* \mathcal{O}_C^{\times},$$

which factors via the quotient

$$\pi_* \overline{M}_C^{\mathsf{gp}} / \overline{M}_S^{\mathsf{gp}} = \mathbf{Div}_q^{\mathsf{rel}}(S).$$

We therefore obtain a *relative Abel–Jacobi* map

$$\mathsf{AJ}^{\mathsf{rel}}:\mathbf{Div}_{g}^{\mathsf{rel}}\longrightarrow\mathfrak{Pic}_{g}^{\mathsf{rel}},$$

which restricts to maps

$$\operatorname{AJ}^{\operatorname{rel}}:\operatorname{Div}_{g,A}^{\operatorname{rel}}\longrightarrow \operatorname{\mathfrak{Pir}}_{g,n,d}^{\operatorname{rel}}.$$

For a first example, suppose S is a geometric log point with $\overline{M}_S = \mathbb{N}$. The data then determines to first order a deformation of the curve over a DVR (which we take generically smooth), and the section of $\pi_* \overline{M}_C^{gp} / \overline{M}_S^{gp}$ gives the multiplicities of components in the special fiber and the twists by the markings.

For another example, consider what happens over the locus of (strict) smooth curves. Writing $N/\mathcal{M}_{g,n}^{\log}$ for the stack of markings (finite étale), we see $\mathbf{Div}_{g}^{\mathsf{rel}}$ is just the category of locally constant functions from N to \mathbb{Z} —in other words, choices of outgoing slope/weight on each leg. The Abel–Jacobi map yields

$$\mathcal{O}_C\bigg(\sum_{i=1}^n a_i p_i\bigg)$$

where the p_i are the markings and a_i are the weights. In particular, we see that

$$\mathbf{Div}_{g,A}^{\mathsf{rel}} \!
ightarrow \! \mathcal{M}_{g,n}^{\mathsf{log}}$$

is birational (as we fixed an ordering of the markings) and log étale.

Definition 3.5. Let \mathbf{Div}_q be the fiber product

$$\mathbf{Div}_{g}^{\mathsf{rel}} \times_{\mathfrak{Pic}_{a}^{\mathsf{rel}}} \mathfrak{Pic}_{g}. \tag{3.9}$$

More concretely, an S-point of \mathbf{Div}_g is a quadruple $(C, P, \alpha, \mathcal{L})$, where (C, P, α) is an S-point of $\mathbf{Div}_q^{\mathsf{rel}}$, and \mathcal{L} is a line bundle on C satisfying⁽²¹⁾

$$[\mathcal{L}] = \operatorname{AJ}^{\operatorname{rel}}(C, P, \alpha) \in \mathfrak{Pic}_{q}^{\operatorname{rel}}(S).$$

We will denote by AJ the resulting Abel–Jacobi map

$$\operatorname{Div}_g \to \operatorname{\mathfrak{Pic}}_q$$
.

Observe that \mathbf{Div}_g is a \mathbb{G}_m -gerbe over $\mathbf{Div}_g^{\mathsf{rel}}$, just as $\mathfrak{Pic}_{g,n,d}$ is a \mathbb{G}_m -gerbe over $\mathfrak{Pic}_{g,n,d}^{\mathsf{rel}}$. Analogously, we define

$$\mathbf{Div}_{g,A} = \mathbf{Div}_{g,A}^{\mathsf{rel}} \times_{\mathfrak{Pic}_{g,n,d}^{\mathsf{rel}}} \mathfrak{Pic}_{g,n,d} \quad \text{and} \quad \mathsf{AJ:} \mathbf{Div}_{g,A} \longrightarrow \mathfrak{Pic}_{g,n,d}.$$
(3.10)

We summarize the key properties of the Abel–Jacobi map. These are proven in [56] for AJ^{rel} , and are stable under base change.

PROPOSITION 3.6. The Abel-Jacobi map

AJ:
$$\operatorname{Div}_{g,A} \longrightarrow \operatorname{\mathfrak{Pic}}_{g,n,d}$$

is proper, relatively representable by algebraic spaces, and is a monomorphism of log stacks.

We obtain an operational class $A_{Jop}[Div_{g,A}]$ associated by Definition 2.8 with the Abel–Jacobi map AJ. Our second definition of the universal twisted double ramification cycles is via AJ:

$$\mathsf{DR}_{q,A}^{\mathsf{op}} = \mathsf{AJ}_{\mathsf{op}}[\mathbf{Div}_{g,A}] \in \mathsf{CH}_{\mathsf{op}}^g(\mathfrak{Pic}_{g,n,d}).$$
(3.11)

The equivalence of definitions (3.2) and (3.11) will be proven in §3.6.

3.3.3. Description of Div_g with log line bundles

Our approach to \mathbf{Div}_g in Definition 3.5 via a fiber product is indirect. While it will not be used in the paper, a more conceptual path is to consider the stack \mathbf{Div}'_g whose objects are tuples

 $(C/S, \mathcal{P}, \alpha),$

248

 $^(^{21})$ Here, [·] denotes the equivalence class under the relations of isomorphism and tensoring with the pullback of a line bundle from the base.

where C/S is a log curve, \mathcal{P} is a *logarithmic* line on S (a \mathbb{G}_m^{\log} torsor), and α is a map from C to the *tropical* line P on S induced from \mathcal{P} by the exact sequence (3.8). There is a natural map

$$\operatorname{Div}_q^{\prime} \longrightarrow \operatorname{Div}_q^{\operatorname{rel}}.$$

We can see α as a section of the tropicalization of the pullback of \mathcal{P} to C. As such, by the sequence (3.8), α induces a \mathbb{G}_m -torsor on C, giving us an Abel–Jacobi map $\mathbf{Div}'_g \to \mathfrak{Pic}_g$. Together these maps induce a map

$$\operatorname{Div}_q' \longrightarrow \operatorname{Div}_g$$

to the fiber product, and a local computation verifies that this is an isomorphism.

The above discussion $points(^{22})$ towards a definition of the double ramification cycle via the *logarithmic Picard functor* of [60], which we hope will be pursued in future.

3.4. Logarithmic rubber definition of DR^{op}

Marcus and Wise introduce a slight variant $\operatorname{Rub}_{g}^{\mathsf{rel}}$ of the stack $\operatorname{Div}_{g}^{\mathsf{rel}}$ which parameterizes pairs (C, P, α) , where P is a tropical line on S and $\alpha: C \to P$ is an S-morphism such that, on each geometric fiber over S, the values taken by α on the irreducible components of C are totally ordered in $(\overline{M}_{S}^{\mathsf{gp}})_{s}$, with the ordering given by declaring the elements of $(\overline{M}_{S})_{s}$ to be the non-negative elements.

The space $\mathbf{Rub}_{q,A}$, defined via pullback

$$\mathbf{Rub}_{g,A} \stackrel{\sim}{=} \mathbf{Div}_{g,A} \times_{\mathbf{Div}_{g,A}} \mathbf{Rub}_{g,A}^{\mathsf{rel}},$$

is pure dimensional and comes with a proper birational map

$$\mathbf{Rub}_{q,A} \longrightarrow \mathbf{Div}_{q,A}.$$
 (3.12)

The stack $\operatorname{\mathbf{Rub}}_{g,A}$ will play an important role in the comparison to classes coming from stable map spaces in §6.

We obtain an operational class $AJ_{op}^{rub}[\mathbf{Rub}_{g,A}]$ associated by Definition 2.8 with

AJ
$$^{\mathsf{rub}}:\mathbf{Rub}_{g,A}\,{\longrightarrow}\,\mathfrak{Pic}_{g,n,d}$$

 $^(^{22})$ The unit section of \mathfrak{Pic} is given by the stack of \mathbb{G}_m torsors on the base. Similarly, the unit section of the logarithmic Picard stack $\mathfrak{Log}\mathfrak{Pic}_g$ is given by the stack of \mathbb{G}_m^{\log} torsors on the base. The natural map $\mathfrak{Pic}_g \to \mathfrak{Log}\mathfrak{Pic}_g$ is neither injective nor surjective: a logarithmic line bundle comes from a line bundle if and only if the associated tropical line bundle is trivial, and a choice of trivialization of that tropical line bundle then determines a lift to a line bundle. Hence, we see that \mathbf{Div}'_g is precisely the pullback of the unit section of $\mathfrak{Log}\mathfrak{Pic}_g$ to \mathfrak{Pic}_g .

obtained by composing (3.12) with AJ. Our third definition of the universal twisted double ramification cycles is via AJ^{rub}:

$$\mathsf{DR}_{g,A}^{\mathsf{op}} = \mathsf{AJ}_{\mathsf{op}}^{\mathsf{rub}}[\mathbf{Rub}_{g,A}] \in \mathsf{CH}_{\mathsf{op}}^g(\mathfrak{Pic}_{g,n,d}).$$
(3.13)

The equivalence with the first two definitions will be proven in $\S3.6$.

3.5. The image of the Abel–Jacobi map

The set theoretic image of the Abel–Jacobi map

AJ:
$$\operatorname{Div}_{g,A} \longrightarrow \operatorname{\mathfrak{Pic}}_{g,n,d}$$

can be characterized in terms of a condition on twisted divisors similar to the conditions of [28] for the moduli spaces $\widetilde{\mathcal{H}}_q(A)$ twisted canonical divisors.

Given a prestable graph Γ_{δ} of degree d, a *twist* on Γ_{δ} is a function $I: H(\Gamma) \to \mathbb{Z}$ which satisfies the following properties:

(i) for all $j \in L(\Gamma_{\delta})$, corresponding to the marking $j \in \{1, ..., n\}$, one has

$$I(j) = a_j;$$

(ii) for all $e \in E(\Gamma_{\delta})$, corresponding to two half-edges $h, h' \in H(\Gamma_{\delta})$, one has

$$I(h) + I(h') = 0;$$

(iii) for all $v \in V(\Gamma_{\delta})$, one has

$$\sum_{v(h)=v} I(h) = \delta(v),$$

where the sum is taken over all n(v) half-edges incident to v;

(iv) there is no strict cycle⁽²³⁾ in Γ .

Let $(C, p_1, ..., p_n)$ together with a line bundle $\mathcal{L} \to C$ of degree d be a geometric point of $\mathfrak{Pic}_{g,n,d}$. Let Γ_{δ} be the prestable graph of C decorated with the degrees $\delta(v)$ of the line bundle \mathcal{L} restricted to the components C_v of C. Given a twist I on Γ_{δ} , let

$$\eta_I : C_I \longrightarrow C$$

 $^(^{23})$ A strict cycle is a sequence $\vec{e}_i = (h_i, h'_i), i=1, ..., \ell$, of directed edges in Γ forming a closed path in Γ such that $I(h_i) \ge 0$ for all *i*, and there exists at least one *i* with $I(h_i) > 0$. Condition (iv) corresponds to the combination of the vanishing, sign, and transitivity conditions for twists in [28, §0.3].

be the partial normalization of C at all nodes $q\!\in\!C$ corresponding to edges $e\!=\!(h,h')$ of Γ with

$$I(h) = -I(h') \neq 0.$$

Denote by $q_h, q_{h'} \in C_I$ the preimages of q corresponding to the half-edges h and h', respectively. Denote by $\hat{p}_i \in C_I$ the unique preimage of the *i*th marking $p_i \in C$.

We say the point $(C, p_1, ..., p_n, \mathcal{L})$ of $\mathfrak{Pic}_{g,n,d}$ satisfies the *twisted divisor condition* for the integer vector A if and only if there exists a twist I on Γ_{δ} such that, on the partial normalization C_I of C, there exists an isomorphism of line bundles

$$\eta_I^* \mathcal{L} \cong \mathcal{O}_C \bigg(\sum_{i=1}^n a_i \hat{p}_i + \sum_{h \in H(\Gamma)} I(h) q_h \bigg).$$
(3.14)

251

For $\mathcal{L} = \omega_C$, this exactly corresponds to the notion [28, Definition 1] of a twisted canonical divisor.

PROPOSITION 3.7. A geometric point $(C, p_1, ..., p_n, \mathcal{L})$ of $\mathfrak{Pic}_{g,n,d}$ lies in the image of the Abel-Jacobi map $AJ: \mathbf{Div}_{g,A} \to \mathfrak{Pic}_{g,n,d}$ if and only if the twisted divisor condition for the vector A is satisfied.

Proof. We may suppose that K is a separably closed field and $(C/S, p_1, ..., p_n)$ is a prestable curve over K. We must show that the twisted divisor condition is equivalent to the existence of a log structure on C/S satisfying the following property: C/S is a log curve with markings given by the p_i which admits a global section α of \overline{M}_C^{gp} with outgoing slope at p_i given by a_i .

Suppose that such a log structure exists. From the log structure, we can determine a twist. With each leg we associate the outgoing slope of α on the corresponding leg. For an edge $\{h, h'\}$, we define $\ell(\{h, h'\})$ to be the element of \overline{M}_S associated via the data of the log morphism $C \to S$ with the node of C corresponding to $\{h, h'\}$. If $\{h, h'\}$ is an edge with half-edge h attached to a vertex u and the opposite half-edge h' attached to v, we set the integer I(h) to be the unique integer such that

$$\alpha(u) + I(h) \cdot \ell(\{h, h'\}) = \alpha(v) \in M_S^{gp}.$$

$$(3.15)$$

That such an I(h) exists follows from the structure of \overline{M}_C^{gp} .

Next, we verify that I is a twist. Conditions (i) and (ii) are immediate from the construction. We deduce condition (iv) because by following a strict cycle starting at some vertex u and applying (3.15) along each edge would yield $\alpha(u) < \alpha(u)$, which is impossible. Condition (iii) is immediate from the twisted divisor condition (3.14) and the fact that isomorphic line bundles have the same degree, so this will be proven once we have checked the latter condition.

For the latter condition, we must work a little harder. To start, we claim that there exists a morphism $\overline{M}_S \to \mathbb{N}$ which does not send the label of any edge to zero. Indeed, \overline{M}_s^{gp} injects into its groupification which is a finitely generated torsion-free abelian group, and hence isomorphic to \mathbb{Z}^m . Since \overline{M}_S is sharp(²⁴) and finitely generated, the non-zero elements of its image in \mathbb{Z}^m land in some strict half-space of \mathbb{Z}^m cut out by a linear equation with integral coefficients. Such a half-space admits a map to \mathbb{N} such that the only preimage of zero is zero.

After base changing over S along such a map, we may assume that $\overline{M}_{S}^{gp} = \mathbb{N}$, and that all edges have non-zero label. We obtain a first-order map passing through our given point,

$$S = \operatorname{Spec} K[[t]] \longrightarrow \operatorname{Div}_{g,A}$$

for which the induced prestable curve C/S is generically smooth. On the curve C, we define a Weil divisor D by assigning to an irreducible component v the integer $\alpha(v)$. The divisor D is then Cartier by (3.15), which still applies after base change, and hence determines a line bundle $\mathcal{O}_C(-D)$, which is exactly the image of the Abel–Jacobi map. In particular, the bundle \mathcal{L} is (up to isomorphism) given by restricting $\mathcal{O}_C(-D)$ to the central fiber, so it suffices to verify (3.14) for the latter bundle, which is a standard local calculation on a prestable curve over a discrete valuation ring.

Conversely, suppose the twisted divisor condition is satisfied. We must build a log structure and a suitable section $\alpha \in \overline{M}_C^{gp}(C)$. We could try equipping $(C/S, p_1, ..., p_n)$ with its minimal log structure (see §1.2), but then the section α is unlikely to exist—if there are no separating edges then there are no non-constant sections of \overline{M}_C^{gp} . Instead, we will construct a log structure by deforming the curve.

First, we claim that there exists an assignment of a positive integer $\ell(e) \in \mathbb{Z}_{>0}$ to each edge and of an integer $d(v) \in \mathbb{Z}$ to each vertex such that the following condition is satisfied:

if
$$e = \{h, h'\}$$
 is an edge with h attached to u and h' to v ,
then $d(u) + I(h) \cdot \ell(e) = d(v)$. (*)

A twist I on Γ induces a binary relation \preceq on $V(\Gamma)$ by

 $u \leq v \iff$ there is an edge $e = \{h, h'\}$ with h at u, h' at v, and $I(h) \ge 0$.

The fact that Γ contains no strict cycles is equivalent by [74] to the existence of an extension of \leq to a total preorder on $V(\Gamma)$ (a reflexive, total, and transitive binary relation). Hence, there exists a level function $d_0: V(\Gamma) \to \mathbb{Z}$ such that

$$u \leq v \iff u \text{ and } v \text{ are connected by an edge and } d_0(u) \leq d_0(v).$$
 (3.16)

 $^(^{24})$ A monoid is *sharp* if zero is indecomposable: a+b=0 implies a=b=0.

253

We define

$$L = lcm\{I(h) : h \in H(\Gamma), I(h) > 0\}.$$

Then, $d(v) = Ld_0(v)$ still has property (3.16) and, for any edge $e = \{h, h'\}$ with h attached to u and h' to v, we have two cases:

• I(h)=0, in which case all edges $\{\tilde{h}, \tilde{h}'\}$ connecting u and v must satisfy $I(\tilde{h})=0$ (due to the strict cycle condition), so we can set $\ell(e)=1$;

• $I(h) \neq 0$, in which case the number

$$\ell(e) = \frac{d(v) - d(u)}{I(h)}$$

is indeed a positive integer (since d has values in $L \cdot \mathbb{Z}$).

Clearly, the functions d and ℓ thus constructed satisfy the condition above.

Such d and ℓ are far from unique, but we pick them. Consider then the space of all smoothings C of C over K[[t]] such that the thickness(²⁵) of C at the node corresponding to edge e is $\ell(e)$. Given such a smoothing C, we construct a vertical Weil divisor D by assigning to the irreducible component corresponding to vertex v the weight d(v). The divisor D is then Cartier by the condition (*). Set

$$\mathcal{L}_{\mathcal{C}} = \mathcal{O}_{\mathcal{C}}(D)|_{C} \otimes \mathcal{O}_{C}\left(\sum_{i} a_{i} p_{i}\right).$$

The smoothing \mathcal{C} also induces a log structure on C by taking the divisorial log structure of the special fiber. The twist I then determines an element α of $\pi_* \overline{M}_C^{gp} / \overline{M}_S^{gp}$. Applying the Abel–Jacobi map to α recovers $\mathcal{L}_{\mathcal{C}}$.

One can readily verify that $\mathcal{L}_{\mathcal{C}}$ satisfies (3.14) by a local computation, but we need to show more: the smoothing \mathcal{C} can be chosen so that $\mathcal{L}_{\mathcal{C}}$ is isomorphic to the line bundle \mathcal{L} that we started with. The space of such smoothings \mathcal{C} naturally surjects onto

$$\bigoplus_{\substack{e=\{h,h'\}\\I(h)\neq 0}} (\mathcal{O}_{C_I}(h) \otimes \mathcal{O}_{C_I}(h'))^{\otimes \ell(e)},$$
(3.17)

where the 1-dimensional K-vector space

$$(\mathcal{O}_{C_I}(h)\otimes\mathcal{O}_{C_I}(h'))^{\otimes\ell(e)}$$

corresponds exactly to the ways to glueing the two branches of $\eta_I^* \mathcal{L}$ together at the points h and h'. In other words, by moving over the space (3.17), we can recover all ways of glueing $\eta_I^* \mathcal{L}$ to a line bundle on C. In particular, we can recover \mathcal{L} , and hence we can realize \mathcal{L} as $\mathcal{L}_{\mathcal{C}}$ for some smoothing \mathcal{C} , as required.

 $^(^{25})$ The local equation of the node is $xy=t^r$ for some positive integer r which we call the *thickness* of the node.

3.6. Proof of the equivalence of the definitions

The equivalence of the classes coming from $\mathbf{Div}_{g,A}$ and from $\mathbf{Rub}_{g,A}$ is immediate by applying Proposition 2.16. We must compare the latter two with the class defined by (3.2) via the schematic image. We will require the following two easy results.

LEMMA 3.8. Let $U \hookrightarrow \operatorname{Div}_{g,A}$ denote the open locus where the log curve is classically smooth. Then, U is schematically dense in $\operatorname{Div}_{g,A}$.

Proof. Since $\mathbf{Div}_{g,A} \to \mathcal{M}_{g,n}^{\log}$ is log étale, we deduce that $\mathbf{Div}_{g,A}$ is log regular. In particular, $\mathbf{Div}_{g,A}$ is reduced, and the locus where the log structure is trivial is dense. \Box

LEMMA 3.9. The Abel-Jacobi map $AJ: \operatorname{Div}_{g,A} \to \operatorname{\mathfrak{Pic}}_{g,n,d}$ factors through the inclusion $\overline{\sigma} \to \operatorname{\mathfrak{Pic}}_{a,n,d}$, and the induced map

$$\mathbf{Div}_{g,A} \longrightarrow \bar{\sigma}$$

is proper and birational.

Proof. That $\mathbf{Div}_{g,a} \to \mathfrak{Pic}_{g,n,d}$ factors through $\bar{\sigma} \to \mathfrak{Pic}$ is immediate from Lemma 3.8 and the definition of the schematic image. The induced map $\mathbf{Div}_{g,A} \to \bar{\sigma}$ is proper since $\mathbf{Div}_{g,A}$ is proper over $\mathfrak{Pic}_{g,n,d}$, and is birational since it is an isomorphism over the locus of smooth curves.

By another application of Proposition 2.16, the definitions of $\mathsf{DR}^{\mathsf{op}}$ via $\mathbf{Div}_{g,A}$ and the schematic image are equivalent.

3.7. Proof of Theorem 0.1

Let $k \ge 0$, and let $A = (a_1, ..., a_n)$ be a vector of integers satisfying

$$\sum_{i=1}^n a_i = k(2g-2)$$

There are three definitions in the literature for the classical twisted double ramification cycle

$$\mathsf{DR}_{q,A,\omega^k} \in \mathsf{CH}_{2q-3+n}(\mathcal{M}_{q,n})$$

on the moduli space of stable curves:

- via birational modifications of $\overline{\mathcal{M}}_{g,n}$ [37];
- via the closure of the image of the Abel–Jacobi section [38];
- via logarithmic geometry and the stack $\mathbf{Div}_{g}^{\mathsf{rel}}$ [56].

All three are shown to be equivalent in [37], [38]. For the proof of Theorem 0.1, we choose the definition of [56], as this will give the shortest path.

For d=k(2g-2), let $\varphi:\overline{\mathcal{M}}_{q,n}\to\mathfrak{Pic}_{q,n,d}$ be the morphism associated with the data

$$\pi: \mathcal{C}_{g,n} \longrightarrow \overline{\mathcal{M}}_{g,n}, \quad \omega_{\pi}^k \longrightarrow \mathcal{C}_{g,n}.$$
(3.18)

To prove Theorem 0.1, we must show that

$$\mathsf{DR}_{g,A}^{\mathsf{op}}(\varphi)([\overline{\mathcal{M}}_{g,n}]) = \mathsf{DR}_{g,A,\omega^k},$$

where $\left[\overline{\mathcal{M}}_{q,n}\right]$ is the fundamental class.

We form the pullback diagram

Following Definition 2.8, we have

$$\mathsf{DR}_{g,A}^{\mathsf{op}}(\varphi)([\overline{\mathcal{M}}_{g,n}]) = \psi_*(\varphi')^!([\mathbf{Div}_{g,A} \times \overline{\mathcal{M}}_{g,n}]).$$

The construction is equivalent to the definition of the class $\mathsf{DR}_{g,A,\omega^k}$ in [56] after making the standard translation between the Gysin pullback and the virtual fundamental class as in [11, Example 7.6].

3.8. The double ramification cycle in b-Chow

The construction of the double ramification cycle in [37] naturally yielded a more refined object: a b-cycle(²⁶) on $\overline{\mathcal{M}}_{g,n}$ which pushes down to the usual double ramification cycle on $\overline{\mathcal{M}}_{g,n}$. The refined cycle was shown in [39] to have better properties with respect to intersection products than the usual double ramification cycle. By considering rational sections of the multidegree-zero relative Picard space over $\mathfrak{Pic}_{g,n,d}$, we can in an analogous way define a b-cycle on $\mathfrak{Pic}_{g,n,d}$ refining the universal twisted double ramification cycle introduced here. In future work, we will show that this refined universal cycle is compatible with intersection products in the sense of [39] and that the *toric contact cycles* of [68] can be obtained by pulling back these products.

 $^(^{26})$ An element of the colimit of the Chow groups of smooth blowups of $\overline{\mathcal{M}}_{g,n}$ with transition maps given by pullback.

4. Pixton's formula

4.1. Reformulation

Recall the cycle $\mathsf{P}_{g,A,d}^c \in \mathsf{CH}_{\mathsf{op}}^c(\mathfrak{Pic}_{g,n,d})$ defined in §0.3.5. We write

$$\mathsf{P}^{\scriptscriptstyle\bullet}_{g,A,d} = \sum_{c=0}^\infty \mathsf{P}^c_{g,A,d} \in \prod_{c=0}^\infty \mathsf{CH}^c_{\mathsf{op}}(\mathfrak{Pic}_{g,n,d})$$

for the associated mixed dimensional class. We will rewrite the formula for $\mathsf{P}^{\bullet}_{g,A,d}$ in a more convenient form for computation.

Several factors in the formula of 0.3.5 can be pulled out of the sum over graphs and weightings. We require the following four definitions:

• Let $G_{g,n,d}^{se}$ be the set of graphs in $G_{g,n,d}$ having exactly two vertices connected by a single edge. Such graphs are thus described by a partition

$$(g_1, I_1, \delta_1 \mid g_2, I_2, \delta_2) \tag{4.1}$$

of the genus, the marking set, and the degree of the universal line bundle.

• Given a vector $A = (a_1, ..., a_n) \in \mathbb{Z}^n$ satisfying

$$\sum_{i=1}^{n} a_i = d$$

and $\Gamma_{\delta} \in \mathsf{G}_{q,n,d}^{\mathsf{se}}$ corresponding to the partition (4.1), we define

$$c_A(\Gamma_{\delta}) = -\left(\delta_1 - \sum_{i \in I_1} a_i\right)^2 = -\left(\delta_2 - \sum_{i \in I_2} a_i\right)^2.$$

• For $\Gamma_{\delta} \in \mathsf{G}_{g,n,d}^{\mathsf{se}}$, we write

$$[\Gamma_{\delta}] = \frac{1}{|\mathrm{Aut}(\Gamma_{\delta})|} j_{\Gamma_{\delta}*}[\mathfrak{Pic}_{\Gamma_{\delta}}]$$

for the class of the boundary divisor of $\mathfrak{Pic}_{g,n,d}$ associated with Γ_{δ} .

• Let $G_{q,n,d}^{nse}$ be the set of graphs in $G_{g,n,d}$ such that every edge is non-separating.

PROPOSITION 4.1. The class $\mathsf{P}^{\bullet}_{q,A,d}$ is the constant term in r of

$$\exp\left(\frac{1}{2}\left(-\eta+\sum_{i=1}^{n}2a_{i}\xi_{i}+a_{i}^{2}\psi_{i}+\sum_{\Gamma_{\delta}\in\mathbf{G}_{g,n,d}^{se}}c_{A}(\Gamma_{\delta})[\Gamma_{\delta}]\right)\right)$$

$$\times\sum_{\substack{\Gamma_{\delta}\in\mathbf{G}_{g,n,d}^{nse}\\w\in\mathbf{W}_{\Gamma_{\delta},r}}}\frac{r^{-h^{1}(\Gamma_{\delta})}}{|\operatorname{Aut}(\Gamma_{\delta})|}j_{\Gamma_{\delta}*}\left[\prod_{e=(h,h')\in\mathrm{E}(\Gamma_{\delta})}\frac{1}{\psi_{h}+\psi_{h'}}\left(1-\exp\left(-\frac{w(h)w(h')}{2}(\psi_{h}+\psi_{h'})\right)\right)\right],$$

$$(4.2)$$

for $r \gg 0$.

In the proof of Proposition 4.1, we will use the following computation which provides an interpretation for parts of the formula (4.2) and which will be used again in §8.

LEMMA 4.2. Let $A=(a_1,...,a_n)\in\mathbb{Z}^n$ with $\sum_{i=1}^n a_i=d$. For the line bundle \mathcal{L} on the universal curve

$$\pi: \mathfrak{C}_{g,n,d} \longrightarrow \mathfrak{Pic}_{g,n,d},$$

with universal sections $p_1, ..., p_n$, we define

$$\mathcal{L}_A = \mathcal{L}\left(-\sum_{i=1}^n a_i[p_i]\right).$$

Then, we have

$$-\pi_* c_1(\mathcal{L}_A)^2 = -\eta + \sum_{i=1}^n 2a_i \xi_i + a_i^2 \psi_i.$$

Proof. The result follows from the definitions of the classes η and ξ_i :

$$-\pi_* c_1(\mathcal{L}_A)^2 = -\pi_* \left(c_1(\mathcal{L})^2 + \sum_{i=1}^n -2a_i c_1(\mathcal{L})|_{[p_i]} + a_i^2 [p_i]^2 \right) = -\eta + \sum_{i=1}^n 2a_i \xi_i + a_i^2 \psi_i,$$

where, for the self-intersection $[p_i]^2$, we have used that the first Chern class of the normal bundle of p_i is given by $-\psi_i$.

Proof of Proposition 4.1. We denote by

$$\Phi_a(x) = \frac{1}{x} \left(1 - \exp\left(-\frac{a}{2}x\right) \right) = \sum_{m=0}^{\infty} (-1)^m \left(\frac{a}{2}\right)^{m+1} \frac{1}{(m+1)!} x^m = \frac{a}{2} - \frac{a^2}{8}x + \dots$$

the power series appearing in the edge-terms of Pixton's formula.

As a first step, we show that the constant term in r of

$$\exp\left(\frac{1}{2}\sum_{\substack{\Gamma_{\delta}\in\mathsf{G}_{g,n,d}^{\mathsf{se}}\\ w\in\mathsf{G}_{g,n,d}^{\mathsf{se}}}} c_{A}(\Gamma_{\delta})[\Gamma_{\delta}]\right) \times \sum_{\substack{\Gamma_{\delta}\in\mathsf{G}_{g,n,d}^{\mathsf{se}}\\ w\in\mathsf{W}_{\Gamma_{\delta},r}}} \frac{r^{-h^{1}(\Gamma_{\delta})}}{|\operatorname{Aut}(\Gamma_{\delta})|} j_{\Gamma_{\delta}*} \prod_{e=(h,h')\in\mathrm{E}(\Gamma_{\delta})} \Phi_{w(h)w(h')}(\psi_{h}+\psi_{h'})$$

$$(4.3)$$

and the constant term in r of

$$\sum_{\substack{\Gamma_{\delta} \in \mathsf{G}_{g,n,d} \\ w \in \mathsf{W}_{\Gamma_{\delta},r}}} \frac{r^{-h^{1}(\Gamma_{\delta})}}{|\operatorname{Aut}(\Gamma_{\delta})|} j_{\Gamma_{\delta}*} \prod_{e=(h,h')\in \operatorname{E}(\Gamma_{\delta})} \Phi_{w(h)w(h')}(\psi_{h}+\psi_{h'})$$
(4.4)

258

are equal. The formula (4.4) is a linear combination of boundary strata decorated by edge-terms $(\psi_h + \psi_{h'})^{m(e)}$ for non-negative integers m(e), $e \in \mathcal{E}(\Gamma_{\delta})$, that is, terms of the form

$$j_{\Gamma_{\delta}*} \prod_{e=(h,h')\in \mathcal{E}(\Gamma_{\delta})} (\psi_h + \psi_{h'})^{m(e)}.$$
(4.5)

A first consequence of the combinatorial rules for computing intersections in the tautological ring⁽²⁷⁾ of $\mathfrak{Pic}_{g,n,d}$ is that (4.3) is also a linear combination of such terms. The decorations $(\psi_h + \psi_{h'})^{m(e)}$ on separating edges e = (h, h') appear naturally in the selfintersection formula for the boundary divisors $[\Gamma_{\delta}]$ since, for $\Gamma_{\delta} \in \mathsf{G}_{g,n,d}^{\mathsf{se}}$, the Chern class of the normal bundle of $j_{\Gamma_{\delta}}$ is given by $-(\psi_h + \psi_{h'})$.

We show that the coefficients of the term (4.5) in (4.3) and (4.4) have the same constant term in r. In (4.4), the coefficient is given by

$$\sum_{w \in \mathsf{W}_{\Gamma_{\delta},r}} \frac{r^{-h^{1}(\Gamma_{\delta})}}{|\operatorname{Aut}(\Gamma_{\delta})|} \prod_{e=(h,h')\in \operatorname{E}(\Gamma_{\delta})} (-1)^{m(e)} \left(\frac{w(h)w(h')}{2}\right)^{m(e)+1} \frac{1}{(m(e)+1)!}.$$
(4.6)

On the other hand, let $e_1, ..., e_\ell \in E(\Gamma_\delta)$ be the separating edges of Γ_δ , and let $\overline{\Gamma}_\delta \in \mathsf{G}_{g,n,d}^{\mathsf{nse}}$ be the graph obtained from Γ_δ by contracting these separating edges. Each separating edge e_j corresponds to a unique graph $(\Gamma_j)_{\delta_j} \in \mathsf{G}_{g,n,d}^{\mathsf{se}}$ obtained by contracting all edges of Γ_δ except for e_j .

In the product (4.3), the intersection rules of the tautological ring of $\mathfrak{Pic}_{g,n,d}$ imply that we obtain multiples of the term (4.5) by combining

• for $j=1,...,\ell$, a total of $m(e_j)+1$ terms $[(\Gamma_j)_{\delta_j}]$ from expanding the power series

$$\exp\bigg(\frac{1}{2}\sum_{\Gamma_{\delta}\in\mathsf{G}_{g,n,d}^{\mathsf{se}}}c_{A}(\Gamma_{\delta})[\Gamma_{\delta}]\bigg),$$

• the terms associated with $\bar{\Gamma}_{\delta} \in \mathsf{G}_{g,n,d}^{\mathsf{nse}}$ in the second factor.

Let $M = \sum_{j=1}^{\ell} (m(j)+1)$. Then, (4.5) appears in (4.3) with coefficient

$$\frac{1}{M!} \binom{M}{m(e_1)+1, \dots, m(e_\ell)+1} \left(\prod_{j=1}^{\ell} \left(\frac{c_A((\Gamma_j)_{\delta_j})}{2} \right)^{m(e_j)+1} (-1)^{m(e_j)} \right) \frac{|\operatorname{Aut}(\bar{\Gamma}_{\delta})|}{|\operatorname{Aut}(\Gamma_{\delta})|} \times \sum_{w \in \mathsf{W}_{\bar{\Gamma}_{\delta}, r}} \frac{r^{-h^1(\bar{\Gamma}_{\delta})}}{|\operatorname{Aut}(\bar{\Gamma}_{\delta})|} \prod_{e=(h, h') \in \mathrm{E}(\bar{\Gamma}_{\delta})} (-1)^{m(e)} \left(\frac{w(h)w(h')}{2} \right)^{m(e)+1} \frac{1}{(m(e)+1)!}.$$
(4.7)

 $^(^{27})$ See [30] for the original treatment of the tautological ring of $\overline{\mathcal{M}}_{g,n}$. A corresponding treatment for $\mathfrak{M}_{g,n}$ will be given in [6], [7]. See also [43, §1.1 and §1.7].

To show the equality of (4.6) and (4.7), we combine a number of observations. First, for the multinomial coefficients, we have

$$\frac{1}{M!} \binom{M}{m(e_1)+1, \dots, m(e_\ell)+1} = \prod_{j=1}^{\ell} \frac{1}{(m(e_j)+1)!}$$

Second, for the graph morphism $\Gamma_{\delta} \rightarrow \overline{\Gamma}_{\delta}$ contracting the separating edges, the following statements hold.

• We have an equality of Betti numbers $h^1(\Gamma_{\delta}) = h^1(\overline{\Gamma}_{\delta})$.

• For the separating edges $e_j = (h_j, h'_j)$ of Γ_{δ} , splitting the graph according to the partition $(g_1, I_1, \delta_1 | g_2, I_2, \delta_2)$, the value of every weighting $w \in W_{\Gamma_{\delta}, r}$ is uniquely determined on h_j and h'_j , since

$$w(h_j) = \delta_1 - \sum_{i \in I_1} a_i \pmod{r}$$
 and $w(h'_j) = \delta_2 - \sum_{i \in I_2} a_i \pmod{r}$.

Hence, the constant term in r of $w(h_j)w(h'_j)$ is precisely given by $c_A((\Gamma_j)_{\delta_j})$.

• Concerning the non-separating edges for fixed Γ_{δ} with contraction $\Gamma_{\delta} \rightarrow \bar{\Gamma}_{\delta}$, the map $W_{\Gamma_{\delta},r} \rightarrow W_{\bar{\Gamma}_{\delta},r}$ given by restricting weightings $w \in W_{\Gamma_{\delta},r}$ to the remaining half-edges $H(\bar{\Gamma}_{\delta}) \subset H(\Gamma_{\delta})$ is a bijection.

The combination of these facts proves equality of (4.6) and (4.7), and hence the equality of (4.3) and (4.4).

To conclude the proof, we must show that the remaining part of the exponential term of (4.2) can be drawn into the graph sum. Using the projection formula, this identity is equivalent to showing that

$$(j_{\Gamma_{\delta}})^* \exp\left(\frac{1}{2}\left(-\eta + \sum_{i=1}^n 2a_i\xi_i + a_i^2\psi_i\right)\right) = \prod_{v \in \mathcal{V}(\Gamma_{\delta})} \exp\left(-\frac{1}{2}\eta(v)\right) \prod_{i=1}^n \exp\left(\frac{1}{2}a_i^2\psi_i + a_i\xi_i\right),$$

which immediately reduces to showing

$$(j_{\Gamma_{\delta}})^* \left(-\eta + \sum_{i=1}^n (2a_i\xi_i + a_i^2\psi_i) \right) = -\sum_{v \in \mathcal{V}(\Gamma_{\delta})} \eta(v) + \sum_{i=1}^n (2a_i\xi_i + a_i^2\psi_i).$$

By Lemma 4.2,

$$-\eta + \sum_{i=1}^{n} (2a_i\xi_i + a_i^2\psi_i) = -\pi_*c_1(\mathcal{L}_A)^2.$$

Now, consider the diagram of universal curves

where the left and right squares are cartesian, and the map G is the gluing map identifying sections of $\mathfrak{C}'_{\Gamma_{\delta}} \to \mathfrak{Pic}_{\Gamma_{\delta}}$ corresponding to pairs of half-edges forming an edge. This map G is proper, representable, and birational.

The space $\mathfrak{C}'_{\Gamma_{\delta}}$ is a disjoint union of universal curves

$$\pi'_{\Gamma_{\delta},v}: \mathfrak{C}'_{\Gamma_{\delta},v} \longrightarrow \mathfrak{Pic}_{\Gamma_{\delta}}$$

for $v \in V(\Gamma)$, and the bundle $G^* J^*_{\Gamma_{\delta}} \mathcal{L}_A$ restricted to the component $\mathfrak{C}'_{\Gamma_{\delta},v}$ is equal to the pullback of the line bundle \mathcal{L}_{v,A_v} from the factor $C_{g(v),n(v),\delta(v)}$ (where A_v is the vector formed by numbers a_i for i a marking at v, extended by zero on the half-edges at v). Then, using the projection formula together with Proposition 2.16, we have

$$(j_{\Gamma_{\delta}})^* \pi_* c_1(\mathcal{L}_A)^2 = (\pi_{\Gamma_{\delta}})_* J_{\Gamma_{\delta}}^* c_1(\mathcal{L}_A)^2$$
$$= (G \circ \pi_{\Gamma_{\delta}})_* (G \circ \pi_{\Gamma_{\delta}})^* c_1(\mathcal{L}_A)^2$$
$$= \sum_{v \in V(\Gamma_{\delta})} (\pi'_{\Gamma_{\delta},v})_* c_1(\mathcal{L}_{v,A_v})^2$$
$$= \sum_{v \in V(\Gamma_{\delta})} \eta(v) + \sum_{i=1}^n a_i^2 \psi_i + 2a_i \xi_i,$$

where, for the last equality, we again use Lemma 4.2.

In the case n=0 and d=0, the formula $\mathsf{P}^{\bullet}_{g,\varnothing,0}$ takes a slightly simpler shape: it is the r=0 term of the formula

$$\exp\left(-\frac{1}{2}\eta\right) \times \sum_{\substack{\Gamma_{\delta} \in \mathsf{G}_{g,0,0} \\ w \in \mathsf{W}_{\Gamma_{\delta},r}}} \frac{r^{-h^{1}(\Gamma_{\delta})}}{|\operatorname{Aut}(\Gamma_{\delta})|} j_{\Gamma_{\delta}*}\left[\prod_{e=(h,h')\in \mathrm{E}(\Gamma_{\delta})} \frac{1}{\psi_{h}+\psi_{h'}} \left(1-\exp\left(-\frac{w(h)w(h')}{2}(\psi_{h}+\psi_{h'})\right)\right)\right].$$
(4.8)

As explained in §0.7, the full formula $\mathsf{P}^{\bullet}_{q,A,d}$ can be reconstructed from $\mathsf{P}^{\bullet}_{q,\varnothing,0}$.

4.2. Comparison to Pixton's k-twisted formula

Given $k \ge 0$ and a vector $A = (a_1, ..., a_n) \in \mathbb{Z}^n$ satisfying

$$\sum_i a_i = k(2g-2),$$

let $\tilde{A} = (\tilde{a}_1, ..., \tilde{a}_n)$ be the vector with entries $\tilde{a}_i = a_i + k$. Denote by

$$P_q^{c,k}(\tilde{A}) \in \mathsf{CH}^c(\overline{\mathcal{M}}_{g,n})$$

Pixton's original formula defined in [42, §1.1].

In the k=0 case, $A=\tilde{A}$, and $2^{-g}P_g^{g,0}(\tilde{A})$ is the class originally conjectured by Pixton to equal the double ramification cycle associated with the vector \tilde{A} . Compatibility with the formula for the universal twisted double ramification cycle is given by the following result.

PROPOSITION 4.3. Via the map $\varphi_{\omega_{\pi}^{k}}: \overline{\mathcal{M}}_{g,n} \to \mathfrak{Pic}_{g,n,k(2g-2)}$ associated with the universal data

$$\pi: \mathcal{C}_{g,n} \longrightarrow \overline{\mathcal{M}}_{g,n}, \quad \omega_{\pi}^k \longrightarrow \mathcal{C}_{g,n},$$

the class $\mathsf{P}^{c}_{g,A,k(2g-2)}$ acts as

$$\mathcal{P}_{g,A,k(2g-2)}^{c}(\varphi_{\omega_{\pi}^{k}})([\overline{\mathcal{M}}_{g,n}]) = 2^{-c} \mathcal{P}_{g}^{c,k}(\tilde{A})$$

$$\tag{4.9}$$

for every $c \ge 0$.

Proof. The left-hand side of (4.9) is obtained from $\mathsf{P}_{q,A,k(2q-2)}^c$ by substituting

$$\mathcal{L} = \omega_{\pi}^{\otimes k} \tag{4.10}$$

in the formula and taking the action. A factor 2^{-c} arises on the left-hand side, since all terms in $\mathsf{P}_{g,A,k(2g-2)}^c$ increasing the codimension of the cycle naturally come with corresponding negative powers of 2 (which is placed as a prefactor on the right-hand side in [42, §1.1]).

Under the substitution (4.10), the edge terms and weightings modulo r in the two formulas naturally correspond to each other. Using Proposition 4.1 and Lemma 4.2, we must show that

$$\exp\left(-\frac{1}{2}\pi_*c_1\left(\omega_\pi^{\otimes k}\left(-\sum_{i=1}^n a_i[p_i]\right)\right)^2\right) = \exp\left(-\frac{1}{2}\left(k^2\kappa_1 - \sum_{i=1}^n \tilde{a}_i^2\psi_i\right)\right),$$

where again $[p_i]$ denotes the class of the image of the section $p_i: \overline{\mathcal{M}}_{g,n} \to \mathcal{C}_{g,n}$. Defining

$$\omega_{\pi}^{\log} = \omega_{\pi} \left(\sum_{i} p_{i} \right),$$

we see that

$$c_1 \left(\omega_{\pi}^{\otimes k} \left(-\sum_{i=1}^n a_i[p_i] \right) \right)^2 = \left(k c_1(\omega_{\log}) - \sum_{i=1}^n \tilde{a}_i[p_i] \right)^2$$
$$= k^2 c_1(\omega_{\log})^2 - 2k \sum_{i=1}^n \tilde{a}_i c_1(\omega_{\log})|_{[p_i]} + \sum_{i=1}^n \tilde{a}_i^2[p_i]^2.$$

After pushing forward, the first term gives $k^2 \kappa_1$, the second vanishes (since ω_{\log} restricts to zero on the section p_i), and the third gives $-\sum_i \tilde{a}_i^2 \psi_i$, as desired.

4.3. Comparison to Pixton's formula with targets

Let X be a non-singular projective variety over K. The moduli space $\overline{\mathcal{M}}_{g,n}(X,\beta)$ parameterizes stable maps

$$f: (C, p_1, ..., p_n) \longrightarrow X$$

from genus-g, n-pointed curves C to X of degree $\beta \in H_2(X, \mathbb{Z})$. The moduli space carries a virtual fundamental class

$$[\overline{\mathcal{M}}_{g,n}(X,\beta)]^{\mathsf{vir}} \in \mathsf{CH}_{\mathrm{vdim}(g,n,\beta)}(\overline{\mathcal{M}}_{g,n}(X,\beta)),$$

where

vdim
$$(g, n, \beta) = (\dim X - 3)(1 - g) + \int_{\beta} c_1(X) + n.$$

See [11] for the construction of virtual fundamental classes.

Given the data of a line bundle \mathcal{L} on X and a vector $A = (a_1, ..., a_n) \in \mathbb{Z}^n$ satisfying

$$\int_{\beta} c_1(\mathcal{L}) = \sum_{i=1}^n a_i,$$

a double ramification cycle

$$\mathsf{DR}_{g,A}(X,\mathcal{L}) \in \mathsf{CH}_{\mathrm{vdim}(g,n,\beta)-g}(\overline{\mathcal{M}}_{g,n}(X,\beta))$$

virtually compactifying the locus of maps $f: (C, p_1, ..., p_n) \rightarrow X$ with

$$f^*\mathcal{L} \cong \mathcal{O}_C\left(\sum_{i=1}^n a_i p_i\right)$$

is defined in [43]. Furthermore, the authors define the notion of tautological classes inside the operational Chow ring $\mathsf{CH}^*_{\mathsf{op}}(\overline{\mathcal{M}}_{g,n}(X,\beta))$ of $\overline{\mathcal{M}}_{g,n}(X,\beta)$. The main result of [43] is a Pixton formula for a codimension-g tautological class whose action on $[\overline{\mathcal{M}}_{g,n}(X,\beta)]^{\mathsf{vir}}$ yields $\mathsf{DR}_{g,A}(X,\mathcal{L})$.

We define a morphism

$$\begin{split} \varphi_{\mathcal{L}} \colon \overline{\mathcal{M}}_{g,n,\beta}(X) & \longrightarrow \mathfrak{Pic}_{g,A,d}, \\ f & \longmapsto (C,p_1,...,p_n,f^*\mathcal{L}). \end{split}$$

The compatibility result here is

$$\mathsf{DR}_{g,A}(X,\mathcal{L}) = \varphi_{\mathcal{L}}^* \mathsf{P}_{g,A,d}^g([\overline{\mathcal{M}}_{g,n}(X,\beta)]^{\mathsf{vir}}).$$
(4.11)

The equality follows by an exact matching of the definition of $\mathsf{P}_{g,\mathbf{A},d}^g$ in §0.3.5 (after pullback by $\varphi_{\mathcal{L}}^*$) with the Pixton formula in the main result of [43].

In fact, the compatibility (4.11) represented the starting point for our investigation of the universal twisted double ramification cycle here.

5. Proof of Theorem 0.7

5.1. Overview

We prove here the main result of the paper: for $A = (a_1, ..., a_n) \in \mathbb{Z}^n$ satisfying

$$\sum_{i=1}^{n} a_i = d,$$

the universal twisted double ramification cycle is calculated by Pixton's formula

$$\mathsf{DR}_{g,A}^{\mathsf{op}} = \mathsf{P}_{g,A,d}^g \in \mathsf{CH}_{\mathsf{op}}^g(\mathfrak{Pic}_{g,n,d}).$$

The result is an equality in the operational Chow group, and therefore an equality on every finite-type family of prestable curves. Given a prestable curve $C \rightarrow B$ and a line bundle \mathcal{L} on C of relative degree d, we obtain a map

$$\varphi_{\mathcal{L}}: B \longrightarrow \mathfrak{Pic}_{g,n,d}.$$

We must prove that

$$\mathsf{DR}_{g,A}^{\mathsf{op}}(\varphi_{\mathcal{L}}) = \mathsf{P}_{g,A,d}^{g}(\varphi_{\mathcal{L}}) \colon \mathsf{CH}_{*}(B) \longrightarrow \mathsf{CH}_{*-g}(B).$$
(5.1)

As explained in §0.7, the result for general $A \in \mathbb{Z}^n$ can be reduced to the case n=0, d=0, though the case of arbitrary A will be important in the proof, as we proceed through a sequence of special cases. We recall that this reduction used the invariances II and III for the double ramification cycle and Pixton's formula. Note that these will be proved separately and independent of Theorem 0.7 in §7, so no circular reasoning occurs.

5.2. On an open subset of $\overline{\mathcal{M}}_{q,n}(\mathbb{P}^l,\beta)$

As before, let $A = (a_1, ..., a_n) \in \mathbb{Z}^n$ with

$$\sum_{i=1}^n a_i = d.$$

We consider here the target $X = \mathbb{P}^l$. Let β be the class of d times a line in \mathbb{P}^l . Let

$$\mathcal{C} \longrightarrow \overline{\mathcal{M}}_{g,n}(\mathbb{P}^l,\beta)$$

be the universal curve over the moduli of stable maps to \mathbb{P}^l , let

$$f: \mathcal{C} \longrightarrow \mathbb{P}^l$$

be the universal map, and let $\mathcal{L}=f^*\mathcal{O}_X(1)$.

We have a tautological map

$$\varphi_{\mathcal{L}}: \overline{\mathcal{M}}_{g,n}(\mathbb{P}^l,\beta) \longrightarrow \mathfrak{Pic}_{g,n,d}.$$
(5.2)

We would like to prove an equality of operational classes

$$\varphi_{\mathcal{L}}^* \mathsf{DR}_{g,A}^{\mathsf{op}} = \varphi_{\mathcal{L}}^* \mathsf{P}_{g,A,d}^g \in \mathsf{CH}_{\mathsf{op}}^g(\overline{\mathcal{M}}_{g,n}(\mathbb{P}^l,\beta)).$$

We will apply the main result of [43] which relates the double ramification cycle there to Pixton's formula. However, only the action of $\varphi_{\mathcal{L}}^* \mathsf{P}_{g,A,d}^g$ on the virtual fundamental class $[\overline{\mathcal{M}}_{g,n}(\mathbb{P}^l,\beta)]^{\text{vir}}$ is computed in [43]. Since we are interested here in the full operational class $\varphi_{\mathcal{L}}^* \mathsf{P}_{g,A,d}^g$, our first idea is to restrict to the open locus

$$\overline{\mathcal{M}}_{g,n}(\mathbb{P}^l,\beta)' \hookrightarrow \overline{\mathcal{M}}_{g,n}(\mathbb{P}^l,\beta)$$

where (on each geometric fiber) we have $H^1(C, \mathcal{L}) = 0$.

LEMMA 5.1. On the smooth Deligne–Mumford stack $\overline{\mathcal{M}}_{g,n}(\mathbb{P}^l,\beta)'$, the fundamental and virtual fundamental classes coincide.

Proof. It suffices to show that $H^1(C, f^*T_{\mathbb{P}^l})=0$ on $\overline{\mathcal{M}}_{g,n}(\mathbb{P}^l, \beta)'$. Pulling back the Euler exact sequence on \mathbb{P}^l via f yields

$$0 \longrightarrow \mathcal{O}_C \longrightarrow \bigoplus_{1}^{l+1} f^* \mathcal{O}_{\mathbb{P}^l}(1) \longrightarrow f^* T_{\mathbb{P}^l} \longrightarrow 0.$$
(5.3)

Taking cohomology yields the exact sequence

$$\bigoplus_{1}^{l+1} H^1(C, f^*\mathcal{O}_{\mathbb{P}^l}(1)) \longrightarrow H^1(C, f^*T_{\mathbb{P}^l}) \longrightarrow H^2(C, \mathcal{O}_C).$$
(5.4)

But $H^1(C, f^*\mathcal{O}_{\mathbb{P}^l}(1))=0$ by assumption, and $H^2(C, \mathcal{O}_C)=0$ for dimension reasons. \Box

The next lemma depends on a careful comparison of the logarithmic and rubber approaches to double ramification cycles, which will be postponed to §6.

LEMMA 5.2. Let $\varphi'_{\mathcal{L}}$ be the restriction of $\varphi_{\mathcal{L}}$ to $\overline{\mathcal{M}}_{g,n}(\mathbb{P}^l,\beta)'$. We have an equality of operational classes

$$\varphi_{\mathcal{L}}^{\prime*}\mathsf{DR}_{g,A}^{\mathsf{op}} = \varphi_{\mathcal{L}}^{\prime*}\mathsf{P}_{g,A,d}^{g} \in \mathsf{CH}_{\mathsf{op}}^{g}(\overline{\mathcal{M}}_{g,n}(\mathbb{P}^{l},\beta)').$$
(5.5)

Proof. By Lemma 2.6, the two sides of (5.5) are equal if and only if their actions on the fundamental class $[\overline{\mathcal{M}}_{g,n}(\mathbb{P}^l,\beta)']$ are equal in $\mathsf{CH}_*(\overline{\mathcal{M}}_{g,n}(\mathbb{P}^l,\beta)')$. By (4.11), the action of the right-hand side of (5.5) on

$$[\overline{\mathcal{M}}_{g,n}(\mathbb{P}^l,\beta)'] = [\overline{\mathcal{M}}_{g,n}(\mathbb{P}^l,\beta)']^{\mathsf{vir}}$$

equals the restriction of $\mathsf{DR}_{g,A}(\mathbb{P}^l,\mathcal{L})$ to $\overline{\mathcal{M}}_{g,n}(\mathbb{P}^l,\beta)'$.

The cycle $\mathsf{DR}_{g,A}(\mathbb{P}^l, \mathcal{L})$ is defined in [43] as the pushforward of the virtual fundamental class of the space of rubber maps.⁽²⁸⁾ By Proposition 6.12 of §6.5, the restriction of $\mathsf{DR}_{g,A}(X, \mathcal{L})$ to $\overline{\mathcal{M}}_{g,n}(\mathbb{P}^l, \beta)'$ is equal to $\varphi'^*_{\mathcal{L}}\mathsf{DR}^{\mathsf{op}}_{g,A}([\overline{\mathcal{M}}_{g,n}(\mathbb{P}^l, \beta)'])$.

5.3. For sufficiently positive line bundles

Let $\pi: C \to B$ be an *n*-pointed prestable curve over a scheme of finite type over K. Let \mathcal{L} on C be a line bundle of relative degree d. Let

$$A = (a_1, \dots, a_n) \in \mathbb{Z}^n$$

with $\sum_{i=1}^{n} a_i = d$. The line bundle \mathcal{L} induces a map

$$\varphi_{\mathcal{L}}: B \longrightarrow \mathfrak{Pic}_{q,n,d}.$$

We say that \mathcal{L} is relatively sufficiently positive if \mathcal{L} is relatively base-point free and satisfies $R^1\pi_*\mathcal{L}=0$.

LEMMA 5.3. Let \mathcal{L} be a line bundle which is relatively sufficiently positive. Then, we have an equality

$$\mathsf{DR}_{g,A}^{\mathsf{op}}(\varphi_{\mathcal{L}}) = \mathsf{P}_{g,A,d}^{g}(\varphi_{\mathcal{L}}) \colon \mathsf{CH}_{*}(B) \longrightarrow \mathsf{CH}_{*-g}(B).$$
(5.6)

Proof. For any finite-type scheme B, the union of irreducible components of B maps properly and surjectively to B. Thus, the pushforward from the Chow groups of the irreducible components to that of B is surjective, and hence it suffices to show the equality (5.6) of maps of Chow groups for B irreducible.

By relative sufficient positivity,

 $R\pi_*\mathcal{L} = \pi_*\mathcal{L}$

 $^(^{28})$ Rubber maps will be discussed in §6.3.

is a vector bundle on B of rank N. For a positive integer l, we define

$$E_l = \bigoplus_{1}^{l+1} R\pi_* \mathcal{L}, \tag{5.7}$$

a vector bundle on B of rank r=N(l+1). Let $U_l \subseteq E_l$ denote the open locus of linear systems which are base-point free. Via pullback along $\psi: U_l \to B$, we obtain a map

$$\psi^*: \mathsf{CH}_*(B) \longrightarrow \mathsf{CH}_{*+r}(U_l).$$

We claim that, for $l > \dim B$, the pullback (5.7) is injective. To prove the injectivity, we show that the boundary $E_l \setminus U_l$ has codimension in E_l greater than dim B. Since $E_l \to B$ is flat with irreducible target, it suffices to bound the codimension on each geometric fiber over B: for a prestable curve C/K and a sufficiently positive line bundle \mathcal{L} on C, we must show that the locus in $\bigoplus_{l=1}^{l+1} H^0(C, \mathcal{L})$ consisting of base-point-free linear systems has a complement of codimension greater than dim B.

Since \mathcal{L} is base-point free on C, the dimension of the locus in $\bigoplus_{1}^{l+1} H^0(C, \mathcal{L})$, where the linear system has a base point at some given $p \in C$ is (N-1)(l+1). Hence, as pvaries, the complement of the base-point-free locus in $\bigoplus_{1}^{l+1} H^0(C, \mathcal{L})$ has dimension at most 1+(N-1)(l+1). So the codimension is at least

$$N(l+1) - 1 - (N-1)(l+1) = l.$$

We have a canonical map $g: C \times_B U_l \to \mathbb{P}^l$ with $g^* \mathcal{O}_{\mathbb{P}^l}(1) = \mathcal{L}$, which induces a map

$$U_l \longrightarrow \overline{\mathcal{M}}_{g,n}(\mathbb{P}^l,\beta),$$

which factors via the locus

266

$$\overline{\mathcal{M}}_{q,n}(\mathbb{P}^l,\beta)' \subset \overline{\mathcal{M}}_{q,n}(\mathbb{P}^l,d),$$

where $H^1(C, f^*\mathcal{O}_{\mathbb{P}^l}(1))=0$. By construction, the composition

$$U_l \xrightarrow{\psi} B \xrightarrow{\varphi_{\mathcal{L}}} \mathfrak{Pic}_{g,n,d}$$

then factors through the map

$$\overline{\mathcal{M}}_{g,n}(\mathbb{P}^l,\beta)' \longrightarrow \mathfrak{Pic}_{g,n,d}$$

induced by the line bundle $f^*\mathcal{O}_{\mathbb{P}^l}(1)$ as before. In other words, we have a commutative diagram

Lemma 5.2 then implies that

$$(\mathsf{DR}^{\mathsf{op}}_{g,A} - \mathsf{P}^{g}_{g,A,d})(\varphi_{\mathcal{L}} \circ \psi) \colon \mathsf{CH}_{*}(U_{l}) \longrightarrow \mathsf{CH}_{*-g}(U_{l})$$

$$(5.8)$$

is the zero map, and we conclude the proof of the lemma from the commutative diagram

5.4. With sufficiently many sections

Let $\pi: C \to B$ be an *n*-pointed prestable curve with markings $p_1, ..., p_n$ over a scheme of finite type over K. Let \mathcal{L} on C be a line bundle of relative degree d. Let

$$A = (a_1, \dots, a_n) \in \mathbb{Z}^n$$

with $\sum_{i=1}^{n} a_i = d$. The line bundle \mathcal{L} induces a map

$$\varphi_{\mathcal{L}}: B \longrightarrow \mathfrak{Pic}_{g,n,d}.$$

LEMMA 5.4. For every geometric fiber of C/B, suppose that the complement of the union of irreducible components which carry markings is a disjoint union of trees of non-singular rational curves on which \mathcal{L} is trivial. Then, we have an equality

$$\mathsf{DR}_{g,A}^{\mathsf{op}}(\varphi_{\mathcal{L}}) = \mathsf{P}_{g,A,d}^{g}(\varphi_{\mathcal{L}}) \colon \mathsf{CH}_{*}(B) \longrightarrow \mathsf{CH}_{*-g}(B).$$
(5.9)

Proof. We can choose $A' = (a'_1, ..., a'_n)$ with entries

$$a_i'\!\gg\!0,\quad \sum_{i=1}^n a_i'\!=\!d',$$

large enough so that

$$\mathcal{L}' = \mathcal{L}\bigg(\sum_{i=1}^n a_i' p_i\bigg)$$

is relatively sufficiently positive (by Riemann-Roch for singular curves).

We obtain an associated map

$$\varphi_{\mathcal{L}'}: B \longrightarrow \mathfrak{Pic}_{g,n,d+d'}.$$

By §5.3,

$$\mathsf{DR}_{g,A+A'}^{\mathsf{op}}(\varphi_{\mathcal{L}'}) = \mathsf{P}_{g,A+A',d+d'}^g(\varphi_{\mathcal{L}'}) \colon \mathsf{CH}_*(B) \longrightarrow \mathsf{CH}_{*-g}(B).$$
(5.10)

Invariance III of 0.6 (proven in 7) implies that

$$\begin{split} \mathsf{DR}^{\mathsf{op}}_{g,A+A'}(\varphi_{\mathcal{L}'}) = \mathsf{DR}^{\mathsf{op}}_{g,A}(\varphi_{\mathcal{L}}), \\ \mathsf{P}^{g}_{g,A+A',d+d'}(\varphi_{\mathcal{L}'}) = \mathsf{P}^{g}_{g,A,d}(\varphi_{\mathcal{L}}), \end{split}$$

which together with (5.10) finishes the proof.

5.5. Proof in the general case

To conclude the proof of Theorem 0.7, will use the invariances of §0.6 (proven in §7). As discussed in §5.1, we can reduce to showing the result in the case n=0, d=0.⁽²⁹⁾

Let B be an irreducible scheme of finite type over K. Let $\pi: C \to B$ a prestable curve, and let \mathcal{L} on C be a line bundle of relative degree zero. The line bundle \mathcal{L} induces a map

$$\varphi_{\mathcal{L}}: B \longrightarrow \mathfrak{Pic}_{q,0,0}$$

LEMMA 5.5. There exist an alteration⁽³⁰⁾ $B' \rightarrow B$ and a destabilization

$$C' \longrightarrow C \times_B B' \tag{5.11}$$

such that C' admits sections $p_1, ..., p_m$ and satisfies the following property:

$$(C'/B', p_1, ..., p_m)$$

is a family of m-pointed prestable curves and, for every geometric fiber of C'/B', the complement of the union of irreducible components which carry markings is a disjoint union of trees of non-singular rational curves which are contracted by the morphism (5.11).

268

 $^(^{29})$ In genus g=1, we follow a slightly modified strategy, since there we must avoid the case n=0 for technical reasons (see Remark 2.1). Instead, we can use the invariances to reduce to the case g=1, n=1, and A=(0). All the proofs below generalize in a straightforward way, since the vector A=(0) does not affect the line bundles involved.

 $^(^{30})$ An alteration here is a proper, surjective, generically finite morphism between irreducible schemes.

Proof. We first claim, after an alteration $\widehat{B} \to B$, there exists a multisection⁽³¹⁾

$$Z \subset C_{\widehat{B}} = C \times_B \widehat{B} \longrightarrow \widehat{B}$$

satisfying the following two conditions:

(i) Over the generic point of \widehat{B} , Z is contained in the smooth locus of $C_{\widehat{B}} \to \widehat{B}$.

(ii) Every component of every geometric fiber of $C_{\widehat{B}} \to \widehat{B}$ carries at least two distinct étale multisection points in the smooth locus. In other words, the étale locus of $Z \to \widehat{B}$ meets the smooth locus of every component of every geometric fiber of $C_{\widehat{B}} \to \widehat{B}$ in at least two points.

To prove the above claim, we observe that, for every geometric point b of B, there exists an étale map $U_p \to B$ and a factorization $U_p \to C$ whose image meets the smooth locus of every irreducible component of every geometric fiber in some Zariski neighbourhood $V_b \subseteq B$ of b at least twice. Choose a finite set of b such that the V_b cover B, define U to be the union of the U_b , and define Z' to be the closure of the image of U in C. Then $Z' \to B$ is proper and generically finite. Let

$$\widehat{B} \longrightarrow B$$

be a modification which flattens Z' (see [69]), and let Z be the strict transform of Z' over \hat{B} . Then

$$Z \longrightarrow \widehat{B}$$

is proper, flat, and generically finite, and hence finite—so condition (i) is satisfied. Moreover, U already satisfies condition (ii), and the strict transform of a flat map is just the fiber product, and hence Z also satisfies condition (ii).

Let $\tilde{B} \to \hat{B}$ be an alteration such that over \tilde{B} the multisection Z becomes a disjoint union of sections. In other words, the pullback

$$C_{\widetilde{B}} = C \times_B \widetilde{B} \longrightarrow \widetilde{B}$$

has sections $\tilde{p}_1, ..., \tilde{p}_m$ such that, as a set, the preimage of Z is given by the union of the images of sections $\tilde{p}_1, ..., \tilde{p}_m$. Such a \tilde{B} exists⁽³²⁾ by [44, Lemma 5.6]. We may assume that the sections \tilde{p}_i are pairwise disjoint over the generic point of \tilde{B} .

By assumption (i) above, the family $C_{\tilde{B}} \to \tilde{B}$ with sections $\tilde{p}_1, ..., \tilde{p}_m$ is generically a stable *m*-pointed curve (since every component has at least *two* of the sections). We therefore obtain a rational map

$$\widetilde{B} \dashrightarrow \overline{\mathcal{M}}_{g,m}$$

^{(&}lt;sup>31</sup>) By a multisection of $C_{\hat{B}} \to \hat{B}$, we mean a closed substack $Z \subset C_{\hat{B}}$ such that $Z \to \hat{B}$ is finite and flat.

 $^(^{32})$ The base \hat{B} is excellent, since it is of finite type over a field.

Let $B' \to \widetilde{B}$ be a blow-up resolving the indeterminacy of this map⁽³³⁾



and let $C' \to B'$ with sections $p_1, ..., p_m: B' \to C'$ be the pullback of the universal curve over $\overline{\mathcal{M}}_{q,n}$ to B'. Let

$$C_{B'} = C \times_B B'$$

be the pullback of C/B under

270

$$B' \longrightarrow \widetilde{B} \longrightarrow \widehat{B} \longrightarrow B.$$

Then, we have a map $f: C' \to C_{B'}$ fitting in a commutative diagram



such that f is a partial destabilization. On geometric fibers of $C' \rightarrow B'$, f collapses trees of rational curves to either nodes or coincident sections \tilde{p}_i on the geometric fibers of $C_{B'}$.

To conclude, we must show that, for every geometric point $b \in B'$ and every irreducible component $D \subset C'_b$ which is not contracted by f, we can find a marking

$$p_i(b) \in D.$$

The image of D under f is a component of $(C_{B'})_b$. By condition (ii) above, f(D) has at least one $\tilde{p}_i(b)$ in the smooth locus of f(D) pairwise distinct from all other $\tilde{p}_j(b)$. Since there are no components of C'_b which collapse to $\tilde{p}_i(b)$, we must have $p_i(b) \in D$.

LEMMA 5.6. We have

$$\mathsf{DR}^{\mathsf{op}}_{g,\varnothing}(\varphi_{\mathcal{L}}) = \mathsf{P}^{g}_{g,\varnothing,0}(\varphi_{\mathcal{L}}) \colon \mathsf{CH}_{*}(B) \longrightarrow \mathsf{CH}_{*-g}(B).$$

Proof. We apply Lemma 5.5 to the family C/B to obtain

 $h: B' \longrightarrow B, \quad C' \longrightarrow C_{B'}.$

 $^(^{33})$ As usual, this blowup is constructed by taking the closure of the graph and flattening. Then we check that this ensures the existence of the map to $C_{B'}$ as written below.

Let \mathcal{L}' be the pullback of \mathcal{L} to C'. After applying Lemma 5.4 with $A = \mathbf{0} \in \mathbb{Z}^m$, we obtain

$$\mathsf{DR}_{g,\mathbf{0}}^{\mathsf{op}}(\varphi_{\mathcal{L}'}) = \mathsf{P}_{g,\mathbf{0},d}^{g}(\varphi_{\mathcal{L}'}) \colon \mathsf{CH}_{*}(B') \longrightarrow \mathsf{CH}_{*-g}(B').$$
(5.14)

Since h is proper and surjective, for any $\alpha \in \mathsf{CH}_*(B)$ there exists $\alpha' \in \mathsf{CH}_*(B')$ satisfying $h_*\alpha' = \alpha$. If any operational class maps α' to zero, then it maps α to zero because the operation commutes with h_* .

It therefore suffices to prove that

$$(\mathsf{DR}^{\mathsf{op}}_{g,\varnothing} - \mathsf{P}^{g}_{g,\varnothing,0})(\varphi_{\mathcal{L}}) \circ h_{*}$$

$$(5.15)$$

is the zero map on $CH_*(B')$. By the compatibilities of operational classes, we have

$$(\mathsf{DR}_{g,\varnothing}^{\mathsf{op}}-\mathsf{P}_{g,\varnothing,0}^g)(\varphi_{\mathcal{L}})\circ h_*=h_*(\mathsf{DR}_{g,\varnothing}^{\mathsf{op}}-\mathsf{P}_{g,\varnothing,0}^g)(\varphi_{\mathcal{L}}\circ h),$$

and the proof below will in fact show that

$$(\mathsf{DR}_{g,\varnothing}^{\mathsf{op}} - \mathsf{P}_{g,\varnothing,0}^g)(\varphi_{\mathcal{L}} \circ h) = 0.$$

By (5.14), we need only show

$$\mathsf{DR}_{g,\varnothing}^{\mathsf{op}}(\varphi_{\mathcal{L}}\circ h) = \mathsf{DR}_{g,\mathbf{0}}^{\mathsf{op}}(\varphi_{\mathcal{L}'}): \mathsf{CH}_*(B') \longrightarrow \mathsf{CH}_{*-g}(B'), \tag{5.16}$$

$$\mathsf{P}^{g}_{g,\varnothing,0}(\varphi_{\mathcal{L}}\circ h) = \mathsf{P}^{g}_{g,\mathbf{0},0}(\varphi_{\mathcal{L}'}): \mathsf{CH}_{*}(B') \longrightarrow \mathsf{CH}_{*-g}(B').$$

$$(5.17)$$

For the map $F: \mathfrak{Pic}_{g,n,0} \to \mathfrak{Pic}_{g,0,0}$ forgetting the markings, Invariance II from §0.6 for the double ramification cycle and the Pixton formula shows that we have

$$\mathsf{DR}^{\mathsf{op}}_{g,\mathbf{0}}(\varphi_{\mathcal{L}'}) = \mathsf{DR}^{\mathsf{op}}_{g,\varnothing}(F \circ \varphi_{\mathcal{L}'}) \quad \text{and} \quad \mathsf{P}^{g}_{g,\mathbf{0},0}(\varphi_{\mathcal{L}'}) = \mathsf{P}^{g}_{g,\varnothing,0}(F \circ \varphi_{\mathcal{L}'}).$$

So, we are reduced to showing that

$$\mathsf{DR}_{g,\varnothing}^{\mathsf{op}}(\varphi_{\mathcal{L}}\circ h) = \mathsf{DR}_{g,\varnothing}^{\mathsf{op}}(F\circ\varphi_{\mathcal{L}'}) \quad \text{and} \quad \mathsf{P}_{g,\varnothing,0}^g(\varphi_{\mathcal{L}}\circ h) = \mathsf{P}_{g,\varnothing,0}^g(F\circ\varphi_{\mathcal{L}'}). \tag{5.18}$$

The claims (5.18) follow from Invariance VI of §0.6. As before, let $C_{B'}$ be the pullback of C under h, and let $\mathcal{L}_{B'}$ be the pullback of \mathcal{L} to $C_{B'}$. The map

$$\varphi_{\mathcal{L}} \circ h: B' \longrightarrow \mathfrak{Pic}_{g,0,0}$$

is induced by the data

$$C_{B'} \longrightarrow B', \quad \mathcal{L}_{B'} \longrightarrow C_{B'},$$

whereas $F \circ \varphi_{\mathcal{L}'}: B' \to \mathfrak{Pic}_{q,0,0}$ is induced by

$$C' \longrightarrow B', \quad \mathcal{L}' \longrightarrow C'.$$

By construction, we have a partial destabilization $C' \to C_{B'}$ over B', and the line bundle \mathcal{L}' is the pullback of $\mathcal{L}_{B'}$ under this map. Hence, the equalities (5.18) follow from Invariance VI of §0.6.

6. Comparing rubber and log spaces

6.1. Overview

Our goal here is to compare the stack of stable rubber maps associated with a line bundle \mathcal{L} on a target X (introduced by Li [54] and studied by Graber–Vakil [31]) to the stack $\operatorname{Rub}_{g,A}$ of Marcus–Wise (see §3.3) and our operational class $\operatorname{DR}_{g,A}^{\operatorname{op}}$. Rubber maps are reviewed in §6.3 and connected to the logarithmic space in §6.4. The relationship between the construction of Marcus–Wise and $\operatorname{DR}_{g,A}^{\operatorname{op}}$ is Lemma 6.11 of §6.5.2. The comparison to the class of Graber–Vakil is carried out in [56], in the case where the target X is a point. We require the case where

$$X = \mathbb{P}^l$$
 and $\mathcal{L} = \mathcal{O}(1)$,

but only over the unobstructed locus

$$\overline{\mathcal{M}}_{g,n}(\mathbb{P}^l,d)' \subset \overline{\mathcal{M}}_{g,n}(\mathbb{P}^l,d),$$

see §5.2. We will treat the case of a general non-singular projective target X, since restricting to \mathbb{P}^l provides no simplification (though the unobstructed locus may be rather small for general X). The final comparison result is Proposition 6.12 in §6.5.3.

6.2. Refined definition of the logarithmic rubber space

As described in §3.4, Marcus and Wise define $\operatorname{\mathbf{Rub}}_{g}^{\mathsf{rel}}$ to be the moduli space of pairs (C, P, α) , where P is a tropical line on S and

$$\alpha \colon C \longrightarrow P$$

is an S-morphism such that on each geometric fiber over S the values taken by α on the irreducible components of C are totally ordered in $(\overline{M}_S^{gp})_s$. However, with the above definition, certain key results of their paper (in particular concerning the comparison to spaces of rubber maps) are not correct as stated.

To explain the problem, we restrict to the case where the base C is a geometric log point. Subdividing P at the images of the vertices of C under the map α yields a divided tropical line Q (in the language of [56]). It is asserted in the discussion above [56, Proposition 5.5.2] that the fiber product $C \times_P Q$ is again a log curve over S, which, in general, is not true. For example, take \overline{M}_S to be the sub-monoid of \mathbb{Z}^2 generated by (1,1), (1,0), and (1,-1), and C and α to be as illustrated in Figure 1. In the fiber product, the edge with length (1,0) must be subdivided into two shorter edges, but (1,0)is an irreducible element of \overline{M}_S . In fact, failure of divisibility is the only thing that can go wrong. LEMMA 6.1. Let (C, P, α) be a point of $\operatorname{Rub}_{g}^{\operatorname{rel}}$ over a geometric log point B, and let Q be obtained from P by subdividing at the image of α . Then, the following conditions are equivalent:

(i) The fiber product $C \times_P Q$ is a log curve over B.

(ii) Let e be an edge of Γ_C between vertices u and v (satisfying $\alpha(v) \ge \alpha(u)$) with length $\ell_e \in \overline{M}_S$ and slope

$$\kappa_e = \frac{\alpha(v) - \alpha(u)}{\ell_e}$$

Then, for every $y \in \text{Image}(\alpha)$ with $\alpha(u) < y < \alpha(v)$, the monoid \overline{M}_B contains the element

$$\frac{y - \alpha(u)}{\kappa_e}.$$

 $\mathit{Proof.}$ The characteristic monoid at a singular point with length ℓ_e is given by the monoid

$$\{(a,b) \in \overline{M}_b^2 : \ell_e \mid a-b\}.$$
(6.1)

Taking the fiber product over P with Q subdivides the characteristic monoid at the element

$$\frac{y\!-\!\alpha(u)}{\kappa_e}\!\in\!\overline{M}_B^{\mathrm{gp}}\!\otimes_{\mathbb{Z}}\!\mathbb{Q}.$$

If $(y-\alpha(u))/\kappa_e$ lies in \overline{M}_B , then the fiber product is easily seen to be a log curve. If not, then the subdivision is not even reduced.

Definition 6.2. We define $\widetilde{\mathbf{Rub}}^{\mathsf{rel}}$ to be the full subcategory of $\mathbf{Rub}^{\mathsf{rel}}$ consisting of objects (C, P, α) which, on each geometric fiber over B, satisfy the equivalent conditions of Lemma 6.1. We define $\widetilde{\mathbf{Rub}}$ to be the fiber product of $\widetilde{\mathbf{Rub}}^{\mathsf{rel}}$ over $\mathfrak{Pic}^{\mathsf{rel}}$ with \mathfrak{Pic} .

Remark 6.3. The double ramification cycle $\mathsf{DR}^{\mathsf{op}}$ can be defined as the operational class induced by the map $\mathsf{Rub} \to \mathfrak{Pic}$ following Definition 2.8. Applying the same definition to the composite map $\widetilde{\mathsf{Rub}} \to \mathfrak{Pic}$ yields the same operational class, by Proposition 2.16.

6.3. The stack of prestable rubber maps

Let $\mathfrak{M}(X)$ be the stack of maps from marked prestable curves to X. An S-point of $\mathfrak{M}(X)$ is a pair

$$(C/S, f: C \to X),$$

where C/S is prestable with markings. To simplify notation, we will often suppress the markings.



Figure 1. A point of **Rub**.

The space of rubber maps associated with a line bundle \mathcal{L} on X is summarized in [43]: a map to rubber with target X is a map to a rubber chain of \mathbb{CP}^1 -bundles $\mathbb{P}(\mathcal{O}_X \oplus \mathcal{L})$ over X attached along their zero and ∞ divisors.

To facilitate our comparison, we begin by writing the definition explicitly. Let **P** denote the projective bundle $\mathbb{P}(\mathcal{O}_X \oplus \mathcal{L})$. The map collapsing the fibers,

$$\rho: \mathbf{P} \longrightarrow X \tag{6.2}$$

admits two sections $r_0, r_\infty: X \to \mathbf{P}$ corresponding to \mathcal{O}_X and \mathcal{L} , respectively.

Definition 6.4. An (X, \mathcal{L}) -rubber target $(R/S, \rho, r_0, r_\infty)$ is flat, proper, and finitely presented

$$R \longrightarrow S$$

and a collapsing map $\rho: R \to X_S$ with two sections

$$r_0, r_\infty: X_S \to R$$

satisfying the following properties:

(i) Every geometric fiber R_s is isomorphic over X_s to a finite chain

$$\mathbf{P} \cup \mathbf{P} \cup \dots \cup \mathbf{P} \tag{6.3}$$

with the components attached successively along the respective zero and ∞ divisors. The collapsing maps (6.2) on the components together define

$$\rho_s : R_s \longrightarrow X_s.$$

The zero and ∞ sections of ρ_s are determined, respectively, by the zero section of first component and the ∞ section of last components of the chain (6.3).

(ii) Étale locally near every point $s \in S$, the data of $(R/S, \rho, r_0, r_\infty)$ is pulled back from a versal deformation space described by Li [53] with one dimension for every component of the singular locus of (6.3).

Definition 6.5. The stack $\operatorname{Rub}^{\operatorname{pre}}(X, \mathcal{L})$ of prestable rubber maps to \mathcal{L} is a fibered category over $\mathfrak{M}(X)$ whose fiber over a map $S \to \mathfrak{M}(X)$ consists of three pieces of data:

(i) a prestable curve \widetilde{C}/S and a partial stabilization⁽³⁴⁾ map $\tau: \widetilde{C} \to C_S$ which is allowed to contract genus-zero components with two special points:

- (ii) an (X, \mathcal{L}) -rubber target $(R/S, \rho, r_0, r_\infty)$;
- (iii) a map $\tilde{f}: \widetilde{C} \to R$ for which the following diagram commutes:

The map \tilde{f} in (iii) is finite over the singularities of R/X_S and predeformable.⁽³⁵⁾ Moreover, over each geometric point $s \in S$, the image $\tilde{f}(\tilde{C}_s)$ meets every component of R_s .

An isomorphism between two objects

$$(\widetilde{C} \to C_S, R, r_0, r_\infty, \widetilde{C} \to R)$$
 and $(\widetilde{C}' \to C_S, R', r'_0, r'_\infty, \widetilde{C}' \to R')$

over $S \to \mathfrak{M}(X)$ is given by the data of isomorphisms

$$\widetilde{C}' \xrightarrow{\sim} \widetilde{C}$$

over C_S and

$$R' \xrightarrow{\sim} R$$

over X_S , compatible with the markings and such that the diagram

$$\begin{array}{c} \widetilde{C}' \xrightarrow{\sim} \widetilde{C} \\ \downarrow \\ R' \xrightarrow{\sim} R \end{array}$$

commutes. We leave the definition of the cartesian morphisms to the careful reader.

Suppose now that we fix a genus g and a vector of integers A of length n. We define the stack $\operatorname{\mathbf{Rub}}_{q,A}^{\mathsf{pre}}(X,\mathcal{L})$ with objects being tuples

$$(\tau: (\tilde{C}, p_1, ..., p_n) \to C_S, R/X_S, \tilde{f}: \tilde{C} \to R),$$
(6.5)

where $(\tilde{C}, p_1, ..., p_n)$ is a prestable curve of genus g with n markings. The data (6.5) are as for $\operatorname{Rub}^{\operatorname{pre}}(X, \mathcal{L})$. Moreover, the following statements hold:

• if $a_i > 0$, $p_i \in \widetilde{C}$ is mapped to the zero-divisor with ramification degree a_i ;

• if $a_i < 0$, $p_i \in \widetilde{C}$ is mapped to the ∞ -divisor with ramification degree $-a_i$;

• if $a_i=0, p_i\in \widetilde{C}$ is mapped to the smooth locus of R away from the zero-divisor and from the ∞ -divisor.

 $[\]begin{pmatrix} 34 \end{pmatrix}$ C_S is not necessarily a stable curve. $\begin{pmatrix} 35 \end{pmatrix}$ See [53].
6.4. Comparison to the logarithmic space

The pullback of \mathcal{L} from X to the universal curve over $\mathfrak{M}(X)$ induces a map $\mathfrak{M}(X) \to \mathfrak{Pic}$. The key comparison result is the following.

PROPOSITION 6.6. The stack $\operatorname{\mathbf{Rub}}_{g,A}^{\operatorname{pre}}(X,\mathcal{L})$ is naturally isomorphic to the fiber product of $\widetilde{\operatorname{\mathbf{Rub}}}_{g,A}$ over $\operatorname{\mathfrak{Pic}}$ with $\mathfrak{M}(X)$ along the map induced by \mathcal{L} :

 $\mathbf{Rub}_{g,A}^{\mathsf{pre}}(X,\mathcal{L}) \cong \widetilde{\mathbf{Rub}}_{g,A} \times_{\mathfrak{Pic}} \mathfrak{M}(X).$

Proof. The right-hand side comes with a built-in log structure, but the left-hand side does not. Our isomorphism will be between the underlying stacks. Our proof is based on the discussion above [56, Proposition 5.5.2], and we will use the language of *divided tropical lines* of [56].

We begin by building a map from the right to the left. We are given a log curve C/S, a tropical line \mathcal{P} on S, a map $\alpha: C \to \mathcal{P}$ whose image is totally ordered, and a map $f: C \to X$, such that $f^*\mathcal{L}$ lies in the isomorphism class $\mathcal{O}_C(\alpha)$.

sThe $\mathbb{G}_m^{\text{trop}}$ -torsor \mathcal{P} is rigidified by the least element among the images of the irreducible components of C (here we use the total ordering condition), and hence comes with a canonical \mathbb{G}_m -torsor $P \to \mathcal{P}$ (if we use the rigidification to identify $\mathcal{P} = \mathbb{G}_m^{\text{trop}}$, then $P = \mathbb{G}_m^{\log}$). The pullback $\alpha^* P$ gives a *canonical* \mathbb{G}_m -torsor on C, which is isomorphic to $f^*\mathcal{L}^*$ up to pullback from S. In other words, the bundle $\alpha^* P \otimes f^* \mathcal{L}^{\vee}$ descends to a line bundle on S which we denote \mathcal{M} .

The images of the irreducible components of C yield a subdivision \mathcal{Q} of \mathcal{P} , and we define a destabilization $\widetilde{C} = C \times_{\mathcal{P}} \mathcal{Q}$ of C, which is a log curve over S, by Lemma 6.1. This \mathcal{Q} comes with a canonical \mathbb{G}_m -torsor Q by pulling back P from \mathcal{P} ; this Q is then a 2-marked semistable genus-zero curve by [56, Proposition 5.2.4]. We define an (X, \mathcal{L}) -rubber target R over S by the formula

$$R = \operatorname{Hom}((\mathcal{L} \otimes \mathcal{M})^*, Q).$$

Here, we pull back and take \mathbb{G}_m -equivariant homomorphisms over X_S .

Write $\tilde{f}: \tilde{C} \to X$. We need a predeformable map $\tilde{C} \to R$, equivalently an equivariant logarithmic map $\tilde{f}^* \mathcal{L}^* \to Q$ over X_S . It is enough to give a map $f^* \mathcal{L} \to P$ (since then we can tensor over \mathcal{P} with \mathcal{Q}), which reduces to writing down an element of

$$\operatorname{Hom}_{C}(f^{*}(\mathcal{L}\otimes\mathcal{M}),\alpha^{*}P) = \operatorname{Hom}_{C}(f^{*}\mathcal{L}\otimes f^{*}\mathcal{L}^{\vee}\otimes\alpha^{*}P,\alpha^{*}P)$$
$$= \operatorname{Hom}_{C}(\alpha^{*}P,\alpha^{*}P), \tag{6.6}$$

which contains the identity. The scheme-theoretic map is predeformable as it comes from a logarithmic map, see [48].

Finally we check that no component of \widetilde{C} is mapped to a non-smooth point of R and that every component is hit. The target R is constructed by subdividing $f^*\mathcal{L}$ at images of components of C, and then \widetilde{C} is constructed by subdividing C at points lying over these divisions, so both assertions are clear.

Now, we construct a map from left to right. Given a prestable rubber map to \mathcal{L} over a base S, we first need to equip S with a suitable log structure.

The curve R/X_S is a map $X_S \to \mathfrak{M}_{0,2}^{ss}$, giving a (minimal) log structure on X_S by pullback. Lemma 6.7 below shows that this log structure descends to S. The curve R/X_S now carries the structure of a log curve, and similarly the quotient $[R/\mathbb{G}_m]$ descends to S (again by the Lemma 6.7), determining our tropical line \mathcal{P} —which evidently satisfies the divisibility condition in Lemma 6.1.

It remains to verify that the map $C \to R$ descends to a map $C \to \mathcal{P}$, and that the total ordering condition is satisfied. Write

$$\tau \colon \widetilde{C} \longrightarrow C.$$

By the proof of [56, Proposition 5.5.2], we see that $R \to \tau_* \tau^* R$ is an isomorphism, and hence the map descends as required. The condition that no components are mapped to the nodes implies that the values of α on the irreducible of C are a subset of the irreducible components of R, in particular are totally ordered.

LEMMA 6.7. Let $(R/S, \rho, r_0, r_\infty)$ be an (X, \mathcal{L}) -rubber target. Then, there exists a (minimal) log structure on S such that R/X_S can be equipped with the structure of a log curve making X_S strict over S. The quotient log stack $[R/\mathbb{G}_m]$ descends to a divided tropical line on S.

Proof. The curve R/X_S with markings r_i is prestable, and hence admits a minimal log structure. We must verify that the resulting log structure on X_S descends to S. After a finite extension of K, we may assume that X has a K point, so that

$$\pi: X_S \longrightarrow S$$

admits a section $x: S \to X_S$, and we can equip S with the pullback log structure. It remains to construct an isomorphism $\pi^* x^* M_{X_S} \to M_{X_S}$. We start by building a map from left to right.

We first build a map on the level of characteristic monoids. The characteristic monoid at a geometric point $t \in X_S$ is given by \mathbb{N}^{ℓ} , where ℓ is the length of the chain of projective lines of R over t. Crucially, the irreducible elements of \mathbb{N}^{ℓ} come with a total order, given by proximity of the corresponding singularity to the r_0 marking.

This rigidifies the characteristic monoid, so as we move along the fiber over $\pi(t)$ the characteristic monoids are *canonically* identified. We obtain canonical identifications

$$(\overline{x^*M_{X_S}})_{\pi(t)} \xrightarrow{\sim} (\overline{M}_{X_S})_t,$$

which give an isomorphism

$$\pi^* x^* \overline{M}_{X_S} \xrightarrow{\sim} \overline{M}_{X_S}.$$

To construct an isomorphism of log structures, we will use the perspective of [13, §3.1] that a log structure is a monoidal functor from the groupified characteristic monoid to the stack of line bundles. The rubber target is by definition pulled back from Li's versal deformation spaces, so it suffices to construct our map in that setting. We may therefore assume that S is regular and the locus of non-smooth curves is a reduced divisor in X_S . Since our map will be canonical, we may further shrink S to be atomic.⁽³⁶⁾ Then, $\overline{M}_{X/S}$ is generated by its global sections, and there is a natural isomorphism of sheaves on X_S :

$$\varphi \colon \mathbb{N}^{\ell} \xrightarrow{\sim} \overline{M}_{X/S}$$

where ℓ is the number of singular points in the fiber of C over any point of X_S lying over the closed stratum of S. Given $1 \leq i \leq \ell$, write D_i for the Cartier divisor in X_S , where the singularity at distance i from the first marking persists. Then φ sends the ith generator of \mathbb{N}^{ℓ} to the section corresponding to the line bundle $\mathcal{O}_{X_S}(D_i)$. To build the required map of log structures

$$\pi^* x^* M_{X_S} \xrightarrow{\sim} M_{X_S},$$

we must construct an isomorphism

$$\pi^* x^* \mathcal{O}_{X_S}(D_i) \xrightarrow{\sim} \mathcal{O}_{X_S}(D_i).$$

Condition (i) of Definition 6.4 implies that the underlying point set of D_i is a union of fibers of X_S/S . Since S is regular and X_S is smooth over S, it follows that

$$D_i = \pi^* x^* D_i$$

giving the required isomorphism.

The quotient log stack

$$[R/\mathbb{G}_m]$$

is a divided tropical line on X_S with divisions coming from the divisors D_i . We may identify the underlying tropical line with $\mathbb{G}_m^{\text{trop}}$ by specifying that the smallest element in the sequence of divisions is mapped to zero. We have already established that these divisions D_i descend to S, and hence so does the divided tropical line.

 $^(^{36})$ [2, Definition 2.2.4].

After restriction to the locus where the infinitesimal automorphisms are trivial, we obtain a stable version of Lemma 6.6. Let $\overline{\mathcal{M}}_{g,n}(X,\beta)$ denote the stack of stable maps from *n*-pointed curves to X representing the class β . The line bundle \mathcal{L} determines a map

$$\overline{\mathcal{M}}_{g,n}(X,\beta) \longrightarrow \mathfrak{Pic},$$

and we can pullback $\widetilde{\mathbf{Rub}}_{g,A}$ as before. Let

$$\operatorname{\mathbf{Rub}}_{q,A}(X,\mathcal{L}) \subset \operatorname{\mathbf{Rub}}_{q,A}^{\operatorname{pre}}(X,\mathcal{L})$$

be the locus where the infinitesimal automorphisms are trivial.

LEMMA 6.8. The stack $\operatorname{Rub}_{g,A}(X,\mathcal{L})$ is the fiber product of $\operatorname{Rub}_{g,A}$ over $\operatorname{\mathfrak{Pic}}$ with $\overline{\mathcal{M}}_{q,n}(X,\beta)$ along the map given by \mathcal{L} :

$$\operatorname{\mathbf{Rub}}_{q,A}(X,\mathcal{L}) \cong \widetilde{\operatorname{\mathbf{Rub}}}_{q,A} \times_{\mathfrak{Pic}} \overline{\mathcal{M}}_{q,n}(X,\beta).$$

Next, we will compare the virtual fundamental classes on these spaces. We will carry out the comparison on a smaller open locus.

(i) We let $\overline{\mathcal{M}}_{g,n}(X,\beta)'$ be the open locus of maps $(C, f: C \to X)$ in $\overline{\mathcal{M}}_{g,n}(X,\beta)$ where

$$H^1(C, f^*\mathcal{L}) = 0.$$

(ii) We define

$$\mathbf{Rub}_{g,A}(X,\mathcal{L})' = \mathbf{Rub}_{g,A}(X,\mathcal{L}) \times_{\overline{\mathcal{M}}_{g,n}(X,\beta)} \overline{\mathcal{M}}_{g,n}(X,\beta)'.$$

In §5.2, we considered the case $X = \mathbb{P}^l$ and showed that this unobstructed locus is large enough to control the cycles relevant to Theorem 0.7. For general X, the unobstructed locus might be very small (and possibly empty).

6.5. Comparing the virtual classes

6.5.1. Overview

We begin by briefly discussing of several spaces which will be relevant in setting up the obstruction theories. Let

$$\mathfrak{M}^{\mathsf{ss}}_{q,n} \subset \mathfrak{M}_{g,n}$$

be the semistable locus (where every rational curve has at least two distinguished points). We write \mathscr{T} for the algebraic stack with log structure which parameterizes tropical lines with at least one division. There are natural maps

$$\mathfrak{M}_{0,2}^{ss} \longrightarrow \mathscr{T} \quad \text{and} \quad \mathfrak{M}_{0,2}^{ss} \longrightarrow B\mathbb{G}_m,$$

the former defined by dividing $\mathbb{G}_m^{\text{trop}}$ at 1 and at the smoothing parameters of the nodes, and the latter defined by the normal bundle at the first marking. The induced map

$$\mathfrak{M}_{0,2}^{\mathrm{ss}} \longrightarrow \mathscr{T} \times B\mathbb{G}_m \tag{6.7}$$

is an isomorphism by [1, Proposition 3.3.3].

As **Rub**^{rel} is the moduli stack of tuples $(C, \alpha: C \to \mathcal{P})$, where \mathcal{P} is a tropical line and the images of the irreducible components of C are totally ordered, there is a natural map

$$\operatorname{Rub}^{\operatorname{rel}} \longrightarrow \mathscr{T}$$
 (6.8)

sending $(C, \alpha: C \to \mathcal{P})$ to the tropical line \mathcal{P} with the division given by the images of the irreducible components of C.

We will construct a map

$$\mathbf{Rub}_{g,A}(X,\mathcal{L})' \longrightarrow \mathfrak{M}_{0,2}^{ss} \tag{6.9}$$

lifting the morphism (6.8) by the following argument. A point of $\operatorname{\mathbf{Rub}}_{g,A}(X,\mathcal{L})'$ is a tuple

$$(C, \alpha: C \to \mathcal{P}, f: C \to X),$$

where $f^*\mathcal{L}$ lies in the class⁽³⁷⁾ $[\mathcal{O}_C(\alpha)]$. However, as \mathcal{P} is divided, there is a unique isomorphism $\mathcal{P} \xrightarrow{\sim} \mathbb{G}_m^{\text{trop}}$, where the smallest division maps to zero. The universal \mathbb{G}_m torsor $\mathbb{G}_m^{\log} \to \mathbb{G}_m^{\text{trop}}$ pulls back to a well-defined \mathbb{G}_m -torsor $\mathcal{O}_C^*(\alpha)$ on C, and the difference $f^*\mathcal{L}^* \otimes_{\mathcal{O}_C^*} \mathcal{O}_C^*(-\alpha)$ descends to a \mathbb{G}_m -torsor on S by the construction of $\operatorname{\mathbf{Rub}}_{g,A}(X,\mathcal{L})'$ as a fiber product. The \mathbb{G}_m -torsor on S induces a map $\operatorname{\mathbf{Rub}}_{g,A}(X,\mathcal{L})' \to B\mathbb{G}_m$. Combined with the map $\operatorname{\mathbf{Rub}}_{g,A}(X,\mathcal{L})' \to \mathscr{T}$ via (6.8), we obtain the map (6.9).

The space $\operatorname{\mathbf{Rub}}_{g,A}(X,\mathcal{L})'$ carries three virtual fundamental classes by the following three constructions:

(i) The class $\mathsf{DR}_{g,A}^{\mathsf{op}}(\varphi_{\mathcal{L}})([\overline{\mathcal{M}}_{g,n}(X,\beta)'])$ obtained by applying $\mathsf{DR}_{g,A}^{\mathsf{op}}$ to the (virtual) fundamental class of $\overline{\mathcal{M}}_{g,n}(X,\beta)'$ via the map $\varphi_{\mathcal{L}}$.

(ii) The class obtained from a 2-step obstruction theory described by Marcus and Wise [56] for the map $\operatorname{\mathbf{Rub}}_{g,A}(X,\mathcal{L})' \to \mathfrak{M}_{g,n} \times \mathscr{T}$.

(iii) A class coming from a 2-step obstruction theory studied by Graber and Vakil [31] for the map

$$\operatorname{\mathbf{Rub}}_{g,A}(X,\mathcal{L})' \to \overline{\mathcal{M}}_{g,n}(X,\beta)' \times \mathfrak{M}_{0,2}^{ss}.$$

We will prove that (i)–(iii) are all equal.

 $^(^{37})$ Here, $[\mathcal{O}_C(\alpha)]$ is an equivalence class under isomorphisms and tensoring with pullbacks from S.

6.5.2. $\mathsf{DR}^{\mathsf{op}}$ and the obstruction theory of Marcus–Wise

The obstruction theory of Marcus–Wise is a 2-step obstruction theory, a notion which we now recall. Unless otherwise stated, by *perfect obstruction theory* we mean an obstruction theory which is perfect in amplitude [-1, 0].

Definition 6.9. A 2-step obstruction theory for a map $f\colon X\to S$ consists of a factorization

$$X \longrightarrow Y \longrightarrow S$$

together with perfect relative obstruction theories for X/Y and for Y/S.

A 2-step obstruction theory induces a virtual pullback by composition.⁽³⁸⁾ If S has a fundamental class [S], the virtual pullback of [S] is the virtual fundamental class of X associated with the 2-step obstruction theory.

We first recall the 2-step obstruction theory of [56] in the case when X is a point. We have a diagram

A perfect relative obstruction theory for the horizontal map is given in [56, §5.6.3], and for the vertical map in [56, Proposition 5.6.5.3]; while the reader might expect that these arguments apply to $\mathbf{Rub}_{g,A}$, rather than the root stack $\mathbf{\widetilde{Rub}}_{g,A}$, Marcus and Wise in fact assume in both constructions the divisibility conditions of Lemma 6.1, and hence their constructions in fact apply to $\mathbf{\widetilde{Rub}}_{g,A}$ (and not to $\mathbf{Rub}_{g,A}$). This 2-step obstruction theory coincides with the rubber theory of Graber–Vakil, as shown in [56, §5.6.6]. Moreover, the virtual fundamental class obtained equals the operational class of $\mathbf{Div}_{g,A}$, see [56, Theorem 5.6.1]; again, these results all assume the divisibility condition of Lemma 6.1, and hence apply to $\mathbf{\widetilde{Rub}}_{g,A}$ in place of $\mathbf{Rub}_{g,A}$.

Returning to the case of arbitrary (X, \mathcal{L}) , we can construct a similar commutative diagram

$$\mathbf{Rub}_{g,A}(X,\mathcal{L})' \longrightarrow \mathbf{\widetilde{Rub}}_{g,A}$$

$$(6.11)$$

$$\mathfrak{M}_{g,n} \times \mathscr{T}.$$

 $^(^{38})$ A 2-step obstruction theory also induces a perfect obstruction theory for X/S in amplitude [-2,0], but we will not use the latter construction.

The vertical map is unchanged, and so again has a perfect relative obstruction theory by [56, Proposition 5.6.5.3]; in fact the morphism is a local complete intersection, and the obstruction theory of [56, Proposition 5.6.5.3] is just the relative tangent complex.

We need to supply a perfect obstruction theory for the horizontal map

$$\mathbf{Rub}_{q,A}(X,\mathcal{L})' \longrightarrow \mathbf{Rub}_{q,A},$$

which we can factor as

$$\mathbf{Rub}_{g,A}(X,\mathcal{L})' \longrightarrow \overline{\mathcal{M}}_{g,n}(X,\beta)' \times_{\mathfrak{M}_{g,n}} \widetilde{\mathbf{Rub}}_{g,A} \longrightarrow \widetilde{\mathbf{Rub}}_{g,A}.$$
(6.12)

The second map is a base change of the unobstructed map $\overline{\mathcal{M}}_{g,n}(X,\beta)' \to \mathfrak{M}_{g,n}$, and hence is unobstructed. For the first map, consider the pullback square

The right vertical arrow is a section of a base change of the smooth morphism

$$\mathfrak{Pic}_{q,n} \longrightarrow \mathfrak{M}_{g,n},$$

and as such is lci and has a perfect relative obstruction theory given by the relative tangent complex $R^1\pi_*\mathcal{O}_C$. Pullback yields a corresponding perfect obstruction theory for the left vertical arrow. This gives a 2-step obstruction theory for the composite map

$$\operatorname{\mathbf{Rub}}_{g,A}(X,\mathcal{L})' \longrightarrow \operatorname{\mathbf{Rub}}_{g,A},$$

from which we obtain a virtual fundamental class following [55].

The discussion here is a very slight generalization of the obstruction theory constructed in [56, Proposition 5.6.3.1].

Definition 6.10. The 2-step obstruction theory for the diagonal map of (6.11),

$$\operatorname{\mathbf{Rub}}_{q,A}(X,\mathcal{L})' \longrightarrow \mathfrak{M}_{q,n} \times \mathscr{T}$$

is the Marcus-Wise obstruction theory.

LEMMA 6.11. The pushforward along

$$\psi: \mathbf{Rub}_{g,A}(X, \mathcal{L})' \longrightarrow \overline{\mathcal{M}}_{g,n}(X, \beta)'$$

of the virtual fundamental class of the Marcus–Wise theory on $\operatorname{Rub}_{g,A}(X,\mathcal{L})'$ equals the class $\operatorname{DR}_{g,A}^{\operatorname{op}}(\varphi_{\mathcal{L}})([\overline{\mathcal{M}}_{g,n}(X,\beta)'])$ obtained via the map

$$\varphi_{\mathcal{L}}: \overline{\mathcal{M}}_{g,n}(X,\beta)' \longrightarrow \mathfrak{Pic}_{g,n}.$$

Proof. From $\varphi_{\mathcal{L}}$, we obtain maps

$$\begin{split} \varphi'_{\mathcal{L}} &: \overline{\mathcal{M}}_{g,n}(X,\beta)' \longrightarrow \overline{\mathcal{M}}_{g,n}(X,\beta)' \times_{\mathfrak{M}_{g,n}} \mathfrak{Pic}_{g,n}, \\ \varphi''_{\mathcal{L}} &: \overline{\mathcal{M}}_{g,n}(X,\beta)' \longrightarrow \overline{\mathcal{M}}_{g,n}(X,\beta)' \times \mathfrak{Pic}_{g,n}. \end{split}$$

Both are lci morphisms because $\mathfrak{Pic}_{g,n}/\mathfrak{M}_{g,n}$ is smooth. By Definition 2.8 and §3.6, we have

$$\mathsf{DR}_{g,A}^{\mathsf{op}}(\varphi_{\mathcal{L}})([\overline{\mathcal{M}}_{g,n}(X,\beta)']) = \psi_*(\varphi_{\mathcal{L}}')^! [\overline{\mathcal{M}}_{g,n}(X,\beta)' \times \widetilde{\mathbf{Rub}}_{g,A}] \\ = \psi_*(\varphi_{\mathcal{L}}')^! [\overline{\mathcal{M}}_{g,n}(X,\beta)' \times_{\mathfrak{M}_{g,n}} \widetilde{\mathbf{Rub}}_{g,A}].$$

The virtual fundamental class of the Marcus–Wise theory is the virtual pullback of the fundamental class of $\mathfrak{M}_{g,n} \times \mathscr{T}$ along the composition

$$\mathbf{Rub}_{g,A}(X,\mathcal{L})' \xrightarrow{(1)} \overline{\mathcal{M}}_{g,n}(X,\beta)' \times_{\mathfrak{M}_{g,n}} \widetilde{\mathbf{Rub}}_{g,A} \xrightarrow{(2)} \widetilde{\mathbf{Rub}}_{g,A} \xrightarrow{(3)} \mathfrak{M}_{g,n} \times \mathscr{T}.$$
(6.14)

The map (3) is lei and the obstruction theory is the relative tangent complex, so the pullback of the fundamental class is the fundamental class of $\widetilde{\mathbf{Rub}}_{g,A}$. The map (2) is unobstructed, so the (virtual) pullback of the fundamental class is again the fundamental class. The obstruction theory of the map (1) is defined by pulling back the relative tangent complex of the lci morphism

$$\overline{\mathcal{M}}_{g,n}(X,\beta)' \longrightarrow \overline{\mathcal{M}}_{g,n}(X,\beta)' \times_{\mathfrak{M}_{g,n}} \mathfrak{Pic}_{g,n}$$

via the pullback square

so the virtual pullback of the fundamental class of $\overline{\mathcal{M}}_{g,n}(X,\beta)' \times_{\mathfrak{M}_{g,n}} \mathbf{\widehat{Rub}}_{g,A}$ is equal to the Gysin pullback

$$(\varphi_{\mathcal{L}}')^{!}[\overline{\mathcal{M}}_{g,n}(X,\beta)' \times_{\mathfrak{M}_{g,n}} \widetilde{\mathbf{Rub}}_{g,A}].$$

6.5.3. Marcus-Wise and Graber-Vakil

As recalled above, Marcus–Wise define a 2-step obstruction theory for the map

$$\operatorname{\mathbf{Rub}}_{q,A}(X,\mathcal{L})' \longrightarrow \mathfrak{M}_{q,n} \times \mathscr{T}.$$

Graber and Vakil consider an obstruction theory for the map

$$\mathbf{Rub}_{g,A}(X,\mathcal{L})' \longrightarrow \overline{\mathcal{M}}_{g,n}(X,\beta)' \times \mathfrak{M}_{0,2}^{\mathsf{ss}}.$$
(6.16)

We wish to show an equality of the corresponding virtual fundamental classes on $\operatorname{\mathbf{Rub}}_{q,A}(X,\mathcal{L})'$. Since

$$\mathfrak{M}_{0,2}^{ss} = \mathscr{T} \times B\mathbb{G}_m,$$

and that the maps $\overline{\mathcal{M}}_{g,n}(X,\beta)' \to \mathfrak{M}_{g,n}$ and $B\mathbb{G}_m \to \operatorname{Spec} K$ are unobstructed, we have an unobstructed map

$$\overline{\mathcal{M}}_{q,n}(X,\beta)' \times \mathfrak{M}_{0,2}^{\mathsf{ss}} \longrightarrow \mathfrak{M}_{q,n} \times \mathfrak{M}_{0,2}^{\mathsf{ss}} \longrightarrow \mathfrak{M}_{q,n} \times \mathscr{T}.$$

Our final step is therefore to compare the obstruction theories (and thereby the corresponding virtual pullbacks) between Marcus–Wise and Graber–Vakil [31], [54]. We will match the obstruction spaces when the base S is a point. The full matching of deformation theories is similar and will be treated in [36]. The claims are also required for [56].

Suppose we are given the data of a point in $\operatorname{Rub}_{q,A}(X,\mathcal{L})'(S)$:

$$(\tau: \widetilde{C} \to C_S, R/X_S, \varphi: \widetilde{C} \to R, f: C_S \to X_S, p_1, \dots, p_n, f^*\mathcal{L} \xrightarrow{\sim} \mathcal{O}_C(\alpha)).$$
(6.17)

PROPOSITION 6.12. The restriction to $\overline{\mathcal{M}}_{g,n}(X,\beta)'$ of the class $\mathsf{DR}_{g,A}(X,\mathcal{L})$ of [43] is equal to the class obtained by letting $\mathsf{DR}_{g,A}^{\mathsf{op}}$ act on the fundamental class of $\overline{\mathcal{M}}_{g,n}(X,\beta)'$ via the map induced by \mathcal{L} .

Proof. The primary obstruction of Graber–Vakil lies in

$$H^{0}(\widetilde{C}, \varphi^{-1} \mathcal{E}\mathrm{xt}^{1}(\Omega_{R/X_{S}}(\log D), \mathcal{O}_{R})).$$
(6.18)

Here, D is the divisor on R given by the sum of the two markings r_0 and r_{∞} , and $\Omega_{R/X_S}(\log D)$ is the sheaf of relative 1-forms on R/X_S allowed logarithmic poles along D (a coherent sheaf on R). The obstruction space (6.18) is isomorphic to the product of the deformation spaces of the nodes of R and coincides with the obstruction space for the map

$$\operatorname{\mathbf{Rub}}_{g,A} \longrightarrow \mathfrak{M}_{g,n} \times \mathscr{T}$$

(the vertical arrow in (6.11)), coming from [56, Proposition 5.6.5.3] (where they assume the and divisibility conditions of Lemma 6.1, hence the results apply to $\widetilde{\mathbf{Rub}}_{g,A}$ and not to $\mathbf{Rub}_{g,A}$).

Suppose that the primary obstruction vanishes. Denote by

$$T_{R/X_S} = \mathcal{H}om_{\mathcal{O}_R}(\Omega_{R/X_S}, \mathcal{O}_R)$$

the relative tangent sheaf. There is a secondary obstruction in

$$H^1(C,\varphi^{\dagger}(T_{R/X_S})), \tag{6.19}$$

where the φ^{\dagger} is the torsion-free part of φ^* , see [31] and [56]. The obstruction space (6.19) is the image of the obstruction space

$$H^1(C_S, f^*T_R) = H^1(C_S, \mathcal{O}_{C_S})$$

for the map

$$\operatorname{\mathbf{Rub}}_{g,A}(X,\mathcal{L})' \longrightarrow \widetilde{\operatorname{\mathbf{Rub}}}_{g,A}$$

(the horizontal arrow in (6.11)), coming from [56, Proposition 5.6.3.1].

When both of these obstructions vanish, the deformations are a torsor under

$$H^0(C,\varphi^{\dagger}T_{R/X_S}),$$

an extension of the first term of the obstruction complex in [56, Proposition 5.6.5.3] by the first term of the obstruction complex of [56, Proposition 5.6.4.1]. The comparison of the obstruction theories is complete. \Box

7. Invariance properties

7.1. Overview

We prove here the invariance properties of the universal twisted double ramification cycle as presented in $\S0.6$.

We start with an object of $\varphi_{\mathcal{L}}: \mathcal{S} \to \mathfrak{Pic}_{g,n,d}$ given by a flat family of prestable *n*-pointed genus-*g* curves together with a line bundle of relative degree *d*,

$$\pi: \mathcal{C} \longrightarrow \mathcal{S}, \quad p_1, ..., p_n: \mathcal{S} \longrightarrow \mathcal{C}, \quad \mathcal{L} \longrightarrow \mathcal{C}.$$
 (7.1)

Let $\mathsf{DR}_{g,A,\mathcal{L}}^{\mathsf{op}} = \varphi_{\mathcal{L}}^* \mathsf{DR}_{g,A}^{\mathsf{op}} \in \mathsf{CH}_{\mathsf{op}}^g(\mathcal{S})$ be the twisted double ramification cycle associated with the above family (7.1) and the vector

$$A = (a_1, ..., a_n), \quad d = \sum_{i=1}^n a_i.$$

Theorem 0.7 asserts that $\mathsf{DR}_{q,A}^{\mathsf{op}}$ is equal to the tautological class

$$\mathsf{P}^{g}_{g,A,d} \in \mathsf{CH}^{g}_{\mathsf{op}}(\mathfrak{Pic}_{g,n,d}).$$

If we assume Theorem 0.7, we have a choice of proving the invariance properties either for $\mathsf{DR}_{g,A}^{\mathsf{op}}$ or for the formula in tautological classes. In fact, since both sides of Invariances II, III, and VI are used *in the proof* of Theorem 0.7, we will have to prove these two sides separately in each of these cases. In fact, we will do this for all the invariances, as each side yields interesting perspectives. Also, we will show the invariances of Pixton's formula hold not just for the codimension-g part $\mathsf{P}_{g,A,d}^g$, but for the full mixed degree class

$$\mathsf{P}^{\scriptscriptstyle\bullet}_{g,A,d} \! \in \! \prod_{c=0}^{\infty} \mathsf{CH}^{c}_{\mathsf{op}}(\mathfrak{Pic}_{g,n,d}).$$

7.2. Proof of Invariance I (Dualizing)

We want to show the invariance

$$\mathsf{DR}_{g,-A,\mathcal{L}^*}^{\mathsf{op}} = \varepsilon^* \mathsf{DR}_{g,A,\mathcal{L}^*}^{\mathsf{op}}$$

where $\varepsilon:\mathfrak{Pic}_{g,n,-d}\to\mathfrak{Pic}_{g,n,d}$ is the natural map obtained via dualizing the line bundle. It is enough to show the invariance

$$\mathsf{DR}_{g,-A}^{\mathsf{op}} = \varepsilon^* \mathsf{DR}_{g,A}^{\mathsf{op}} \tag{7.2}$$

of the universal twisted double ramification cycles. The invariance (7.2) can be deduced by applying Lemma 2.17 to the following commutative diagram of morphisms, where the horizontal morphisms are the corresponding Abel–Jacobi maps and the vertical morphisms are isomorphisms:

$$\begin{array}{c} \mathbf{Div}_{g,A} \longrightarrow \mathfrak{Pic}_{g,n,d} \\ & \downarrow^{\hat{\varepsilon}} \qquad \qquad \downarrow^{\varepsilon} \\ \mathbf{Div}_{g,-A} \longrightarrow \mathfrak{Pic}_{g,n,-d}. \end{array}$$

Here, in the language of §3.3, the morphism $\hat{\varepsilon}$ is induced by the natural map

$$\pi_*\overline{M}_C^{\rm gp}/\overline{M}_S^{\rm gp} \longrightarrow \pi_*\overline{M}_C^{\rm gp}/\overline{M}_S^{\rm gp}$$

given by inversion in \overline{M}_C^{gp} .

We now prove the invariance

$$\mathsf{P}^{\bullet}_{g,-A,-d} = \varepsilon^* \mathsf{P}^{\bullet}_{g,A,d} \tag{7.3}$$

286

using the formulas for these cycles from Proposition 4.1. The equality is then implied by the following observations:

• We write

$$\mathcal{L}_A = \mathcal{L}\left(-\sum_{i=1}^n a_i p_i\right)$$

for the twisted universal line bundle on the universal curve $\pi: \mathfrak{C} \to \mathfrak{Pic}_{g,n,d}$, and we use Lemma 4.2 to obtain

$$\varepsilon^{*} \left(-\eta + \sum_{i=1}^{n} 2a_{i}\xi_{i} + a_{i}^{2}\psi_{i} \right) = -\varepsilon^{*}\pi_{*}c_{1}(\mathcal{L}_{A})^{2} = -\pi_{*}c_{1}((\mathcal{L}_{A})^{*})^{2}$$
$$= -\pi_{*}(-c_{1}(\mathcal{L}_{A}))^{2} = -\pi_{*}c_{1}(\mathcal{L}_{A})^{2}$$
$$= -\eta + \sum_{i=1}^{n} 2a_{i}\xi_{i} + a_{i}^{2}\psi_{i}.$$

• Given a prestable graph Γ_{δ} describing a stratum in $\mathfrak{Pic}_{g,n,-d}$, the map ε sends this stratum isomorphically to the stratum of $\Gamma_{-\delta}$ (with an associated commutative diagram of gluing morphisms over $\mathfrak{M}_{g,n}$). Combined with the equality $c_A(\Gamma_{\delta}) = c_{-A}(\Gamma_{-\delta})$ for $\Gamma_{\delta} \in \mathsf{G}^{\mathsf{se}}_{g,n,d}$, we see that the first line of formula (4.2) for $\mathsf{P}^{\bullet}_{g,A,d}$ has the desired invariance.

• For the sum over graphs and weightings, we clearly have $h^1(\Gamma_{\delta}) = h^1(\Gamma_{-\delta})$ and $\operatorname{Aut}(\Gamma_{\delta}) = \operatorname{Aut}(\Gamma_{-\delta})$. Moreover, we have a natural bijection of the admissible weightings modulo r,

$$\begin{split} \mathsf{W}_{\Gamma_{\delta},r} &\longrightarrow \mathsf{W}_{\Gamma_{-\delta},r}, \\ & w \longmapsto (h \mapsto r - w(h) \ \mathrm{mod} \ r). \end{split}$$

The map of weightings leaves the edge terms of the formula (4.2) invariant since they only depend on products w(h)w(h') for and edge (h, h')—which are of the form a(r-a) and thus sent to (r-a)a.

Therefore, the formula of Proposition 4.1 applied to the two sides of (7.3) yields the same result.

7.3. Proof of Invariance II (Unweighted markings)

Assume we have an additional section $p_{n+1}: \mathcal{S} \to \mathcal{C}$ of π which yields an object of $\mathfrak{Pic}_{q,n+1,d}$,

$$\pi: \mathcal{C} \longrightarrow \mathcal{S}, \quad p_1, \dots, p_n, p_{n+1}: \mathcal{S} \longrightarrow \mathcal{C}, \quad \mathcal{L} \longrightarrow \mathcal{C}.$$

$$(7.4)$$

Then, for the vector $A_0 \in \mathbb{Z}^{n+1}$ obtained by appending zero (as the last coefficient) to A, we want to show the invariance

$$\mathsf{DR}_{g,A_0,\mathcal{L}}^{\mathsf{op}} = \mathsf{DR}_{g,A,\mathcal{L}}^{\mathsf{op}}.$$
(7.5)

For the map $\mathfrak{M}_{g,n+1} \to \mathfrak{M}_{g,n}$ induced by forgetting the last marking, we have a diagram of cartesian squares



where the morphism F is syntomic. In particular, for $\mathbf{0} \in \mathbb{Z}^n$ the zero vector, the stack $\mathbf{Div}_{g,\mathbf{0}}$ can be obtained by pulling back $\mathbf{Div}_{g,\emptyset}$ from $\mathfrak{Pic}_{g,0,0}$. Then, as a consequence of the above cartesian square and the definition of the double ramification cycle, Lemma 2.17 yields

$$F^*\mathsf{DR}^{\mathsf{op}}_{q,A} = \mathsf{DR}^{\mathsf{op}}_{q,A_0}.$$
(7.7)

Since the morphisms

288

$$\mathcal{S} \longrightarrow \mathfrak{Pic}_{g,n,d} \quad \text{and} \quad \mathcal{S} \longrightarrow \mathfrak{Pic}_{g,n+1,d}$$

used to define $\mathsf{DR}^{\mathsf{op}}_{g,A,\mathcal{L}}$ and $\mathsf{DR}^{\mathsf{op}}_{g,A_0,\mathcal{L}}$ fit in a diagram



the equation (7.7) immediately proves the invariance (7.5).

We now prove the corresponding invariance

$$F^*\mathsf{P}^{\bullet}_{g,A,d} = \mathsf{P}^{\bullet}_{g,A_0,d} \tag{7.8}$$

of Pixton's formula. First, since the map F does not change the curve or the line bundle, we have a cartesian diagram



The universal line bundles $\mathcal{L}_{g,n,d}$ and $\mathcal{L}_{g,n+1,d}$ on $\mathfrak{C}_{g,n,d}$ and $\mathfrak{C}_{g,n+1,d}$ satisfy

$$\widehat{F}^*\mathcal{L}_{g,n,d} = \mathcal{L}_{g,n+1,d}$$

Similarly, for the canonical line bundles of π and π' we have $\widehat{F}^*\omega_{\pi} = \omega_{\pi'}$. Combining these facts, we see that F pulls back the operational classes η , ξ_i , and ψ_i on $\mathfrak{Pic}_{g,n,d}$ to the corresponding classes on $\mathfrak{Pic}_{g,n+1,d}$.

Next, given a graph $\Gamma_{\delta} \in \mathsf{G}_{g,n,d}$, we have a fiber diagram

where, for $v \in V(\Gamma)$, we denote by $\Gamma_{v,\delta} \in \mathsf{G}_{g,n+1,d}$ the graph obtained from Γ_{δ} by adding marking n+1 at v and leaving the remaining data fixed.

Using the expression (4.2) given in Proposition 4.1, we conclude the proof of (7.8) by the following observations:

• The invariance of η , ξ_i , and ψ_i under F implies that

$$F^*\left(-\eta + \sum_{i=1}^n 2a_i\xi_i + a_i^2\psi_i\right) = -\eta + \sum_{i=1}^{n+1} 2a_i\xi_i + a_i^2\psi_i,$$

where we use $a_{n+1}=0$.

• From the fiber diagram (7.9) and the equality $c_A(\Gamma_{\delta}) = c_{A_0}(\Gamma_{v,\delta})$ for $\Gamma_{\delta} \in \mathsf{G}_{g,n,d}^{\mathsf{se}}$ and any $v \in \mathcal{V}(\Gamma)$, the sum over the terms $c_A(\Gamma_{\delta})[\Gamma_{\delta}]$ pulls back correctly.

• For all $\Gamma_{\delta} \in \mathsf{G}_{g,n,d}$ and $v \in V(\Gamma)$, we have $h^1(\Gamma_{\delta}) = h^1(\Gamma_{v,\delta})$ since the Betti number is independent of the position of the markings. Moreover, the automorphism group $\operatorname{Aut}(\Gamma_{\delta})$ acts on $V(\Gamma_{\delta})$ and, by the orbit-stabilizer formula, the size of the orbit $\operatorname{Aut}(\Gamma_{\delta}) \cdot v$ of vand the size of its stabilizer $\operatorname{Aut}(\Gamma_{\delta})_v$ satisfy

$$|\operatorname{Aut}(\Gamma_{\delta}) \cdot v| = \frac{|\operatorname{Aut}(\Gamma_{\delta})|}{|\operatorname{Aut}(\Gamma_{\delta})_{v}|}.$$
(7.10)

The stabilizer $\operatorname{Aut}(\Gamma_{\delta})_v$ is exactly equal to the automorphism group $\operatorname{Aut}(\Gamma_{v,\delta})$ of the graph $\Gamma_{v,\delta}$, since the marking n+1 at v forces this vertex to be fixed.

• As Γ_{δ} runs through $\mathsf{G}_{g,n,d}^{\mathsf{nse}}$, the graphs $\Gamma_{v,\delta}$ run through $\mathsf{G}_{g,n+1,d}^{\mathsf{nse}}$. The equality (7.10) precisely implies that the corresponding graph sums (weighted by the inverse size of automorphism groups) correspond to each other under pullback by F.

• Finally, the weightings $W_{\Gamma_{\delta},r}$ and $W_{\Gamma_{v,\delta},r}$ are naturally bijective.

Combining the above observations, we see that also the second line of (4.2) transforms under pullback of F as expected.

289

7.4. Proof of Invariance III (Weight translation)

Let $B = (b_1, ..., b_n) \in \mathbb{Z}^n$ satisfy $\sum_{i=1}^n b_i = e$. Then, for the family

$$\pi: \mathcal{C} \longrightarrow \mathcal{S}, \quad p_1, \dots, p_n: \mathcal{S} \longrightarrow \mathcal{C}, \quad \mathcal{L}\left(\sum_{i=1}^n b_i p_i\right) \longrightarrow \mathcal{C},$$
 (7.11)

defining an object of $\mathfrak{Pic}_{q.n.d+e},$ we want to show the invariance

$$\mathsf{DR}_{g,A+B,\mathcal{L}(\sum_{i}b_{i}p_{i})}^{\mathsf{op}} = \mathsf{DR}_{g,A,\mathcal{L}}^{\mathsf{op}}.$$
(7.12)

To show this, consider the smooth map

$$\tau_B:\mathfrak{Pic}_{g,n,d}\longrightarrow\mathfrak{Pic}_{g,n,d+e},$$
$$\mathcal{L}\longmapsto\mathcal{L}\bigg(\sum_{i=1}^n b_i p_i\bigg).$$

over $\mathfrak{M}_{g,n}$. We have a natural cartesian diagram

$$\begin{aligned}
\mathbf{Div}_{g,A} &\longrightarrow \mathbf{Div}_{g,A+B} \\
& \downarrow & \downarrow \\
\mathfrak{Pic}_{g,n,d} &\xrightarrow{\tau_B} \mathfrak{Pic}_{g,n,d+e}.
\end{aligned} \tag{7.13}$$

In particular, for any ramification data A, we can obtain $\operatorname{Div}_{g,A}$ from $\operatorname{Div}_{g,0}$ by such translations. The diagram above together with Lemma 2.17 implies

$$\tau_B^* \mathsf{DR}_{g,A+B}^{\mathsf{op}} = \mathsf{DR}_{g,A}^{\mathsf{op}}.$$
(7.14)

Since the morphisms

$$\mathcal{S} \longrightarrow \mathfrak{Pic}_{g,n,d} \quad \text{and} \quad \mathcal{S} \longrightarrow \mathfrak{Pic}_{g,n,d+e}$$

used to define $\mathsf{DR}^{\mathsf{op}}_{g,A,\mathcal{L}}$ and $\mathsf{DR}^{\mathsf{op}}_{g,A+B,\mathcal{L}(\sum_i b_i p_i)}$ fit in a diagram



the equation (7.14) immediately proves the invariance (7.12).

Now, we prove the invariance

$$\tau_B^* \mathsf{P}^{\bullet}_{g,A+B,d+e} = \mathsf{P}^{\bullet}_{g,A,d} \tag{7.15}$$

for Pixton's formula. Recall the notation

$$\mathcal{L}_A = \mathcal{L}\left(-\sum_i a_i[p_i]\right)$$

for the twisted universal line bundles from Lemma 4.2, observe that in the cartesian diagram

$$\begin{array}{c} \mathfrak{C}_{g,n,d} \xrightarrow{\tau_B} \mathfrak{C}_{g,n,d+e} \\ \\ \downarrow^{\pi'} \qquad \qquad \downarrow^{\pi} \\ \mathfrak{Pic}_{g,n,d} \xrightarrow{\tau_B} \mathfrak{Pic}_{g,n,d+e} \end{array}$$

we have

$$\hat{\tau}_B^* \mathcal{L}_{A+B} = \mathcal{L}_A.$$

By Lemma 4.2, we have

$$\tau_B^* \left(-\eta + \sum_{i=1}^n 2(a_i + b_i)\xi_i + (a_i + b_i)^2 \psi_i \right) = -\tau_B^* \pi_* c_1 (\mathcal{L}_{A+B})^2$$
$$= -\pi_*' c_1 (\mathcal{L}_A)^2 = -\eta + \sum_{i=1}^n 2a_i \xi_i + a_i^2 \psi_i,$$

which shows the compatibility of the first part of formula (4.2) for $\mathsf{P}^{\bullet}_{g,A+B,d+e}$.

For the second part, we can combine the exponential of the graph sum over $G_{g,n,d+e}^{se}$ with the graph sum over $G_{g,n,d+e}^{nse}$ in (4.2) to recover the sum

$$\sum_{\substack{\Gamma_{\delta} \in \mathsf{G}_{g,n,d+e} \\ w \in \mathsf{W}_{\Gamma_{\delta},r}}} \frac{r^{-h^{1}(\Gamma_{\delta})}}{|\operatorname{Aut}(\Gamma_{\delta})|} j_{\Gamma_{\delta}*} \left[\prod_{e=(h,h')\in \mathsf{E}(\Gamma_{\delta})} \frac{1}{\psi_{h} + \psi_{h'}} \left(1 - \exp\left(-\frac{w(h)w(h')}{2}(\psi_{h} + \psi_{h'})\right) \right) \right]$$

$$(7.16)$$

over all graphs in $G_{g,n,d+e}$, as in the proof of Proposition 4.1. It will be more convenient to simply show the compatibility of the full graph sum (7.16) under pullback by τ_B .

Given a graph $\Gamma_{\delta} \in \mathsf{G}_{g,n,d+e}$, denote by $\delta^B : \mathcal{V}(\Gamma) \to \mathbb{Z}$ the map defined by

$$\delta^B(v) = \delta(v) - \sum_{\substack{i \text{ marking} \\ \text{at } v}} b_i.$$

We have a fiber diagram

$$\begin{aligned}
\mathfrak{Pic}_{\Gamma_{\delta^B}} & \xrightarrow{\mathcal{T}_{\delta^B}} \mathfrak{Pic}_{g,n,d} \\
\downarrow & \downarrow^{\tau_B} \\
\mathfrak{Pic}_{\Gamma_{\delta}} & \xrightarrow{j_{\Gamma_{\delta}}} \mathfrak{Pic}_{g,n,d+e}.
\end{aligned} (7.17)$$

The proof that (7.16) pulls back under τ_B as desired follows from the following observations:

• As Γ_{δ} runs through $\mathsf{G}_{g,n,d+e}$, the graphs Γ_{δ^B} run through $\mathsf{G}_{g,n,d}$. From the definitions, we verify that the conditions defining admissible weightings $w \mod r$ for Γ_{δ} and Γ_{δ^B} are identical (the shift from δ to δ^B cancels the shift from A+B to A).

• Since the underlying graphs of Γ_{δ} and Γ_{δ^B} agree, we have $h^1(\Gamma_{\delta}) = h^1(\Gamma_{\delta^B})$. Concerning the automorphisms, they appear to take into account the degree functions δ and δ^B on the graphs. But any vertex v such that $\delta(v) \neq \delta^B(v)$ must carry a marking and thus must anyway be fixed under an automorphism. Hence, $\operatorname{Aut}(\Gamma_{\delta}) = \operatorname{Aut}(\Gamma_{\delta^B})$.

• To conclude using the diagram (7.17), we observe that the map

$$\mathfrak{Pic}_{\Gamma_{sB}} \longrightarrow \mathfrak{Pic}_{\Gamma_{\delta}}$$

appearing there is a map over \mathfrak{M}_{Γ} (since only the line bundle is changed). Hence, the classes ψ_h and $\psi_{h'}$ appearing in the edge terms of (7.16) are invariant.

7.5. Proof of Invariance IV (Twisting by pullback)

Let $\mathcal{B} \to \mathcal{S}$ be a line bundle on the base. We obtain a new object of $\mathfrak{Pic}_{g,n,d}$ over \mathcal{S} by tensoring (7.1) with $\pi^* \mathcal{B}$:

$$\pi: \mathcal{C} \longrightarrow \mathcal{S}, \quad p_1, ..., p_n: \mathcal{S} \longrightarrow \mathcal{C}, \quad \mathcal{L} \otimes \pi^* \mathcal{B} \longrightarrow \mathcal{C}.$$
 (7.18)

We want to show the invariance

$$\mathsf{DR}_{g,A,\mathcal{L}\otimes\pi^*\mathcal{B}}^{\mathsf{op}} = \mathsf{DR}_{g,A,\mathcal{L}}^{\mathsf{op}}.$$

The universal twisted double ramification cycle $\mathsf{DR}_{g,A}^{\mathsf{op}}$ is the class associated with the Abel–Jacobi map

AJ:
$$\operatorname{Div}_{g,A} \longrightarrow \operatorname{\mathfrak{Pic}}_{g,n},$$

and the latter is constructed (Definition 3.5) by pulling back the morphism

$$AJ^{rel}: \mathbf{Div}_{g,A}^{rel} \longrightarrow \mathfrak{Pic}_{g,n}^{rel}.$$

292

Thus, by Lemma 2.17, the corresponding cycle $\mathsf{DR}_{g,A}^{\mathsf{op}}$ is a pullback of $\mathsf{AJ}_{\mathsf{op}}^{\mathsf{rel}}[\mathbf{Div}_{g,A}^{\mathsf{rel}}]$ from $\mathfrak{Pic}_{g,n,d}^{\mathsf{rel}}$. But twisting the family (7.1) by a line bundle pulled back from the base does not change the map to $\mathfrak{Pic}_{g,n,d}^{\mathsf{rel}}$, and so does not change the resulting operational class, proving the invariance.

We now prove the invariance

$$\varphi^*_{\mathcal{L}\otimes\pi^*\mathcal{B}}\mathsf{P}^{\bullet}_{g,A,d} = \varphi^*_{\mathcal{L}}\mathsf{P}^{\bullet}_{g,A,d}.$$

Using that $\mathsf{P}^{\bullet}_{g,A,d}$ is a pullback of $\mathsf{P}^{\bullet}_{g,\emptyset,0}$ as described in §0.7, it suffices to show that the cycle P^{\bullet}_{q} on $\mathfrak{P}^{\bullet}_{g,0,0}$ pulls back to the same expression under the two maps

$$\varphi_{\mathcal{L}_A}, \varphi_{\mathcal{L}_A \otimes \pi^* \mathcal{B}} : \mathcal{S} \longrightarrow \mathfrak{Pic}_{q,0,0}$$

induced by

$$\mathcal{L}_A = \mathcal{L}\left(-\sum_{i=1}^n a_i p_i\right) \quad \text{and} \quad \mathcal{L}_A \otimes \pi^* \mathcal{B},$$

respectively. We will use formula (4.8) for P_g^{\bullet} and show that both parts of the formula are invariant separately.

• For the term $\exp\left(-\frac{1}{2}\eta\right)$, we use Lemma 4.2 to obtain

$$\varphi_{\mathcal{L}_A\otimes\pi^*\mathcal{B}}^*\eta = \pi_*c_1(\mathcal{L}_A\otimes\pi^*\mathcal{B})^2$$

= $\pi_*(c_1(\mathcal{L}_A)^2 + 2c_1(\mathcal{L}_A)\pi^*c_1(\mathcal{B}) + \pi^*c_1(\mathcal{B})^2)$
= $\pi_*(c_1(\mathcal{L}_A)^2) + 2c_1(\mathcal{B})\underbrace{\pi_*c_1(\mathcal{L}_A)}_{=0} + c_1(\mathcal{B})^2\underbrace{\pi_*1}_{=0}$

where $\pi_* c_1(\mathcal{L}_A)$ vanishes since \mathcal{L}_A has degree zero, and $\pi_* 1$ vanishes for dimension reasons. The vertex term is therefore invariant under twisting by $\pi^* \mathcal{B}$.

• For the graph sum

$$\sum_{\substack{\Gamma_{\delta} \in \mathsf{G}_{g,0,0}\\w \in \mathsf{W}_{\Gamma_{\delta},r}}} \frac{r^{-h^{1}(\Gamma_{\delta})}}{|\operatorname{Aut}(\Gamma_{\delta})|} j_{\Gamma_{\delta}*} \left[\prod_{e=(h,h')\in \mathsf{E}(\Gamma_{\delta})} \frac{1}{\psi_{h} + \psi_{h'}} \left(1 - \exp\left(-\frac{w(h)w(h')}{2}(\psi_{h} + \psi_{h'})\right) \right) \right]$$

$$(7.19)$$

in (4.8), we show that it is a pullback under the morphism

$$\mathfrak{Pic}_{g,0,0} \longrightarrow \mathfrak{Pic}_{g,0,0}^{\mathrm{rel}},\tag{7.20}$$

which finishes the proof since the compositions of $\varphi_{\mathcal{L}_A}$ and $\varphi_{\mathcal{L}_A \otimes \pi^* \mathcal{B}}$ with the morphism (7.20) agree. For a prestable graph Γ , we have cartesian diagrams



In formula (7.19), the edge terms use only the classes ψ_h and $\psi_{h'}$, which are pullbacks from \mathfrak{M}_{Γ} . Therefore, (7.19) is the pullback under (7.20) of the identical formula with $j_{\Gamma_{\delta}}$ replaced with $j_{\Gamma_{\delta}}^{\mathsf{rel}}$.

7.6. Proof of Invariance V (Vertical twisting)

Consider the boundary divisor Δ of $\mathfrak{Pic}_{q,n,d}$ given by the partition

$$g_1+g_2=g, \quad N_1\sqcup N_2=\{1,...,n\}, \quad d_1+d_2=d_2$$

which is not symmetric. In

294

$$\mathfrak{C}_{g,n,d} \longrightarrow \mathfrak{Pic}_{g,n,d},$$

let Δ_1 and Δ_2 be, respectively, the (g_1, N_1, d_1) and (g_2, N_2, d_2) components of the universal curve over Δ . Then, we have a morphism

$$\Phi_{\Delta_1}: \mathfrak{Pic}_{g,n,d} \longrightarrow \mathfrak{Pic}_{g,n,d}$$

associated with the twisted line bundle $\mathcal{L}(\Delta_1)$ on the universal curve

$$\mathfrak{C}_{g,n,d} \longrightarrow \mathfrak{Pic}_{g,n,d}.$$

Remark 7.1. The map Φ_{Δ_1} is *not* an isomorphism. Indeed, the map is equal to the identity away from $\Delta \subset \mathfrak{Pic}_{g,n,d}$, but it sends the generic point of Δ to the generic point of the boundary divisor

$$\tilde{\Delta} = \Delta(g_1, N_1, d_1 - 1 | g_2, N_2, d_2 + 1),$$

which itself is fixed under Φ_{Δ_1} . Hence, Φ_{Δ_1} is not injective, though it is easily seen to be étale.

We want to show the invariance

$$\Phi_{\Delta_1}^* \mathsf{DR}_{g,A}^{\mathsf{op}} = \mathsf{DR}_{g,A}^{\mathsf{op}}.$$
(7.22)

Using the data

$$g_1 + g_2 = g, \quad N_1 \sqcup N_2 = \{1, ..., n\}, \quad d_1 + d_2 = d,$$

we will define a map

$$\Phi': \mathbf{Div}_{g,A} \longrightarrow \mathbf{Div}_{g,A}.$$

In fact, we will define a map $\mathbf{Div}_{g,A}^{\mathsf{rel}} \to \mathbf{Div}_{g,A}^{\mathsf{rel}}$, and then lift it to $\mathbf{Div}_{g,A}$ by fiber product with \mathfrak{Pic} . The invariance (7.22) will be deduced from Φ' .

Suppose we are given a map $S \to \mathbf{Div}_{g,A}^{\mathsf{rel}}$ compatible with C/S defined by a $\overline{M}_S^{\mathsf{gp}}$ torsor \mathcal{P} on S and a map $\alpha: C \to \mathcal{P}$. The divisor Δ determines an element of the characteristic sheaf of the log structure on $\mathfrak{Pic}_{g,n,d}$, which pulls back under the composition

$$S \longrightarrow \mathbf{Div}_{g,A}^{\mathsf{rel}} \xrightarrow{\mathsf{AJ}} \mathfrak{Pic}_{g,n,d}$$

to an element $\delta \in \overline{M}_S(S)$. All lifts of δ to $M_S(S)$ generate the same ideal sheaf on S, whose closed subscheme is exactly the pullback Δ_S of Δ . We write

$$\Delta_1', \Delta_2' \subset C$$

for the two components of the universal curve over $\Delta_S \subset S$.

We define a new map $(\mathcal{P}', \alpha'): S \to \mathbf{Div}_{g,A}^{\mathsf{rel}}$ as follows. We take the same torsor $\mathcal{P}' = \mathcal{P}$. On the locus $C_1 \hookrightarrow C$ which is the complement of Δ'_1 , we define $\alpha' = \alpha$. On the locus $C_2 \hookrightarrow C$ which is the complement of Δ'_2 , we define $\alpha' = \alpha - \delta$. Since δ vanishes on the overlap $C_1 \cap C_2$, we just have to check that the defined section extends from $C_1 \cup C_2$ to the whole of C (across the separating node $\Delta'_1 \cap \Delta'_2$). The extension can be checked étale locally, and then the claim follows from the local description of the log structure.

We have defined a map $\Phi': \mathbf{Div}_{q,A} \to \mathbf{Div}_{q,A}$, and we verify easily that the diagram

$$\begin{aligned}
\mathbf{Div}_{g,A} &\xrightarrow{\mathsf{AJ}} \mathfrak{Pic}_{g,n,d} \\
& \downarrow^{\Phi'} & \downarrow^{\Phi_{\Delta_1}} \\
\mathbf{Div}_{g,A} &\xrightarrow{\mathsf{AJ}} \mathfrak{Pic}_{g,n,d}
\end{aligned} (7.23)$$

commutes. We will prove (7.23) is a pullback square which by Lemma 2.17 yields the invariance (7.22).

To prove that (7.23) is a pullback square, since the horizontal arrows are monomorphisms, we need to show the following: given $(\mathcal{P}, \alpha) \in \mathbf{Div}_{g,A}^{\mathsf{rel}}(S)$, a line bundle \mathcal{L} on C and an isomorphism $\mathsf{AJ}(\mathcal{P}, \alpha) \xrightarrow{\sim} \Phi_{\Delta_1}(\mathcal{L})$, there exists $(\mathcal{P}_0, \alpha_0) \in \mathbf{Div}_{g,A}^{\mathsf{rel}}(S)$ such that

$$\operatorname{AJ}(\mathcal{P}_0, \alpha_0) \cong \mathcal{L}.$$

If the element (\mathcal{P}, α) has only one preimage under Φ' , we are done by commutativity. If there are two preimages (the only other case), the Abel–Jacobi images differ by a twist by Δ'_1 , and the bundle \mathcal{L} will determine which one we choose. More formally, by uniqueness, we may assume S to be strictly henselian local, then Φ' has exactly one preimage whenever $\Delta(g_1, N_1, d_1 - 1 | g_2, N_2, d_2 + 1)$ does not meet S, and the result is clear as the diagram commutes. If, on the other hand, $\Delta(g_1, N_1, d_1 - 1 | g_2, N_2, d_2 + 1)$ does meet S, then the two preimages under Φ' will have multidegrees

$$(d_1, d_2)$$
 and $(d_1 - 1, d_2 + 1)$,

and only one of these can be sent by the Abel–Jacobi map to \mathcal{L} .

We now prove the invariance

$$\Phi_{\Delta_1}^* \mathsf{P}_{g,A,d}^{\bullet} = \mathsf{P}_{g,A,d}^{\bullet}.$$

We will use Proposition 4.1 and prove that the two lines of formula (4.2) are separately invariant.

• For the exponential term, we must show that the divisor

$$-\eta + \sum_{i=1}^{n} (2a_i\xi_i + a_i^2\psi_i) + \sum_{\Gamma_{\delta} \in \mathsf{G}^{\mathsf{se}}_{g,n,d}} c_A(\Gamma_{\delta})[\Gamma_{\delta}]$$

is invariant. By Lemma 4.2, we see

$$-\pi_* c_1(\mathcal{L}_A)^2 = -\eta + \sum_{i=1}^n (2a_i \xi_i + a_i^2 \psi_i).$$

After pulling back via Φ_{Δ_1} , we obtain

$$\begin{aligned} -\pi_* c_1 (\mathcal{L}_A(\Delta_1))^2 &= -\pi_* (c_1 (\mathcal{L}_A)^2 + 2c_1 (\mathcal{L}_A) \Delta_1 + \Delta_1^2) \\ &= -\pi_* (c_1 (\mathcal{L}_A)^2) - 2 \mathrm{deg} (\mathcal{L}_A|_{\Delta_1}) \pi_* \Delta_1 - \pi_* (\Delta_1 \cdot (\pi^* \Delta - \Delta_2)) \\ &= -\pi_* (c_1 (\mathcal{L}_A)^2) - 2 \left(d_1 - \sum_{i \in N_1} a_i \right) \Delta + \Delta. \end{aligned}$$

297

We have used

$$\pi^*\Delta = \Delta_1 + \Delta_2$$

and the fact that the intersection of Δ_1 and Δ_2 has degree 1 over Δ .

For the pullback of the linear combination

$$\sum_{\Gamma_{\delta}\in\mathsf{G}^{\mathsf{se}}_{g,n,d}}c_{A}(\Gamma_{\delta})[\Gamma_{\delta}]$$

of boundary divisors, recall the divisor $\tilde{\Delta} = \Delta(g_1, N_1, d_1 - 1 | g_2, N_2, d_2 + 1)$, which satisfies

$$\tilde{\Delta} = \Phi_{\Gamma_1}(\tilde{\Delta}) = \Phi_{\Gamma_1}(\Delta).$$

We see $\Phi_{\Delta_1}^*[\Gamma_{\delta}] = [\Gamma_{\delta}]$ for all boundary divisors $[\Gamma_{\delta}]$ different from $[\Delta]$ and $[\tilde{\Delta}]$. Moreover,

$$\Phi_{\Delta_1}^*[\tilde{\Delta}] = [\tilde{\Delta}] + [\Delta] \quad \text{and} \quad \Phi_{\Delta_1}^*[\Delta] = 0.$$

Writing Γ_{Δ} and $\Gamma_{\tilde{\Delta}}$ for the graphs associated with Δ and $\tilde{\Delta}$, respectively, we see that

$$\Phi_{\Delta_1}^* \sum_{\Gamma_{\delta} \in \mathsf{G}_{g,n,d}^{\mathsf{se}}} c_A(\Gamma_{\delta})[\Gamma_{\delta}] = \sum_{\Gamma_{\delta} \in \mathsf{G}_{g,n,d}^{\mathsf{se}}} c_A(\Gamma_{\delta})[\Gamma_{\delta}] + (c_A(\Gamma_{\tilde{\Delta}}) - c_A(\Gamma_{\Delta}))[\Delta].$$

After expanding the last term further, we obtain the coefficient

$$c_A(\Gamma_{\tilde{\Delta}}) - c_A(\Gamma_{\Delta}) = -\left(d_1 - 1 - \sum_{i \in I_1} a_i\right)^2 + \left(d_1 - \sum_{i \in I_1} a_i\right)^2 = 2\left(d_1 - \sum_{i \in I_1} a_i\right) - 1,$$

which exactly balances out the error term we obtained in the pullback of

$$-\pi_*c_1(\mathcal{L}_A(\Delta_1))^2.$$

We have finished the proof of the invariance of the exponential term in (4.8).

• For the invariance of the sum over $\Gamma_{\delta} \in \mathsf{G}_{g,n,d}^{\mathsf{nse}}$, we claim that given any graph Γ_{δ} , we have the diagram

The lower and outer diagrams are cartesian as we have seen in §0.3.2, thus the upper diagram is also cartesian. While for a general graph Γ_{δ} the map $\Phi_{\Delta_1,\Gamma}$ induces a nontrivial map on the set of components of

$$\coprod_{\delta}\mathfrak{Pic}_{\Gamma_{\delta}},$$

for Γ having only non-separating edges, we obtain

$$\Phi_{\Delta_1,\Gamma}:\mathfrak{Pic}_{\Gamma_\delta}\longrightarrow\mathfrak{Pic}_{\Gamma_\delta}$$

over \mathfrak{M}_{Γ} . The classes ψ_h (for $h \in H(\Gamma)$) are pullbacks from \mathfrak{M}_{Γ} , in particular $\Phi^*_{\Delta_1,\Gamma}\psi_h = \psi_h$ for all such h. As a result, each term

$$j_{\Gamma_{\delta}*} \bigg[\prod_{e=(h,h')\in \mathcal{E}(\Gamma_{\delta})} \frac{1}{\psi_h + \psi_{h'}} \bigg(1 - \exp\bigg(-\frac{w(h)w(h')}{2}(\psi_h + \psi_{h'})\bigg) \bigg) \bigg]$$

in the sum over $\Gamma_{\delta} \in \mathsf{G}_{q,n,d}^{\mathsf{nse}}$ and $w \in \mathsf{W}_{\Gamma_{\delta},r}$ in formula (4.2) is indeed invariant.

7.7. Proof of Invariance VI (Partial stabilization)

7.7.1. The stack $\mathfrak N$

We begin by introducing a stack \mathfrak{N} that allows us to reformulate Invariance VI as an equality of operational classes on \mathfrak{N} . The stack \mathfrak{N} parameterizes data

$$f: C' \longrightarrow C, \quad \mathcal{L}/C,$$

where f is a map of prestable genus-g curves which is a partial stabilization (a surjection which contracts some unstable rational components of C') and \mathcal{L} is a line bundle on Cof degree zero. By a small extension of the arguments of [50], \mathfrak{N} is an algebraic stack and comes with two maps to $\mathfrak{Pic}_{q,0,0}$ given by

$$\begin{split} \ell \colon \mathfrak{N} & \longrightarrow \mathfrak{Pic}_{g,0,0}, \\ (f \colon C' \to C, \mathcal{L}) \longmapsto (C, \mathcal{L}), \end{split}$$

and

$$\begin{split} \ell' \colon \mathfrak{N} &\longrightarrow \mathfrak{Pic}_{g,0,0}, \\ (f \colon C' \to C, \mathcal{L}) \longmapsto (C', f^* \mathcal{L}). \end{split}$$

LEMMA 7.2. We have

$$\ell^* \mathsf{DR}_{g,0}^{\mathsf{op}} = (\ell')^* \mathsf{DR}_{g,0}^{\mathsf{op}}, \quad \ell^* \mathsf{P}_g^{\bullet} = (\ell')^* \mathsf{P}_g^{\bullet}.$$
 (7.25)

This lemma will be proven in the next subsection, but first we show that Invariance VI is implied by the lemma.

For Invariance VI, we are given the data

$$C' \xrightarrow{f} C \longrightarrow S, \quad \mathcal{L} \longrightarrow C, \quad p_1, ..., p_n : \mathcal{S} \longrightarrow C, \quad p'_1, ..., p'_n : \mathcal{S} \longrightarrow C',$$

with f a partial stabilization satisfying $f \circ p'_i = p_i$ and a vector $A \in \mathbb{Z}^n$ satisfying $a_i = 0$ if p'_i meets the exceptional locus of f. Invariance VI then says that for the maps

$$\varphi_{\mathcal{L}}, \varphi_{f^*\mathcal{L}} \colon \mathcal{S} \longrightarrow \mathfrak{Pic}_{g,n,d}$$

induced by

$$C \longrightarrow \mathcal{S}, \quad p_1, ..., p_n {:} \, \mathcal{S} \longrightarrow C, \quad \mathcal{L} \longrightarrow C,$$

and

$$C' \longrightarrow \mathcal{S}, \quad p'_1, ..., p'_n : \mathcal{S} \longrightarrow C', \quad f^* \mathcal{L} \longrightarrow C,$$

respectively, we have

$$\varphi_{\mathcal{L}}^{*}\mathsf{DR}_{g,A}^{\mathsf{op}} = \varphi_{f^{*}\mathcal{L}}^{*}\mathsf{DR}_{g,A}^{\mathsf{op}} \quad \text{and} \quad \varphi_{\mathcal{L}}^{*}\mathsf{P}_{g,A,d}^{\bullet} = \varphi_{f^{*}\mathcal{L}}^{*}\mathsf{P}_{g,A,d}^{\bullet}.$$

The condition $a_i = 0$ if p'_i meets the exceptional locus of f implies the equality

$$f^*\left(\mathcal{L}\left(-\sum_{i=1}^n a_i p_i\right)\right) = (f^*\mathcal{L})\left(-\sum_{i=1}^n a_i p_i'\right)$$

of line bundles on C'. Denote by $f: \mathcal{S} \to \mathfrak{N}$ the map associated with the data

$$C' \xrightarrow{f} C \longrightarrow S, \quad \mathcal{L}\left(-\sum_{i=1}^n a_i p_i\right) \longrightarrow C.$$

Writing

$$\mathcal{L}_A = \mathcal{L}\left(-\sum_{i=1}^n a_i p_i\right) \quad \text{and} \quad f^* \mathcal{L}_A = (f^* \mathcal{L})\left(-\sum_{i=1}^n a_i p_i'\right)$$

we obtain a commutative diagram



299

From the arguments presented in $\S0.7$, it follows that

$$\varphi_{\mathcal{L}}^*\mathsf{DR}_{g,A}^{\mathsf{op}} = \varphi_{\mathcal{L}_A}^*\mathsf{DR}_{g,\varnothing}^{\mathsf{op}} \quad \text{and} \quad \varphi_{f^*\mathcal{L}}^*\mathsf{DR}_{g,A}^{\mathsf{op}} = \varphi_{f^*\mathcal{L}_A}^*\mathsf{DR}_{g,\varnothing}^{\mathsf{op}},$$

with parallel equations for $\mathsf{P}^{\bullet}_{q,A,d}$ and $\mathsf{P}^{\bullet}_{q,\varnothing,0}$. Assuming (7.25), we have

$$\varphi_{\mathcal{L}}^* \mathsf{DR}_{g,A}^{\mathsf{op}} = \varphi_{\mathcal{L}_A}^* \mathsf{DR}_{g,\varnothing}^{\mathsf{op}} = g^* \ell^* \mathsf{DR}_{g,\varnothing}^{\mathsf{op}} = g^* \ell'^* \mathsf{DR}_{g,\varnothing}^{\mathsf{op}} = \varphi_{f^*\mathcal{L}_A}^* \mathsf{DR}_{g,\varnothing}^{\mathsf{op}} = \varphi_{f^*\mathcal{L}}^* \mathsf{DR}_{g,A}^{\mathsf{op}},$$

and similarly for $\mathsf{P}^{\bullet}_{q,A,d}$. Thus (7.25) implies Invariance VI.

7.7.2. Invariance for $\mathfrak N$

300

Here we prove Lemma 7.2; it follows immediately from (7.25) in Lemmas 7.4 and 7.6 below. We start with a preliminary result.

LEMMA 7.3. The map

$$\ell \colon \mathfrak{N} \longrightarrow \mathfrak{Pic}_{g,0,0},$$
$$(f \colon C' \to C, \mathcal{L}) \longmapsto (C, \mathcal{L}),$$

is syntomic, $(^{39})$ and the map

$$\begin{split} \ell' \colon \mathfrak{N} & \longrightarrow \mathfrak{Pic}_{g,0,0}, \\ (f \colon C' \to C, \mathcal{L}) \longmapsto (C', f^* \mathcal{L}). \end{split}$$

is smooth.

Proof. The stack of partial stabilizations $(f: C' \to C)$ has smooth charts given by $\overline{\mathcal{M}}_{g,n+m}$, where $(C, p_1, ..., p_n, q_1, ..., q_m)$ maps to the contraction map from C to the stabilization of $(C, p_1, ..., p_n)$ by forgetting the q markings. Charts for \mathfrak{N} are then given by

$$\overline{\mathcal{M}}_{g,n+m} \times_{\overline{\mathcal{M}}_{g,n}} \mathfrak{Pic}_{\overline{\mathcal{M}}_{g,n+1}/\overline{\mathcal{M}}_{g,n}} \xrightarrow{G} \mathfrak{N}.$$

Charts for the map ℓ are given by the composition of the top horizontal arrows in the commutative diagram



 $^(^{39})$ Flat and lci.

Both squares here are pullbacks, the bottom right horizontal map is smooth, and the bottom left horizontal map is syntomic. Hence ℓ is syntomic, using that syntomicity is a flat-local property on the target.

Charts for the map ℓ' are given by commutative diagrams

The right vertical arrow is a base change of the smooth map $\overline{\mathcal{M}}_{g,n+m} \to \mathfrak{M}_g$, so once we have shown F to be smooth, we can conclude using that smoothness is a flat-local property on the target.

In fact, F is an open immersion: F is isomorphic to the inclusion of the locus

$$U \hookrightarrow \mathfrak{Pic}_{\overline{\mathcal{M}}_{g,n+m+1}/\overline{\mathcal{M}}_{g,n+m}}$$

of line bundles which are trivial on the contracted rational components. We must verify the induced map

$$F': \overline{\mathcal{M}}_{g,n+m} \times_{\overline{\mathcal{M}}_{g,n}} \mathfrak{Pic}_{\overline{\mathcal{M}}_{g,n+1}/\overline{\mathcal{M}}_{g,n}} \longrightarrow U$$

is an isomorphism. The source and target are smooth over $\overline{\mathcal{M}}_{g,n+m}$. On each geometric fiber over $\overline{\mathcal{M}}_{g,n+m}$, the map F' is an isomorphism via the explicit description of the Jacobian of a prestable curve. But then F' is flat (by the fiberwise criterion), is unramified (by a pointwise check), and is universally injective (again by a pointwise check), and hence is an isomorphism.

LEMMA 7.4. We have

$$\ell^*\mathsf{DR}^{\mathsf{op}}_{g,\mathbf{0}} = (\ell')^*\mathsf{DR}^{\mathsf{op}}_{g,\mathbf{0}} \in \mathsf{CH}^g_{\mathsf{op}}(\mathfrak{N}).$$

Proof. The maps ℓ and ℓ' are syntomic by Lemma 7.3. It therefore suffices by Lemma 2.17 to construct an isomorphism

$$\varphi: \ell^* \mathbf{Div}_{g,\mathbf{0}} \xrightarrow{\sim} (\ell')^* \mathbf{Div}_{g,\mathbf{0}}$$

of stacks over \mathfrak{N} and even to construct the isomorphism on the level of $\mathbf{Div}^{\mathsf{rel}}$.

An object of $\ell^* \operatorname{Div}_{g,0}^{\operatorname{rel}}$ consists of a stabilization map $f: C' \to C$, a line bundle \mathcal{L} on C, a $\mathbb{G}_m^{\operatorname{trop}}$ torsor \mathcal{P} on S, and a map $\alpha: C \to \mathcal{P}$ such that $\mathcal{O}(\alpha)$ and \mathcal{L} are isomorphic up to pullback from the base. An object of $(\ell')^* \operatorname{\mathbf{Div}}_{g,\mathbf{0}}^{\operatorname{rel}}$ consists of almost the same data, but α is replaced by any $\alpha': C' \to \mathcal{P}$, with $\mathcal{O}(\alpha')$ and $f^*\mathcal{L}$ isomorphic up to pullback. We then define the map φ simply by composing, setting $\alpha' = f \circ \alpha$.

We must show that φ is an isomorphism. Suppose we are given the data $f: C' \to C$, \mathcal{L} on $C, \mathcal{P}, \alpha': C' \to \mathcal{P}$, with $\mathcal{O}(\alpha') \cong f^*\mathcal{L}$ up to pullback. Since the degree of $f^*\mathcal{L}$ vanishes on components contracted by f, we see that the same is true of the degree of $\mathcal{O}(\alpha')$ —the slopes of the restriction of α' to the contracted graph of the curve are linear on edges. In other words, the restriction is still piecewise linear with integer slopes, hence we can set α to be the restriction.

More work will be required to prove the second equality

$$\ell^*\mathsf{P}_g^{\scriptscriptstyle\bullet} = (\ell')^*\mathsf{P}_g^{\scriptscriptstyle\bullet}.$$

We start by considering the morphism ℓ . For a prestable graph Γ_{δ} of degree zero, consider the diagram

where the left and the right squares are pullbacks and the middle square is commutative.

• The stacks $\mathfrak{N}_{g(v),n(v),\delta(v)}$ are the natural generalizations of \mathfrak{N} to the case of marked curves and line bundles of arbitrary total degrees.

• The stack $\mathfrak{N}_{\Gamma_{\delta}}$ parameterizes data

$$(C_v)_{v\in\mathcal{V}(\Gamma)}, \quad \mathcal{L}/C, \quad f: C' \longrightarrow C,$$

where f is a partial stabilization and \mathcal{L} is a multidegree δ line bundle on

$$C = \bigsqcup_{v \in \mathcal{V}(\Gamma)} C_v.$$

• The stack $\mathfrak{N}'_{\Gamma_{\delta}}$ parameterizes data

$$(C_v)_{v\in\mathcal{V}(\Gamma)}, \quad \mathcal{L}/C, \quad (f_v:C'_v\to C_v)_{v\in\mathcal{V}(\Gamma)}.$$

• The gluing map $G: \mathfrak{N}'_{\Gamma_{\delta}} \to \mathfrak{N}_{\Gamma_{\delta}}$ sending $(f_v: C'_v \to C_v)_{v \in \mathcal{V}(\Gamma)}$ to

$$f \colon \bigsqcup_{v} C'_{v} = C' \longrightarrow C = \bigsqcup_{v} C_{v}$$

is proper, representable and birational.

Properness of G can be checked using the valuative criterion. The difference between $\mathfrak{N}_{\Gamma_{\delta}}$ and $\mathfrak{N}'_{\Gamma_{\delta}}$ is that, in the first space, we have sections of the non-smooth locus of \mathfrak{C} for each half-edge (telling us where to cut apart the curve), whereas, for the second, the sections go to the non-smooth locus of \mathfrak{C}' . Fibres of G correspond to choices of lifts of these sections along

$$f: \mathfrak{C}' \longrightarrow \mathfrak{C}.$$

Existence and uniqueness of such lifts follows from properness of f and of the inclusion of the non-smooth locus. Representability of G is a short argument showing G is injective on stabilizer groups. Birationality follows since G is an isomorphism over the dense open locus where f is an isomorphism. Let

$$\hat{J}_{\Gamma_{\delta}}:\mathfrak{N}'_{\Gamma_{\delta}}\longrightarrow\mathfrak{N}$$

be the composition of G and $J_{\Gamma_{\delta}}$.

In the following lemma, we compare pullback formulas under ℓ and ℓ' for the stacks $\mathfrak{N}_{g,n,d}$ above. Denote by $\psi_i = \ell^* \psi_i$ and $\psi'_i = (\ell')^* \psi_i \in \mathsf{CH}^1_{\mathsf{op}}(\mathfrak{N}_{g,n,d})$ the pullbacks of ψ classes under ℓ and ℓ' , respectively.

LEMMA 7.5. (i) $\ell^* \eta = (\ell')^* \eta$.

(ii) $\psi_i = \psi'_i - D_i$, where D_i is the class in $CH^1_{op}(\mathfrak{N}_{g,n,d})$ associated with the boundary divisor of $\mathfrak{N}_{g,n,d}$ generically parameterizing a partial stabilization



Proof. (i) There are two pairs of universal curves with line bundle $(\mathfrak{C}, \mathfrak{L})$ and $(\mathfrak{C}', \mathfrak{L}')$ over the stack $\mathfrak{N}_{q,n,d}$:



with sections

$$\sigma_i:\mathfrak{N}_{g,n,d}\longrightarrow\mathfrak{C}$$
 and $\sigma'_i:\mathfrak{N}_{g,n,d}\rightarrow\mathfrak{C}'$.

Because $f_*[\mathfrak{C}'] = [\mathfrak{C}]$, we have

$$\ell^* \eta = \pi_*(c_1(\mathfrak{L})^2) = \pi_*(c_1(\mathfrak{L})^2 f_*[\mathfrak{C}']) = \pi'_*(c_1(f^*\mathfrak{L})^2) = (\ell')^* \eta$$

(ii) Let D'_0 be the divisor

304



in \mathfrak{C}' , and let D'_i be the divisor



in \mathfrak{C}' . Here, the arrows pointing to the vertices with genus and degree zero indicate which component of the universal curve over the corresponding boundary divisor in $\mathfrak{N}_{g,n,d}$ we take. The divisors D'_0, D'_1, \dots, D'_n are precisely the divisorial loci in \mathfrak{C}' which are contracted by the map $f: \mathfrak{C}' \to \mathfrak{C}$. Then,

$$\ell^*\psi_i = c_1(\sigma_i^*\omega_\pi) = (\sigma_i')^*c_1(f^*\omega_\pi) = (\sigma_i')^*c_1\left(\omega_{\pi'}\left(-D_0' - \sum_{i=1}^n D_i'\right)\right) = (\ell')^*\psi_i - D_i,$$

where the sections are denoted by σ_i .

For the morphism ℓ' , form the fiber diagram

By definition, the fiber product $\mathfrak{Pic}'_{\Gamma_{\delta}}$ parameterizes data

$$(C'_v)_{v\in\mathcal{V}(\Gamma)}, \quad \mathcal{L}'/C' = \bigsqcup_v C'_v, \quad f: C' \longrightarrow C, \quad \mathcal{L}/C, \quad f^*\mathcal{L} \xrightarrow{\sim} \mathcal{L}',$$

which simplifies to

$$(C'_v)_{v\in\mathcal{V}(\Gamma)}, \quad f:\bigsqcup_v C'_v = C' \longrightarrow C, \quad \mathcal{L}/C.$$

On the other hand, the stack $\mathfrak{N}'_{\Gamma_{\delta}}$ parameterizes

$$(C_v)_{v \in \mathcal{V}(\Gamma)}, \quad \mathcal{L}/C, \quad (f_v: C'_v \to C_v)_{v \in \mathcal{V}(\Gamma)}.$$

There is a subtle difference here. For $\mathfrak{Pic}'_{\Gamma_{\delta}}$, the map f is allowed to contract entire components C'_{v} to points, whereas in the second case the target C_{v} is always 1-dimensional.

Our next step is to show that

$$\mathfrak{Pic}'_{\Gamma_{\delta}} \cong \bigsqcup_{\Gamma_{\delta} \to \widetilde{\Gamma}_{\delta}} \mathfrak{N}_{\Gamma_{\delta} \to \widetilde{\Gamma}_{\delta}}.$$
(7.31)

More precisely, the connected components of $\mathfrak{Pic}'_{\Gamma_{\delta}}$ are in bijective correspondence to partial stabilizations

$$\Gamma_{\delta} \rightarrow \Gamma_{\delta}.$$

We will prove, given a vertex $v \in V(\Gamma)$ which can be contracted (with $g(v)=0, n(v) \leq 2$, and $\delta(v)=0$), that the locus of points in $\operatorname{\mathfrak{Pic}}_{\Gamma_{\delta}}$ where $f: C' \to C$ contracts C'_v is open and closed.

For a vertex v with n(v)=2, the universal curve $\mathfrak{C}'_v \to \mathfrak{Pic}'_{\Gamma_{\delta}}$ has two sections (corresponding to the half-edges at v), and the locus where C'_v is contracted equals the locus where the sections coincide, which is closed since $\mathfrak{C}'_v \to \mathfrak{Pic}'_{\Gamma_{\delta}}$ is separated.

To show closedness of the locus where C_v^\prime is not contracted, assume that we have a family

$$(C'_{v,S})_{v\in\mathcal{V}(\Gamma)}, \quad f:\bigsqcup_{v}C'_{v,S}=C'_{S}\longrightarrow C_{S}, \quad \mathcal{L}/C_{S},$$

of $\mathfrak{Pic}_{\Gamma_{\delta}}^{\prime}$ over the spectrum S of a strictly henselian DVR such that the fiber $C'_{v,\eta}$ of $C'_{v,S}$ over the generic point η of S is not contracted by f. We want to show that then also the fiber $C'_{v,L}$ over the closed point L of S is not contracted. By assumption, $C'_{v,\eta}$ maps to a union $C_{v,\eta}$ of components of the fiber C_{η} of C_S over η . Then $C_{v,\eta}$ specializes to a union $C_{v,L}$ of components of C_L . Since f is proper, f maps the closure $C'_{v,S}$ of $C'_{v,\eta}$ to the closure of its image $C_{v,\eta}$. Since $C_{v,L}$ is still positive-dimensional, the curve $C'_{v,L}$ is indeed not contracted. For related arguments, see the proof of [50, Proposition 2.2].

For a vertex v with n(v)=1, the universal curve $\mathfrak{C}'_v \to \mathfrak{Pic}'_{\Gamma_{\delta}}$ has a single section σ_h . On the one hand, the locus where C'_v is contracted is exactly the locus where $f:\mathfrak{C}'\to\mathfrak{C}$ maps σ_h to the smooth locus of $\mathfrak{C}\to\mathfrak{Pic}'_{\Gamma_{\delta}}$, thus it is open. On the other hand, it is also the preimage under σ_h of the exceptional locus of $\mathfrak{C}'\to\mathfrak{C}$, and thus closed.

We have proven that the connected components of $\mathfrak{Pic}'_{\Gamma_{\delta}}$ are in bijective correspondence to partial stabilizations $\Gamma_{\delta} \to \widetilde{\Gamma}_{\delta}$. But a point

$$(C'_v)_{v\in\mathcal{V}(\Gamma)}, \quad f:\bigsqcup_v C'_v = C' \to C, \quad \mathcal{L}/C,$$

on the corresponding component is equivalent to the data of any collection of curves C'_v for $v' \in \mathcal{V}(\Gamma) \setminus \mathcal{V}(\widetilde{\Gamma})$, which are contracted by f, together with a point

$$(C'_v)_{v \in \mathcal{V}(\widetilde{\Gamma})}, \quad f: (C'_v \to C_v), \quad \mathcal{L}/C = \bigsqcup_{v \in V(\widetilde{\Gamma})} C_v,$$

305

of $\mathfrak{N}'_{\widetilde{\Gamma}_s}$. Hence, we have a isomorphism

$$\mathfrak{N}_{\Gamma_\delta \to \widetilde{\Gamma}_\delta} \cong \mathfrak{N}'_{\widetilde{\Gamma}_\delta} \times \prod_{v \in \mathcal{V}(\Gamma) \backslash \mathcal{V}(\widetilde{\Gamma})} \mathfrak{M}_{0,\mathbf{n}(v)},$$

where, in the last expression, n(v) is necessarily 1 or 2.

For each partial stabilization $\Gamma_{\delta} \rightarrow \widetilde{\Gamma}_{\delta}$, we denote by

$$J_{\Gamma_{\delta} \to \widetilde{\Gamma}_{\delta}} : \mathfrak{N}_{\Gamma_{\delta} \to \widetilde{\Gamma}_{\delta}} \longrightarrow \mathfrak{N}$$

the restriction of $J'_{\Gamma_{\delta}}$ to $\mathfrak{N}_{\Gamma_{\delta} \to \widetilde{\Gamma}_{\delta}}$.

LEMMA 7.6. We have

$$\ell^*\mathsf{P}_g^{\scriptscriptstyle\bullet} = (\ell')^*\mathsf{P}_g^{\scriptscriptstyle\bullet} \in \prod_{c=0}^\infty \mathsf{CH}^c_{\mathsf{op}}(\mathfrak{N}) \quad for \ all \ c \geqslant 0.$$

Proof. We will use formula (4.8) for P_g^{\bullet} . By Lemma 7.5, the terms $\exp\left(-\frac{1}{2}\eta\right)$ have identical pullback under ℓ and ℓ' . We can therefore focus on the sum over graphs and weighting mod r.

We start with a few remarks about the combinatorial factors in P_g^{\bullet} which will arise in the proof. Let $\Gamma_{\delta} \to \widetilde{\Gamma}_{\delta}$ be a partial stabilization, then the Betti numbers agree:

$$h^1(\Gamma_\delta) = h^1(\widetilde{\Gamma}_\delta).$$

Given r, the map $W_{\Gamma_{\delta},r} \to W_{\widetilde{\Gamma}_{\delta},r}$ of admissible weightings mod r (induced by the inclusion $H(\widetilde{\Gamma}) \to H(\Gamma)$ of half-edge sets) is a bijection.

Moreover, if the map $\Gamma_{\delta} \to \widetilde{\Gamma}_{\delta}$ only contracts components with

$$(g(v), n(v), \delta(v)) = (0, 2, 0),$$

there is a canonical isomorphism $\operatorname{Aut}(\Gamma_{\delta}) \cong \operatorname{Aut}(\widetilde{\Gamma}_{\delta})$. On the other hand, in the formula for P_{q}^{\bullet} , every term such that Γ_{δ} has a vertex with

$$(g(v), n(v), \delta(v)) = (0, 1, 0)$$

necessarily vanishes. Indeed, the half-edge h at this vertex must have w(h)=0 such that the corresponding edge term vanishes.

To keep the notation concise, we write Φ_a for the power-series

$$\Phi_a(x) = \frac{1}{x} \left(1 - \exp\left(-\frac{a}{2}x\right) \right) = \sum_{m=0}^{\infty} (-1)^m \left(\frac{a}{2}\right)^{m+1} \frac{1}{(m+1)!} x^m = \frac{a}{2} - \frac{a^2}{8}x + \dots$$

307

appearing in the edge terms of P_{g}^{\bullet} . Moreover, given a graph $\widetilde{\Gamma}_{\delta}$ with a half-edge h incident to a vertex v, denote by ψ_{h} and ψ'_{h} the classes on $\mathfrak{N}'_{\widetilde{\Gamma}_{\delta}}$ pulled back from $\mathfrak{N}_{g(v),n(v),\delta(v)}$ in the diagram (7.29). Similarly, given a partial stabilization $\Gamma_{\delta} \to \widetilde{\Gamma}_{\delta}$, the space $\mathfrak{N}_{\Gamma_{\delta} \to \widetilde{\Gamma}_{\delta}}$ contains $\mathfrak{N}'_{\widetilde{\Gamma}_{\delta}}$ as a factor, and hence the notation also makes sense on $\mathfrak{N}_{\Gamma_{\delta} \to \widetilde{\Gamma}_{\delta}}$ (provided h is a half-edge of $\widetilde{\Gamma}_{\delta}$).

Let us first compute the pullback of the graph sum in $\mathsf{P}_{g}^{\bullet,r}$ via $\ell': \mathfrak{N} \to \mathfrak{Pic}_{g,0,0}$. Using the diagram (7.30) and the decomposition (7.31), we see that

$$(\ell')^{*} \sum_{\Gamma_{\delta},w} \frac{r^{-h^{1}(\Gamma_{\delta})}}{|\operatorname{Aut}(\Gamma_{\delta})|} j_{\Gamma_{\delta}*} \bigg[\prod_{e=(h,h')\in \operatorname{E}(\Gamma_{\delta})} \Phi_{w(h)w(h')}(\psi_{h}+\psi_{h'}) \bigg] \\ = \sum_{\Gamma_{\delta}\to\tilde{\Gamma}_{\delta},w} \frac{r^{-h^{1}(\Gamma_{\delta})}}{|\operatorname{Aut}(\Gamma_{\delta})|} J_{\Gamma_{\delta}\to\tilde{\Gamma}_{\delta}*} \bigg[\prod_{e=(h,h')\in \operatorname{E}(\Gamma_{\delta})} \Phi_{w(h)w(h')}(\psi_{h}'+\psi_{h'}') \bigg].$$
(7.32)

In the second line, the sum is over all partial stabilizations $\Gamma_{\delta} \rightarrow \widetilde{\Gamma}_{\delta}$.

Second, we compute the pullback of the graph sum in $\mathsf{P}_{g}^{\bullet,r}$ via $\ell:\mathfrak{N}\to\mathfrak{Pic}_{g,0,0}$:

$$\ell^{*} \sum_{\widetilde{\Gamma}_{\delta},w} \frac{r^{-h^{1}(\Gamma_{\delta})}}{|\operatorname{Aut}(\widetilde{\Gamma}_{\delta})|} j_{\widetilde{\Gamma}_{\delta}*} \left[\prod_{e=(h,h')\in \operatorname{E}(\widetilde{\Gamma}_{\delta})} \Phi_{w(h)w(h')}(\psi_{h}+\psi_{h'}) \right] \\ = \sum_{\widetilde{\Gamma}_{\delta},w} \frac{r^{-h^{1}(\widetilde{\Gamma}_{\delta})}}{|\operatorname{Aut}(\widetilde{\Gamma}_{\delta})|} \hat{J}_{\widetilde{\Gamma}_{\delta}*} \left[\prod_{e=(h,h')\in \operatorname{E}(\widetilde{\Gamma}_{\delta})} \Phi_{w(h)w(h')}(\psi_{h}+\psi_{h'}) \right].$$
(7.33)

We use here the right fiber diagram in (7.29) together with the fact that G is proper, representable, and birational. So, by Proposition 2.16, the pushforward of fundamental classes under $J_{\tilde{\Gamma}_{\delta}}$ and $\hat{J}_{\tilde{\Gamma}_{\delta}}$ agree.

To compare to the formula for the pullback under ℓ' , we use

$$\psi_h + \psi_{h'} = \psi'_h - D_h + \psi'_{h'} - D_{h'}$$

on $\mathfrak{N}'_{\Gamma_{\delta}}$, by Lemma 7.5. The next step of the proof is to use the self-intersection formula for D_h and $D_{h'}$ (similar to the formula described in [30, Appendix A]) to expand the edge term

$$\Phi_{w\,w'}(\psi_h' - D_h + \psi_{h'}' - D_{h'}) = \sum_{m=0}^{\infty} (-1)^m \left(\frac{w\,w'}{2}\right)^{m+1} \frac{1}{(m+1)!} (\psi_h' - D_h + \psi_{h'}' - D_{h'})^m.$$

For example, $(D_h)^2$ is equal to



and similarly for $(D_{h'})^2$.

The result will be a linear combination of terms

where the edge (h, h') is at position ℓ in the above chain $(1 \leq \ell \leq L)$. The total degree of this term (before the pushforward by $\hat{J}_{\tilde{\Gamma}_{\delta}}$) is

$$m=\sum_{j\neq\ell}(e_j+1)+a+b.$$

The total coefficient of this particular term in

$$\Phi_{w\,w'}(\psi'_{h} - D_{h} + \psi'_{h'} - D_{h'})$$

is then

$$\underbrace{(-1)^m \left(\frac{w \, w'}{2}\right)^{m+1} \frac{1}{(m+1)!}}_{\text{coefficient in } \Phi_{w \, w'}} \cdot \begin{pmatrix} m \\ e_1 + 1, \dots, a, b, \dots, e_L + 1 \end{pmatrix}}_{\text{coefficient in } \Phi_{w \, w'}} \underbrace{(-1)^{L-1}}_{\text{or excess intersection of } -D_h \text{ and } -D_{h'}}^{(-1)}$$

where the multinomial coefficient comes from the expansion of

$$(\psi'_h - D_h + \psi'_{h'} - D_{h'})^m.$$

Writing $e_{\ell} = a + b$, we can simplify to obtain

$$\frac{1}{m+1} \left(\prod_{j=1}^{L} (-1)^{e_j} \left(\frac{w \, w'}{2} \right)^{e_j+1} \frac{1}{(e_j+1)!} \right) \binom{e_\ell}{a} (e_\ell+1).$$

Summing over all choices $a+b=e_{\ell}$ for fixed e_{ℓ} , the coefficient of the term

$$: \underbrace{(\psi_{h_1} + \psi_{h'_1})^{e_1}}_{(g_v, \delta(v))} \underbrace{(\psi_{h_1} + \psi_{h'_1})^{e_\ell}}_{(0,0)} \underbrace{(\psi_{h_1} + \psi_{h'_1})^{e_\ell}}_{(0,0)} : (7.35)$$

is exactly equal to

$$\frac{e_{\ell}+1}{m+1} \left(\prod_{j=1}^{L} (-1)^{e_j} \left(\frac{w \, w'}{2} \right)^{e_j+1} \frac{1}{(e_j+1)!} \right).$$

Pushing forward by $\hat{J}_{\tilde{\Gamma}_{\delta}}$, we forget where in the chain above the edge (h, h') has been. Summing over the *L* possible positions and using

$$m + 1 = \sum_{\ell=1}^{L} (e_{\ell} + 1),$$

we obtain the coefficient

$$\prod_{j=1}^{L} \underbrace{(-1)^{e_j} \left(\frac{w \, w'}{2}\right)^{e_j+1}}_{\text{coefficient of } x^{e_j} \text{ in } \Phi_{w \, w'}(x)}.$$

From the above discussion, we see that (7.33) equals

$$\sum_{\Gamma_{\delta} \to \widetilde{\Gamma}_{\delta}, w} \frac{r^{-h^{1}(\Gamma_{\delta})}}{|\operatorname{Aut}(\widetilde{\Gamma}_{\delta})|} J_{\Gamma_{\delta} \to \widetilde{\Gamma}_{\delta}*} \bigg[\prod_{e=(h,h')\in \operatorname{E}(\Gamma_{\delta})} \Phi_{w(h)w'(h)}(\psi'_{h} + \psi'_{h'}) \bigg].$$

The sum goes over stabilizations $\Gamma_{\delta} \rightarrow \widetilde{\Gamma}_{\delta}$ contracting chains of curves with

$$(g, n, d) = (0, 2, 0).$$

By the previous remarks concerning the combinatorial factors, we have

$$h^1(\widetilde{\Gamma}_{\delta}) = h^1(\Gamma_{\delta}) \text{ and } |\operatorname{Aut}(\widetilde{\Gamma}_{\delta})| = |\operatorname{Aut}(\Gamma_{\delta})|.$$

The sum does not change if we allow arbitrary stabilizations $\Gamma_{\delta} \to \widetilde{\Gamma}_{\delta}$, since for Γ_{δ} having a vertex with (g, n, d) = (0, 1, 0), the summand automatically vanishes. Thus the sum above equals the term computed in (7.32).

8. Applications

8.1. Proofs of Theorem 0.9 and Conjecture A

We start by recalling notions presented in §0.5, but now in the more general setting of k-differentials. Let $A=(a_1,...,a_n)$ be a vector of zero and pole multiplicities satisfying

$$\sum_{i=1}^{n} a_i = k(2g-2).$$

Let $\mathcal{H}_{g}^{k}(A) \subset \mathcal{M}_{g,n}$ be the closed (generally non-proper) locus of pointed curves

$$(C, p_1, \dots, p_n)$$

satisfying the condition

310

$$\mathcal{O}_C\bigg(\sum_{i=1}^n a_i p_i\bigg) \simeq \omega_C^{\otimes k}$$

In other words, $\mathcal{H}_{g}^{k}(A)$ is the locus of (possibly) meromorphic k-differentials with zero and pole multiplicities prescribed by A. In paper [28], a compact moduli space of twisted k-canonical divisors

$$\widetilde{\mathcal{H}}_{g}^{k}(A) \subset \overline{\mathcal{M}}_{g,n}$$

is constructed extending $\mathcal{H}_{g}^{k}(A) = \widetilde{\mathcal{H}}_{g}^{k}(A) \cap \mathcal{M}_{g,n}$ to the boundary of $\overline{\mathcal{M}}_{g,n}$.

For $k \ge 1$ and A not of the form $A = k \cdot A'$ with a vector A' of non-negative integers, the locus $\widetilde{\mathcal{H}}_g^k(A)$ is of pure codimension g in $\overline{\mathcal{M}}_{g,n}$ by [28, Theorem 3] (for k=1) and [72, Theorem 1.1] (for k>1). A weighted fundamental cycle of $\widetilde{\mathcal{H}}_g^k(A)$,

$$\mathsf{H}_{g,A}^k \in \mathsf{CH}_{2g-3+n}(\overline{\mathcal{M}}_{g,n}),\tag{8.1}$$

is constructed in [28, Appendix A] and [72, §3.1] with explicit non-trivial weights on the irreducible components. The closure

$$\overline{\mathcal{H}}_{g,A}^k \subset \overline{\mathcal{M}}_{g,n}$$

contributes to the weighted fundamental class $\mathsf{H}^k_{g,A}$ with multiplicity 1, but there are additional boundary contributions, as described in the references above.

The weighted fundamental class $\mathsf{H}_{g,A}^k$ was conjectured in [28], [72] to equal the class given by Pixton's formula for the double ramification cycle. To state the conjecture, consider the shifted $(^{40})$ vector

$$\tilde{A} = (a_1 + k, \dots, a_n + k).$$

Conjecture A. For $k \ge 1$ and A not of the form $A = k \cdot A'$ with a vector A' of non-negative integers, we have an equality

$$\mathsf{H}_{q,A}^k = 2^{-g} P_q^{g,k}(\tilde{A}),$$

where $P_q^{g,k}(\tilde{A})$ is Pixton's cycle class defined in [42, §1.1].

By combining Theorem 0.7 with previous results of [40], we can now prove the conjecture.

 $[\]binom{40}{C}$ The shift is needed since Pixton's original formula worked with powers of the log-canonical line bundle $\omega_C^{\log} = \omega_C(\sum_{i=1}^n p_i)$ instead of ω_C .

311

THEOREM 8.1. Conjecture A is true.

Proof. By [40, Theorem 1.1], the weighted fundamental class $\mathsf{H}_{g,A}^k$ is equal to the double ramification cycle $\mathsf{DR}_{g,A,\omega^k}$ constructed in [37]. By Theorem 0.1, $\mathsf{DR}_{g,A,\omega^k}$ is in turn given by the action of $\mathsf{DR}_{g,A}^{\mathsf{op}}$ on the fundamental class of $\overline{\mathcal{M}}_{g,n}$ via the morphism $\varphi_{\omega_k^k}: \overline{\mathcal{M}}_{g,n} \to \mathfrak{Pic}_{g,n,k(2g-2)}$ associated with the family

$$\pi: \mathcal{C}_{g,n} \longrightarrow \overline{\mathcal{M}}_{g,n}, \quad \omega_{\pi}^k \longrightarrow \mathcal{C}_{g,n}.$$

By Theorem 0.7, the class $\mathsf{DR}_{g,A}^{\mathsf{op}}$ is computed by the tautological class

$$\mathsf{P}^{g}_{g,A,d} \in \mathsf{CH}^{g}_{\mathsf{op}}(\mathfrak{Pic}_{g,n,k(2g-2)}).$$

By Proposition 4.3, the action of $\mathsf{P}_{g,A,d}^g$ on $[\overline{\mathcal{M}}_{g,n}]$ is indeed given by Pixton's original formula $2^{-g}P_q^{g,k}(\tilde{A})$, finishing the proof.

The steps of the proof of Theorem 8.1 are summarized as follows:

$$\begin{split} \mathsf{H}_{g,A}^{k} &= \mathsf{DR}_{g,A,\omega^{k}} & ([40, \text{ Theorem 1.1}]) \\ &= \mathsf{DR}_{g,A}^{\mathsf{op}}(\varphi_{\omega_{\pi}^{k}})([\overline{\mathcal{M}}_{g,n}]) & (\text{Theorem 0.1}) \\ &= \mathsf{P}_{g,A,d}^{g}(\varphi_{\omega_{\pi}^{k}})([\overline{\mathcal{M}}_{g,n}]) & (\text{Theorem 0.7}) \\ &= 2^{-g} P_{g}^{g,k}(\tilde{A}). & (\text{Proposition 4.3}). \end{split}$$

The result provides a completely geometric representative of Pixton's cycle in terms of twisted k-differentials. Theorem 0.9 of $\S0.5$ is the k=1 case of Theorem 8.1.

8.2. Closures

Let $A = (a_1, ..., a_n)$ be a vector of integers satisfying

$$\sum_{i=1}^{n} a_i = k(2g-2).$$

A careful investigation of the closure

$$\mathcal{H}_g^k(A) \subset \overline{\mathcal{H}}_g^k(A) \subset \overline{\mathcal{M}}_{g,n}$$

is carried out in [8] and [9]. By a simple method presented in [28, Appendix A] and [72, §3.4], Theorem 8.1 easily determines the cycle classes of the closures

$$[\overline{\mathcal{H}}_{g}^{k}(A)] \in \mathsf{CH}_{*}(\overline{\mathcal{M}}_{g,n})$$

for the following two cases:

• k=1 and all a_i are non-negative, when $\overline{\mathcal{H}}_q^k(A)$ has pure codimension g-1;

• $k \ge 1$ and A is not of the form $A = k \cdot A'$ with a vector A' of non-negative integers, when $\overline{\mathcal{H}}_{q}^{k}(A)$ has pure codimension g.
In particular, from the recursive formula for $[\overline{\mathcal{H}}_g^k(A)]$ and the fact that Pixton's cycle on $\overline{\mathcal{M}}_{q,n}$ is tautological, the following is immediate.

COROLLARY 8.2. The cycles $[\overline{\mathcal{H}}_{q}^{k}(A)]$ are tautological classes in $\mathsf{CH}_{*}(\overline{\mathcal{M}}_{q,n})$.

In the case k=1, Corollary 8.2 was known by work of Sauvaget [70], who gave a different approach to $[\overline{\mathcal{H}}_g^1(A)]$ in terms of tautological classes. The recursive formulas for $[\overline{\mathcal{H}}_g^k(A)]$ from Corollary 8.2 have been implemented⁽⁴¹⁾ in the software [23] for computations in the tautological ring of $\overline{\mathcal{M}}_{q,n}$.

Another application of Conjecture A is presented in the recent paper [71] by Sauvaget. The paper studies moduli spaces of flat surfaces of genus g with conical singularities at marked points $p_1, ..., p_n$. The singularities have fixed cone angles $2\pi\alpha_i$, for $\alpha_1, ..., \alpha_n \in \mathbb{R}$, summing to 2g-2+n. If all α_i are rational, the spaces of flat surfaces naturally contain $\mathcal{H}_a^k(kA)$, for

$$A = (\alpha_i - 1)_{i=1}^n$$

as closed subsets (for k sufficiently divisibly). These subsets equidistribute (with respect to natural measures) as $k \to \infty$. Using the equidistribution, Sauvaget is able to apply the recursive expression for $\overline{\mathcal{H}}_g^k(kA)$ from Conjecture A to derive an explicit recursion for the volumes of the moduli spaces of flat surfaces.

8.3. k-twisted DR cycles with targets

We define here k-twisted double ramification cycles with targets via the class $\mathsf{DR}_{a,A}^{\mathsf{op}}$.

Let X be a non-singular projective variety with line bundle \mathcal{L} and an effective curve class $\beta \in H_2(X, \mathbb{Z})$. Let

$$d_{\beta} = \int_{\beta} c_1(\mathcal{L}).$$

Let $k \in \mathbb{Z}$ and $A = (a_1, ..., a_n) \in \mathbb{Z}^n$ satisfy

$$d_{\beta} + k(2g - 2 + n) = \sum_{i=1}^{n} a_i.$$

Consider the morphism

$$\varphi: \mathcal{M}_{g,n}(X,\beta) \longrightarrow \mathfrak{Pic}_{g,n,d_{\beta}+k(2g-2+n)}$$

 $^(^{41})$ In the ongoing project [22], the authors study formulas for Euler characteristics of strata of differentials in terms of intersection numbers on the compactification of these strata constructed in [10]. The implementation of $[\overline{\mathcal{H}}_a^1(A)]$ has played a role in corroborating their formulas.

defined by the universal data

$$\pi: \mathcal{C}_{g,n,\beta} \longrightarrow \overline{\mathcal{M}}_{g,n}(X,\beta), \quad f^* \mathcal{L} \otimes \omega_{log}^{\otimes k} \longrightarrow \mathcal{C}_{g,n,\beta}, \tag{8.2}$$

where $f: \mathcal{C}_{g,n,\beta} \to X$ is the universal map.

Definition 8.3. The k-twisted X-valued double ramification cycle is defined by

$$\mathsf{DR}^{k}_{g,n,\beta}(X,\mathcal{L}) = \mathsf{DR}^{\mathsf{op}}_{g,A}(\varphi)([\overline{\mathcal{M}}_{g,n}(X,\beta)]^{\mathsf{vir}}) \in \mathsf{CH}_{\mathrm{vdim}(g,n,\beta)-g}(\overline{\mathcal{M}}_{g,n}(X,\beta))$$

In the notation of [43, §0.4], let $\mathsf{P}^{c,k,r}_{g,A,\beta}(X,\mathcal{L})$ be the codimension c part of the following expression

$$\sum_{\substack{\Gamma \in \mathsf{G}_{g,n,\beta}(X)\\w \in \mathsf{W}_{\Gamma,r,k}(X)}} \frac{r^{-h^{1}(\Gamma)}}{|\operatorname{Aut}(\Gamma)|} j_{\Gamma_{\delta}*} \left[\prod_{i=1}^{n} \exp\left(\frac{1}{2}a_{i}^{2}\psi_{i} + a_{i}\xi_{i}\right) \right]$$
$$\times \prod_{v \in \mathcal{V}(\Gamma)} \exp\left(-\frac{1}{2}\eta(v) - k\eta_{1,1}(v) - \frac{k^{2}}{2}\eta_{0,2}(v)\right)$$
$$\times \prod_{e=(h,h')\in \mathcal{E}(\Gamma)} \frac{1}{\psi_{h} + \psi_{h'}} \left(1 - \exp\left(-\frac{w(h)w(h')}{2}(\psi_{h} + \psi_{h'})\right)\right).$$

The definition of the admissible k-weightings $w \in W_{\Gamma,r,k}(X)$ is similar to that in §0.3.4, but with the condition (iii) replaced by

$$k(2g(v)-2+n(v))+\int_{\beta(v)}c_1(\mathcal{L})=\sum_{v(h)=v}w(h)\quad\text{for }v\in \mathcal{V}(\Gamma).$$

As in the case k=0 discussed in [43, Proposition 1], the class $\mathsf{P}_{g,A,\beta}^{c,k,r}(X,\mathcal{L})$ is polynomial in r for all sufficiently large r. Denote by $\mathsf{P}_{g,A,\beta}^{c,k}(X,\mathcal{L})$ the value at r=0 of this polynomial.

By Theorem 0.7 and a slight generalization of the procedure for pulling back Chow cohomology classes from $\mathfrak{Pic}_{g,n,d_{\beta}+k(2g-2+n)}$ to $\overline{\mathcal{M}}_{g,n}(X,\beta)$ described in [43, §1.5], we have

$$\begin{split} \mathsf{DR}^k_{g,n,\beta}(X,\mathcal{L}) &= \mathsf{DR}^{\mathsf{op}}_{g,A}(\varphi)([\overline{\mathcal{M}}_{g,n}(X,\beta)]^{\mathsf{vir}}) \\ &= \mathsf{P}^g_{g,A,d_\beta+k(2g-2+n)}(\varphi)([\overline{\mathcal{M}}_{g,n}(X,\beta)]^{\mathsf{vir}}) \\ &= \mathsf{P}^{g,k}_{g,A,\beta}(X,\mathcal{L}). \end{split}$$

8.4. Proof of Theorem 0.8

For all c > g, we will prove

$$\mathsf{P}_{g,A,d}^c = 0 \in \mathsf{CH}_{\mathsf{op}}^c(\mathfrak{Pic}_{g,n,d}). \tag{8.3}$$

The path is parallel to the proof of Theorem 0.7.

By definition, the claim is equivalent to showing that the map

$$\mathsf{P}^{c}_{q,A,d}(\varphi): \mathsf{CH}_{*}(B) \longrightarrow \mathsf{CH}_{*-c}(B)$$
(8.4)

is zero for every morphism $\varphi: B \to \mathfrak{Pic}_{g,n,d}$ from an (irreducible) finite-type scheme B corresponding to the data

$$C \longrightarrow B, \quad \mathcal{L} \longrightarrow C.$$

Retracing the steps of §5 (and using the invariance Lemma 7.6 for the codimension-c part $\mathsf{P}_{g,A,d}^c$ of Pixton's formula), we can reduce to the situation where \mathcal{L} on C is relatively sufficiently positive with respect to $C \to B$. As in §5.3, we can then find

$$\psi: U_l \longrightarrow B$$

such that ψ^* is injective on Chow groups and such that the composition

$$U_l \longrightarrow B \longrightarrow \mathfrak{Pic}_{g,n,d}$$

factors through $\overline{\mathcal{M}}_{q,n}(\mathbb{P}^l, d)'$. By [4, Theorem 3.2], we have the vanishing

$$\mathsf{P}_{q,A,d}^{c}(\mathbb{P}^{l},\mathcal{O}(1)) = 0 \in \mathsf{CH}_{\mathrm{vdim}(q,n,d)-c}(\overline{\mathcal{M}}_{q,n}(\mathbb{P}^{l},d)).$$

The same combination of Lemma 2.6 and the injectivity of ψ^* then shows the desired vanishing of the map (8.4).

8.5. Connections to past and future results

The relations of Theorem 0.8 generalize several previous results. For g=0 and c=1, the vanishing (8.3) was observed in [45, Proposition 1.2]. In fact, in genus zero, there are many connections to past equations, see [4, §4] for a full discussion with many examples including classical equations and the relations of [52].

Randal-Williams [67] proves a vanishing result in cohomology with integral coefficients on the locus $\mathfrak{Pic}_{g,0,d}^{\mathsf{sm}}$ of smooth curves for every $d \in \mathbb{Z}$. We can recover a version of Randal-William's vanishing in operational Chow with \mathbb{Q} -coefficients which extends to all of $\mathfrak{Pic}_{g,0,d}$. By Proposition 4.1 and Lemma 4.2, Pixton formula's on the locus $\mathfrak{Pic}_{g,0,0}^{\mathsf{sm}}$ takes the simple form

$$\mathsf{P}^{c}_{g,\varnothing,0} = \frac{1}{c!} (\mathsf{P}^{1}_{g,\varnothing,0})^{c}, \quad \mathsf{P}^{1}_{g,\varnothing,0} = -\frac{1}{2} \pi_{*}(c_{1}(\mathcal{L})^{2})$$

for the universal curve and the universal line bundle

$$\pi: \mathfrak{C} \longrightarrow \mathfrak{Pic}_{q,0,0}, \quad \mathcal{L} \longrightarrow \mathfrak{C}.$$

We claim that, up to scaling, the relation

$$\Omega^{g+1} = 0$$

of [67, Theorem A] is exactly the restriction of the pullback of the relation

$$(\mathsf{P}^{1}_{g,\varnothing,0})^{g+1} = (g+1)! \, \mathsf{P}^{g+1}_{g,\varnothing,0} = 0 \in \mathsf{CH}^{g+1}_{\mathsf{op}}(\mathfrak{Pic}_{g,0,0})$$

under the morphism

$$\begin{split} \mathfrak{Pic}_{g,0,d} &\longrightarrow \mathfrak{Pic}_{g,0,0}, \\ (C,\mathcal{L}) &\longmapsto (C,\mathcal{L}^{\otimes 2g-2} \otimes \omega_C^{\otimes (-d)}). \end{split}$$

Indeed, over the locus of smooth curves, the pullback of $\mathsf{P}^1_{q,\varnothing,0}$ is given by

$$\begin{aligned} &-\frac{1}{2}\pi_*(c_1(\mathcal{L}^{\otimes 2g-2}\otimes\omega_C^{\otimes(-d)})^2) \\ &= -\frac{1}{2}((2g-2)^2\pi_*(c_1(\mathcal{L})^2) - 2d(2g-2)\pi_*(c_1(\mathcal{L})c_1(\omega_\pi)) + d^2\pi_*(c_1(\omega_\pi)^2)), \end{aligned}$$

which matches the definition of Ω given in [67, Theorem A] up to scalars.

In Gromov–Witten theory, pulling back (8.3) under the morphisms

$$\overline{\mathcal{M}}_{g,n}(X,\beta) \longrightarrow \mathfrak{Pic}_{g,n,d}$$

described in §8.3 and capping with the virtual class $[\overline{\mathcal{M}}_{g,n}(X,\beta)]^{\mathsf{vir}}$ simply recovers the known vanishing

$$\mathsf{P}_{g,A,\beta}^{c,k}(X,\mathcal{L}) = 0 \in \mathsf{CH}_{\mathrm{vdim}(g,n,\beta)-c}(\overline{\mathcal{M}}_{g,n}(X,\beta))$$
(8.5)

for c > g proven in [4]. However, there are new applications for *reduced* Gromov–Witten theory. Indeed, for a target X having a non-degenerate holomorphic 2-form, the virtual class of $\overline{\mathcal{M}}_{g,n}(X,\beta)$ vanishes when $\beta \neq 0$. To define invariants for such targets, the reduced class

$$[\overline{\mathcal{M}}_{g,n}(X,\beta)]^{\mathsf{red}} \in \mathsf{CH}_*(\overline{\mathcal{M}}_{g,n}(X,\beta))$$

is used instead; see [15], [58]. By pulling back (8.3) and capping with $[\overline{\mathcal{M}}_{g,n}(X,\beta)]^{\mathsf{red}}$, we obtain new relations among reduced Gromov–Witten invariants. An application to the Gromov–Witten theory of K3 surfaces will appear in [5] related to conjectures of [61].

References

- ABRAMOVICH, D., CADMAN, C., FANTECHI, B. & WISE, J., Expanded degenerations and pairs. Comm. Algebra, 41 (2013), 2346–2386.
- [2] ABRAMOVICH, D. & WISE, J., Birational invariance in logarithmic Gromov–Witten theory. Compos. Math., 154 (2018), 595–620.
- [3] ARTIN, M., Versal deformations and algebraic stacks. Invent. Math., 27 (1974), 165–189.
- [4] BAE, Y., Tautological relations for stable maps to a target variety. Ark. Mat., 58 (2020), 19–38.
- [5] BAE, Y. & BUELLES, T. H., Curves on K3 surfaces in divisibility 2. Forum Math. Sigma, 9 (2021), Paper No. e9, 37 pp.
- [6] BAE, Y., SCHMITT, J. & SKOWERA, J., Chow rings of stacks of prestable curves I. Forum Math. Sigma, 10 (2022), Paper No. e28, 47 pp.
- [7] BAE, Y. & SCHMITT, J., Chow rings of stacks of prestable curves II. To appear in *J. Reine* Angew. Math.
- [8] BAINBRIDGE, M., CHEN, D., GENDRON, Q., GRUSHEVSKY, S. & MÖLLER, M., Compactification of strata of Abelian differentials. Duke Math. J., 167 (2018), 2347–2416.
- [9] Strata of k-differentials. Algebr. Geom., 6 (2019), 196–233.
- [10] The moduli space of multi-scale differentials. Preprint, 2019. arXiv:1910.13492[math.AG].
- [11] BEHREND, K. & FANTECHI, B., The intrinsic normal cone. Invent. Math., 128 (1997), 45–88.
- [12] BIESEL, O. & HOLMES, D., Fine Compactified Moduli of Enriched Structures on Stable Curves. Mem. Amer. Math. Soc., 285 (2023).
- [13] BORNE, N. & VISTOLI, A., Parabolic sheaves on logarithmic schemes. Adv. Math., 231 (2012), 1327–1363.
- [14] BOSCH, S., LÜTKEBOHMERT, W. & RAYNAUD, M., Néron Models. Ergebnisse der Mathematik und ihrer Grenzgebiete, 21. Springer, Berlin-Heidelberg, 1990.
- [15] BRYAN, J. & LEUNG, N. C., The enumerative geometry of K3 surfaces and modular forms. J. Amer. Math. Soc., 13 (2000), 371–410.
- [16] BURYAK, A., GUÉRÉ, J. & ROSSI, P., DR/DZ equivalence conjecture and tautological relations. *Geom. Topol.*, 23 (2019), 3537–3600.
- [17] BURYAK, A., SHADRIN, S., SPITZ, L. & ZVONKINE, D., Integrals of ψ-classes over double ramification cycles. Amer. J. Math., 137 (2015), 699–737.
- [18] CAVALIERI, R., MARCUS, S. & WISE, J., Polynomial families of tautological classes on *M^{rt}_{g,n}*. J. Pure Appl. Algebra, 216 (2012), 950–981.
- [19] CHEN, Q., JANDA, F., RUAN, Y. & SAUVAGET, A., Towards logarithmic GLSM: the *r*-spin case. *Geom. Topol.*, 26 (2022), 2855–2939.
- [20] CLADER, E., GRUSHEVSKY, S., JANDA, F. & ZAKHAROV, D., Powers of the theta divisor and relations in the tautological ring. Int. Math. Res. Not. IMRN, 24 (2018), 7725–7754.
- [21] CLADER, E. & JANDA, F., Pixton's double ramification cycle relations. Geom. Topol., 22 (2018), 1069–1108.
- [22] COSTANTINI, M., MÖLLER, M. & ZACHHUBER, J., The Chern classes and Euler characteristic of the moduli spaces of Abelian differentials. *Forum Math. Pi*, 10 (2022), Paper No. e16, 55 pp.
- [23] DELECROIX, V., SCHMITT, J. & VAN ZELM, J., Admcycles—a Sage package for calculations in the tautological ring of the moduli space of stable curves. J. Softw. Algebra Geom., 11 (2021), 89–112.
- [24] EDIDIN, D. & SATRIANO, M., Towards an intersection Chow cohomology theory for GIT quotients. Transform. Groups, 25 (2020), 1103–1124.

- [25] FABER, C. & PANDHARIPANDE, R., Relative maps and tautological classes. J. Eur. Math. Soc. (JEMS), 7 (2005), 13–49.
- [26] FAN, H., WU, L. & YOU, F., Structures in genus-zero relative Gromov–Witten theory. J. Topol., 13 (2020), 269–307.
- [27] FANTECHI, B., GÖTTSCHE, L., ILLUSIE, L., KLEIMAN, S. L., NITSURE, N. & VISTOLI, A., Fundamental Algebraic Geometry. Mathematical Surveys and Monographs, 123. Amer. Math. Soc., Providence, RI, 2005.
- [28] FARKAS, G. & PANDHARIPANDE, R., The moduli space of twisted canonical divisors. J. Inst. Math. Jussieu, 17 (2018), 615–672.
- [29] FULTON, W., Intersection Theory. Ergebnisse der Mathematik und ihrer Grenzgebiete, 2. Springer, Berlin-Heidelberg, 1984.
- [30] GRABER, T. & PANDHARIPANDE, R., Constructions of nontautological classes on moduli spaces of curves. *Michigan Math. J.*, 51 (2003), 93–109.
- [31] GRABER, T. & VAKIL, R., Relative virtual localization and vanishing of tautological classes on moduli spaces of curves. *Duke Math. J.*, 130 (2005), 1–37.
- [32] GROSS, M. & SIEBERT, B., Logarithmic Gromov–Witten invariants. J. Amer. Math. Soc., 26 (2013), 451–510.
- [33] GRUSHEVSKY, S. & ZAKHAROV, D., The double ramification cycle and the theta divisor. Proc. Amer. Math. Soc., 142 (2014), 4053–4064.
- [34] GUÉRÉ, J., A generalization of the double ramification cycle via log-geometry. Preprint, 2016. arXiv:1603.09213[math.AG].
- [35] HAIN, R., Normal functions and the geometry of moduli spaces of curves, in *Handbook of Moduli*. Vol. I, Adv. Lect. Math. (ALM), 24, pp. 527–578. Int. Press, Somerville, MA, 2013.
- [36] HERR, L., MOLCHO, S., PANDHARIPANDE, R. & WISE, J., Birational models for logarithmic Gromov–Witten theory. In preparation.
- [37] HOLMES, D., Extending the double ramification cycle by resolving the Abel–Jacobi map. J. Inst. Math. Jussieu, 20 (2021), 331–359.
- [38] HOLMES, D., KASS, J. L. & PAGANI, N., Extending the double ramification cycle using Jacobians. Eur. J. Math., 4 (2018), 1087–1099.
- [39] HOLMES, D., PIXTON, A. & SCHMITT, J., Multiplicativity of the double ramification cycle. Doc. Math., 24 (2019), 545–562.
- [40] HOLMES, D. & SCHMITT, J., Infinitesimal structure of the pluricanonical double ramification locus. Compos. Math., 157 (2021), 2280–2337.
- [41] HOLMES, D. & SCHWARZ, R., Logarithmic intersections of double ramification cycles. Algebr. Geom., 9 (2022), 574–605.
- [42] JANDA, F., PANDHARIPANDE, R., PIXTON, A. & ZVONKINE, D., Double ramification cycles on the moduli spaces of curves. *Publ. Math. Inst. Hautes Études Sci.*, 125 (2017), 221–266.
- [43] Double ramification cycles with target varieties. J. Topol., 13 (2020), 1725–1766.
- [44] DE JONG, A. J., Smoothness, semi-stability and alterations. Inst. Hautes Études Sci. Publ. Math., 83 (1996), 51–93.
- [45] DE JONG, A. J. & STARR, J., Divisor classes and the virtual canonical bundle for genus 0 maps, in *Geometry over Nonclosed Fields*, Simons Symp., pp. 97–126. Springer, Cham, 2017.
- [46] KATO, F., Log smooth deformation and moduli of log smooth curves. Internat. J. Math., 11 (2000), 215–232.

- [47] KATO, K., Logarithmic structures of Fontaine-Illusie, in Algebraic Analysis, Geometry, and Number Theory (Baltimore, MD, 1988), pp. 191–224. Johns Hopkins Univ. Press, Baltimore, MD, 1989.
- [48] KIM, B., Logarithmic stable maps, in New Developments in Algebraic Geometry, Integrable Systems and Mirror Symmetry (RIMS, Kyoto, 2008), Adv. Stud. Pure Math., 59, pp. 167–200. Math. Soc. Japan, Tokyo, 2010.
- [49] KRESCH, A., Cycle groups for Artin stacks. Invent. Math., 138 (1999), 495–536.
- [50] Flattening stratification and the stack of partial stabilizations of prestable curves. Bull. Lond. Math. Soc., 45 (2013), 93–102.
- [51] LAUMON, G. & MORET-BAILLY, L., Champs algébriques. Ergebnisse der Mathematik und ihrer Grenzgebiete, 39. Springer, Berlin-Heidelberg, 2000.
- [52] LEE, Y.-P. & PANDHARIPANDE, R., A reconstruction theorem in quantum cohomology and quantum K-theory. Amer. J. Math., 126 (2004), 1367–1379.
- [53] LI, J., Stable morphisms to singular schemes and relative stable morphisms. J. Differential Geom., 57 (2001), 509–578.
- [54] A degeneration formula of GW-invariants. J. Differential Geom., 60 (2002), 199–293.
- [55] MANOLACHE, C., Virtual pull-backs. J. Algebraic Geom., 21 (2012), 201–245.
- [56] MARCUS, S. & WISE, J., Logarithmic compactification of the Abel–Jacobi section. Proc. Lond. Math. Soc., 121 (2020), 1207–1250.
- [57] Stable maps to rational curves and the relative jacobian. Preprint, 2013. arXiv:1310.5981[math.AG].
- [58] MAULIK, D. & PANDHARIPANDE, R., Gromov-Witten theory and Noether-Lefschetz theory, in A Celebration of Algebraic Geometry, Clay Math. Proc., 18, pp. 469–507. Amer. Math. Soc., Providence, RI, 2013.
- [59] MOLCHO, S., PANDHARIPANDE, R. & SCHMITT, J., The Hodge bundle, the universal 0section, and the log Chow ring of the moduli space of curves. *Compos. Math.*, 159 (2023), 306–354.
- [60] MOLCHO, S. & WISE, J., The logarithmic Picard group and its tropicalization. Compos. Math., 158 (2022), 1477–1562.
- [61] OBERDIECK, G. & PANDHARIPANDE, R., Curve counting on K3×E, the Igusa cusp form χ_{10} , and descendent integration, in K3 Surfaces and their Moduli, Progr. Math., 315, pp. 245–278. Birkhäuser/Springer, Cham, 2016.
- [62] OBERDIECK, G. & PIXTON, A., Holomorphic anomaly equations and the Igusa cusp form conjecture. *Invent. Math.*, 213 (2018), 507–587.
- [63] OLSSON, M., Algebraic Spaces and Stacks. Amer. Math. Soc. Colloq. Publ., 62. Amer. Math. Soc., Providence, RI, 2016.
- [64] PANDHARIPANDE, R., A calculus for the moduli space of curves, in Algebraic Geometry (Salt Lake City, UT, 2015), Proc. Sympos. Pure Math., 97, pp. 459–487. Amer. Math. Soc., Providence, RI, 2018.
- [65] PANDHARIPANDE, R., PIXTON, A. & ZVONKINE, D., Tautological relations via r-spin structures. J. Algebraic Geom., 28 (2019), 439–496.
- [66] PIXTON, A., Generalized boundary strata classes, in *Geometry of Moduli*, Abel Symp., 14, pp. 285–293. Springer, Cham, 2018.
- [67] RANDAL-WILLIAMS, O., Relations among tautological classes revisited. Adv. Math., 231 (2012), 1773–1785.
- [68] RANGANATHAN, D., A note on cycles of curves in a product of pairs. Preprint 2019. arXiv:1910.00239[math.AG].
- [69] RAYNAUD, M. & GRUSON, L., Critères de platitude et de projectivité. Techniques de "platification" d'un module. *Invent. Math.*, 13 (1971), 1–89.

- [70] SAUVAGET, A., Cohomology classes of strata of differentials. Geom. Topol., 23 (2019), 1085–1171.
- [71] Volumes of moduli spaces of flat surfaces. Preprint, 2020. arXiv: 2004.03198 [math.AG].
- [72] SCHMITT, J., Dimension theory of the moduli space of twisted k-differentials. Doc. Math., 23 (2018), 871–894.
- [73] STACKS PROJECT AUTHORS, T. The stacks project. Open source textbook. Available at https://stacks.math.columbia.edu.
- [74] SUZUMURA, K., Remarks on the theory of collective choice. *Economica*, 43 (1976), 381–390.
- [75] TSENG, H.-H. & YOU, F., Higher genus relative and orbifold Gromov–Witten invariants. Geom. Topol., 24 (2020), 2749–2779.
- [76] VISTOLI, A., Intersection theory on algebraic stacks and on their moduli spaces. Invent. Math., 97 (1989), 613–670.
- [77] YIN, Q., Cycles on curves and Jacobians: a tale of two tautological rings. Algebr. Geom., 3 (2016), 179–210.

YOUNGHAN BAE Department of Mathematics ETH Zürich Rämistrasse 101 8092 Zürich Switzerland younghan.bae@math.ethz.ch

RAHUL PANDHARIPANDE Department of Mathematics ETH Zürich Rämistrasse 101 8092 Zürich Switzerland rahul@math.ethz.ch

ROSA SCHWARZ Mathematisch Instituut Leiden Niels Bohrweg 1 2333 CA Leiden Netherlands r.m.schwarz@math.leidenuniv.nl

Received May 5, 2020 Received in revised form May 11, 2021 DAVID HOLMES Mathematisch Instituut Leiden Niels Bohrweg 1 2333 CA Leiden Netherlands holmesdst@math.leidenuniv.nl

JOHANNES SCHMITT Institut für Mathematik University of Zürich Winterthurerstrasse 190 8057 Zürich Switzerland jo314schmitt@googlemail.com